The impact of water erosion and climate change on global cereal yields

Tony William Carr

A dissertation submitted in partial fulfillment of the requirements for the degree of **Doctor of Philosophy**

of University College London.

Institute for Sustainable Resources Bartlett Faculty of the Built Environment University College London

August 3, 2022

Declaration of ownership

I, Tony Carr, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

.....

Tony Carr

List of Publications

- Carr, T. W., Balkovič, J., Dodds, P. E., Folberth, C., and Skalsky, R. (2021). The impact of water erosion on global maize and wheat productivity. Agriculture, Ecosystems & Environment, 322, 107655.
- Carr, T. W., Balkovič, J., Dodds, P. E., Folberth, C., Fulajtar, E., and Skalsky, R. (2020). Uncertainties, sensitivities and robustness of simulated water erosion in an EPIC-based global-gridded crop model. Biogeosciences, 17, 5263–5283.
- Carr, T. W., Balkovič, J., Folberth, C., Fulajtar, E., and Skalsky, R. (2019). Simulation and evaluation of global sediment runoff and soil organic carbon removal by erosion in maize fields under varying field management, In: FAO. Proceedings of the Global Symposium on Soil Erosion 2019. Rome.

Acknowledgements

First, I would like to thank my supervisor, Paul Dodds, for supporting me throughout, taking the time for many discussions and giving me useful feedback on the drafts of this thesis.

This thesis would not have been possible without the guidance and generosity of Juraj Balkovič, Christian Folberth and Rastislav Skalsky at IIASA in Laxenburg, Austria. I was a regular visitor at IIASA and was given access to their resources. They have spent much of their valuable time sharing ideas, data and contacts and showing me how to use their global crop model. I am extremely grateful to them for that. I would also like to thank Emil Fulajtar from IAEA in Vienna, Austria, who helped me understand the complexities of soil erosion and collecting field data. I also received much help from Jason Neff and Won Seok Jang, who invited me to the University of Colorado Boulder, USA, and provided some valuable ideas that have been used in this work.

I am grateful for the generous grant from the Grantham Foundation for the Protection of the Environment that enabled me to complete my PhD at UCL ISR, and for the funding from the German Federal Ministry of Education and Research for an extended stay at IIASA.

At UCL ISR, I would like to thank Alvaro Calzadilla and Paul Ekins for their support and introduction to my collaborators. Alvaro and Paul paved the way for funding my visits to IIASA, the University of Colorado Boulder and to three international conferences, which were crucial in making the contacts and developing the ideas that led to this work. I am also grateful to David Norse for sharing his vast experience in soil and agriculture. Thanks for all the fun I have had over the years with the many PhD students at UCL, especially Theodoros Arvanitopoulos, Wan-Ting Hsu, Alexandros Sfyridis and Arkaitz Usubiaga.

A final thanks to my partner Patricia for her solid belief and support, and to my parents for supporting me in countless ways that have led to this PhD.

Abstract

Water erosion can degrade soils and reduce agricultural productivity. Global modelling studies are increasingly used to explore the potential impacts of environmental change, especially climate change, on crop yields. Despite its impact on crops and its direct link to the climate system, water erosion has not previously been considered in projections of the impacts of climate change on global crop yields. Similarly, global water erosion assessments have not considered how water erosion affects crops, and vice versa. This thesis identifies a method to estimate global water erosion patterns and evaluates it using a global database of erosion measurements. It then examines the global potential impacts of erosion on maize and wheat yields, and explores how these might be affected by climate change in the future.

The Environmental Policy Integrated Climate (EPIC) crop model is used with global datasets on soil, topography, climate and field management to estimate erosion and its impacts on crop yields. Estimates using weather data for the years 1980–2010 suggest that water erosion reduced maize and wheat yields by 3% annually in around half of global arable land under current land management regimes. The estimated annual maize and wheat production losses due to water erosion accounted for less than 1% of the global production volume. However, water erosion in maize and wheat fields at low latitudes is likely to increase in the future due to the adverse effects of climate change on both crops.

Uncertainties remain about estimates of water erosion that need to be addressed through better integration of models and observations. Nevertheless, the integrated biophysical modelling framework developed in this study can provide a link between robust estimates of water erosion, economics, and policy making that have so far been lacking in global agricultural assessments.

Impact statement

It is necessary to improve the quality of indicators on the health of soil resources and to quantify the current and future impacts of water erosion in agriculture in physical and economic terms (Montanarella, 2007; Montanarella et al., 2016; Nkonya et al., 2011). Although water erosion is considered in numerous agricultural and environmental policy programs (Alewell et al., 2019), soil information available for policy formulation is frequently based on studies from the 1990s based on observations made in the 1980s or earlier (Montanarella et al., 2016).

Global gridded crop models (GGCMs), typically combinations of crop or ecosystem models and global gridded input data infrastructures, have become essential tools for analysing global phenomena, most notably climate change, that affect agriculture and ecosystems (Mueller et al., 2017). GGCMs have the capacity to develop large-scale indicators and to inform about agricultural productivity in a transparent and consistent way across large areas (Mueller et al., 2017). Although water erosion plays an important part in the productivity of cropland and the impact of agriculture on the environment, it is frequently neglected in commonly used GGCMs. Moreover, integrated assessment modelling studies that aim to assess environmental problems in a holistic way by linking social, economic, and environmental considerations (Laniak et al., 2013) require data on water erosion impacts on agricultural production (Porwollik et al., 2019).

The GGCM EPIC-IIASA is used to address the gaps in the literature regarding the links between water erosion and crop cultivation in various large-scale and global modelling studies to provide insights into the impact of water erosion in current and future agricultural environments. EPIC-IIASA has been confirmed as a reliable tool for global crop yield projections and stands out against comparable global crop models due to its detailed representation of soil processes including water erosion (Folberth et al., 2019; Balkovič et al., 2014).

By simulating the interactions between crops and water erosion on a global scale, EPIC-IIASA can provide much-needed indicators on the effects of water erosion on soil quality and productivity, and analyse patterns of water erosion in response to field management and environmental drivers. Moreover, by analysing the robustness of global water erosion estimates across different environments, this study addresses concerns about the uncertainty of large-scale and global water erosion estimates (Evans and Boardman, 2016b,a). Transparency of the robustness and uncertainty of model outputs is especially important if they are intended to inform policy programs (Panagos et al., 2016).

Finally, this study contributes to a better understanding of potential future pathways of global cereal

production by analysing the combined impact of water erosion and climate change on maize and wheat yields. Although climate change and water erosion may affect farmers simultaneously, and in combination can multiply pressures on crops, they are often studied separately (Webb et al., 2017). Both climate change and water erosion impacts on crop production are spread unevenly around the globe. Therefore, research on both environmental change topics together will help identify hotspots where maize and wheat production are at risk from both climate change and water erosion, and thus increase knowledge of regional differences in the resilience of future crop production.

Contents

De	Declaration of ownership 3					
Lis	List of Publications 5					
Ac	know	ledgements	7			
Ab	ostrac	t	9			
Im	pact s	statement	11			
Co	ontent	S	13			
Lis	st of F	igures	19			
Lis	st of T	ables	23			
Lis	st of A	bbreviations	25			
1	Intro	oduction	27			
	1.1	The pressure on soil resources through the intensification of agriculture	27			
	1.2	Land degradation	28			
	1.3	Land degradation by water erosion	29			
	1.4	Processes leading to water erosion	30			
	1.5	The impact of agriculture on water erosion	32			
	1.6	The impact of water erosion on crop productivity	33			
	1.7	The impact of climate change on water erosion	34			
	1.8	The impact of climate change on crops	37			
	1.9	The use of crop models in impact assessments	38			
	1.10	Thesis aims and objectives	39			
	1.11	Thesis overview	40			
2	Liter	ature review of water erosion assessments	43			
	2.1	Introduction	43			
	2.2	Global land degradation assessments	44			

	2.3	Water erosion assessments	46
		2.3.1 Large-scale water erosion assessments	46
		2.3.2 Common water erosion modelling techniques	48
		2.3.3 The Universal Soil Loss Equation	49
		2.3.4 Uncertainties in water erosion models	50
		2.3.5 Common methods for measuring water erosion based on field experiments	53
	2.4	Studies on the impact of water erosion on crop yields	55
		2.4.1 Methods to quantify the impact of water erosion on crop yields	56
		2.4.2 Large-scale assessments of the impact of water erosion on crop yields	56
	2.5	Studies on the impact of climate change on water erosion	57
	2.6	Discussion	60
		2.6.1 Models are the preferred method for assessing water erosion at large scales	60
		2.6.2 A better evaluation of the modelled water erosion estimates is needed	60
		2.6.3 The impact of water erosion on crops is not adequately quantified	61
		2.6.4 The impact of climate change on water erosion is unclear	61
	2.7	Conclusion	62
3	Lite	rature review of crop models	65
	3.1	Introduction	65
	3.2	Crop model types	66
	3.3	Typical crop model routines to simulate crop growth	67
	3.4	Crop models and their applications	68
	3.5	Global climate change impact assessments using crop models	70
	3.6	Uncertainty and limitations of crop models	73
	3.7	Discussion	76
	3.8	Conclusion	78
4	Ove	rall methodology	79
-	4.1	Study framework	79
	4.2	The EPIC crop model	80
		4.2.1 Crop growth	81
		4.2.2 Crop growth constraints	82
		4.2.3 Soil	83
		4.2.4 Hydrology	84
		4.2.5 Water erosion	84
	4.3	EPIC-IIASA	87
		4.3.1 Global grids and input data	88
		4.3.2 Climate change scenarios	89
		4.3.3 Field management scenarios	89
		-	

		4.3.4 Baseline field management scenario
	4.4	Uncertainty analysis of model outputs
	4.5	Sensitivity analysis of water erosion simulation
	4.6	Evaluation of model outputs
		4.6.1 Evaluation of simulated water erosion against field measurements 95
		4.6.2 Evaluation of simulated crop yields against reported yields
	4.7	The computation process
5	Unc	ertainties, sensitivities and robustness of simulated water erosion with EPIC-IIASA 93
	5.1	Introduction
	5.2	Methodology
		5.2.1 Summary of methodology
		5.2.2 Aggregation of model outputs
	5.3	Results
		5.3.1 Sources of model uncertainty related to management assumptions and method
		selection
		5.3.2 Main drivers of the global erosion model
		5.3.3 Evaluation of simulation results against field data
	5.4	Discussion
		5.4.1 Varying robustness of simulated water erosion in different global regions 100
		5.4.2 Sources of uncertainties in global water erosion estimates
		5.4.3 The difficulty of evaluating large-scale erosion estimates with field data 110
	5.5	Conclusion
6	The	impact of water erosion on global maize and wheat yields 11:
	6.1	Introduction
	6.2	Methodology
		6.2.1 Summary of methodology
		6.2.2 Aggregation of model outputs
	6.3	Results
		6.3.1 The impact of water erosion on global maize and wheat yields
		6.3.2 Fertiliser use and environmental drivers affect the impacts of water erosion 120
		6.3.3 The impact of water erosion on total maize and wheat production
		6.3.4 The relevance of slope and field management assumptions
		6.3.5 Evaluation of simulated crop yields
	6.4	Discussion
		6.4.1 Erosion-induced crop yield losses and fertiliser requirements for compensation . 123
		6.4.2 Potential impacts of water erosion on livelihoods
		6.4.3 Uncertainties in water erosion estimates

		6.4.4	Uncertainties in crop yield estimates	131
	6.5	Conclu	ision	132
7	The	combin	ed effects of climate change and water erosion on global maize and wheat yield	s135
	7.1	Introdu	action	135
	7.2	Metho	dology	137
		7.2.1	Summary of methodology	137
		7.2.2	Aggregation of model outputs	137
	7.3	Results	s	138
		7.3.1	Regions and countries vulnerable to both climate change and water erosion	138
		7.3.2	Interactions between crops and water erosion in response to climate change	142
		7.3.3	Global production changes due to climate change and water erosion	144
	7.4	Discus	sion	147
		7.4.1	Factors driving regional differences in climate change and water erosion impacts	147
		7.4.2	The effects of precipitation and temperature changes on water erosion	149
		7.4.3	Where is water erosion increasing due to climate change?	150
		7.4.4	Field management and land use planning will largely influence the future impact	
			of water erosion	152
		7.4.5	Measures to reduce the adverse effects of water erosion and climate change can	
			be combined	152
		7.4.6	Uncertainties related to crop yield projections and climate change impacts on	
			crops and water erosion	153
	7.5	Conclu	ision	156
8	Over	rall sun	nmary and future research needs	159
	8.1	Summ	ary of insights	159
		8.1.1	Simulation and evaluation of water erosion with EPIC-IIASA	160
		8.1.2	The impact of water erosion on global maize and wheat yields	160
		8.1.3	The impact of climate change and water erosion on maize and wheat yields	161
	8.2	Summ	ary of research contributions	161
		8.2.1	The first evaluation of global water erosion estimates from a GGCM	162
		8.2.2	New large-scale water erosion indicators	162
		8.2.3	New evidence on regional differences in the impact of water erosion in agriculture	e 163
		8.2.4	Improvements in crop yield projections by considering water erosion	164
		8.2.5	New insights into the potential impacts of climate change on global water erosion	164
		8.2.6	New data on global water erosion rates and impacts on crops	165
	8.3	Resear	ch needs	165
		8.3.1	Most important needs to improve large-scale water erosion estimates	165

		8.3.2	Scenarios on future land use and field management changes are needed to esti-	
			mate future water erosion rates	166
		8.3.3	The effects of other land degradation processes must be considered \ldots	168
	8.4	Final c	onclusion	168
A	Cha	pter 4 -	Methodology	171
B	Cha	pter 5	- Uncertainties, sensitivities and robustness of simulated water erosion with	1
	EPI	C-IIASA	A	179
С	Cha	pter 6 -	The impact of water erosion on global maize and wheat yields	183
D	Cha	pter 7 -	The combined effects of climate change and water erosion on global maize and	I
	whe	at yields	5	191
Bi	bliogr	aphy		203

List of Figures

1.1	Increase in global maize, wheat and total cereal production between the years 1961–2018
	and increase in global fertiliser use between the years 1961-2002
1.2	The main links between climate change, erosion, crops and field management
3.1	AgMIP median yield changes (%) for RCP8.5
4.1	Scheme of the modelling framework for generating the model outputs and the different
	steps for analysing the uncertainties, sensitivities and robustness of the modelling approach 80
4.2	Distribution of selected tillage and cover crop scenarios determined by country-specific
	and environmental indicators used for compiling the baseline scenario
5.1	Soil loss due to water erosion in maize and wheat fields simulated with the baseline
	scenario
5.2	Water erosion uncertainty due to field management assumptions and water erosion equa-
	tions
5.3	Prevailing uncertainty due to field management assumptions and water erosion equations. 102
5.4	First-order and total-order Sensitivity indices for slope steepness and precipitation. $$ 104
5.5	Comparison of simulated erosion with measured erosion for different slope and precipi-
	tation classes
5.6	Distribution of low to high slope steepness and annual precipitation in maize and wheat
	fields
5.7	Photos of: Sugar cane cultivation on steep slopes in southern China, maize cultivation
	on strongly eroded slopes in south-west Uganda, abandoned fields and maize cultivation
	on a steep slope in south-west Uganda, degraded and abandoned maize fields on steep
	slopes in northern El Salvador
5.8	Simulated years left until the whole soil profile is eroded under permanent maize and
	wheat cultivation
6.1	The impact of water erosion on global maize and wheat yields simulated with the base-
	line scenario

6.2	Maize and wheat yield losses for the 40 most vulnerable countries estimated with the	
	baseline scenario	20
6.3	Median maize and wheat yield loss per country plotted against erosion and maize and	
	wheat yield loss per grid classified by slope steepness and annual precipitation 12	21
6.4	water erosion vulnerability map for maize and wheat fields	22
6.5	The impact of water erosion on global maize and wheat production	23
6.6	Range of simulated maize and wheat yield losses (% yr^{-1}) in Italy simulated with dif-	
	ferent cropland distribution scenarios	25
6.7	Uncertainty range of the impact of water erosion on maize (a) and wheat (b) yields due	
	to contrasting management scenarios	26
6.8	Comparison of national maize and wheat yields and production simulated with EPIC and	
	reported by the FAO	27
6.9	relative difference between national maize and wheat yields and production simulated	
	with EPIC and reported by the FAO	28
7.1	Relative reduction of maize yields due to water erosion per grid cell and change of maize	
	yields due to climate change per latitude in the medium-term and the long-term 14	10
7.2	Relative reduction of wheat yields due to water erosion per grid cell and change of wheat	
	yields due to climate change per latitude in the medium-term and the long-term 14	1
7.3	Relative medium and long-term changes in maize yields due to climate change and water	
	erosion in major world regions	13
7.4	Relative medium and long-term changes in wheat yields due to climate change and water	
	erosion in major world regions	4
7.5	Relative medium and long-term changes in median maize and wheat yields due to climate	
	change and water erosion per country and RCP	15
7.6	Relative change in water erosion and crop yields due to changing precipitation and in-	
	creasing temperature between the years 2010-2090	16
7.7	Increase in water erosion (%) per 1% increase in precipitation	16
7.8	Changes in global annual average maize and wheat production due to climate change	
	and water erosion	17
Δ 1	Global dataset of slope classes that are used to derive the slope steepness per grid cell 17	71
A.1	Nitrogen (N) Phosphorus (P) and Potassium (K) application rates for maize and wheat	1
A.2	on clobal cropland	77
A 2	Crid calls with invited and reinfed wheet and mains sultivation around the user 2000	12
A.5	Main climate and an the undeted Keenen Coince climate clearification	' Э 1 Э
A.4	Palating and account described to the de	·) 7 4
A.5	Relative area covered by the dominant slope and cropland per grid cell.	'4 74
A.6	Locations of water erosion field data from cropland where coordinates were recorded 17	'4
A.7	Distribution of erosion values measured in agricultural fields	15

A.8	Distribution of slope steepness, annual precipitation, and recorded measurement periods	
	for measured erosion values	175
A.9	Number of measured water erosion records per country	176
B .1	World regions classified according to the United Nations geo-schema	179
B.2	Median deviation between measured and simulated water erosion using the baseline sce-	
	nario with different water erosion equations	180
B.3	Distribution of average water erosion values from 1980 - 2010 simulated with the base-	
	line scenario and weighted for each grid cell.	181
B.4	Comparison of slope inputs and simulated water erosion outputs between the cropland	
	distribution scenario using the most common slopes and the cropland distribution sce-	
	nario using the flattest terrain available in Italy	182
C.1	Maize and wheat yields for each grid cell simulated with the baseline scenario 1	184
C.2	Maize yield losses due to water erosion (% yr^{-1}) for all countries with significant crop	
	yield losses due to water erosion	185
C.3	Wheat yield losses due to water erosion (% yr^{-1}) for all countries with significant crop	
	yield losses due to water erosion	186
C.4	Yield loss due to water erosion plotted against Nitrogen and Phosphorus fertiliser use	
	per grid cell	87
C.5	Soil organic nitrogen loss with sediment runoff from the topsoil layer simulated for each	
	grid cell in maize fields and wheat fields	188
C.6	Soil organic carbon loss with sediment runoff from the topsoil layer simulated for each	
	grid cell in maize fields and wheat fields	189
D.1	Relative changes in mean maize yields for the years 2050-2059 and 2090-2099 simulated	
	with RCP 2.6 and RCP 8.5 compared to mean maize yields of the years 1980-2010	
	simulated with the AGMERRA climate data.	191
D.2	Relative changes in mean wheat yields for the years 2050-2059 and 2090-2099 simulated	
	with RCP 2.6 and RCP 8.5 compared to mean wheat yields of the years 1980-2010	
	simulated with the AGMERRA climate data.	192
D.3	Regional trends in maize yields 1980 - 2099 simulated with the AGMERRA climate data	
	and the RCP 2.6 and RCP 8.5	193
D.4	Regional trends in wheat yields 1980 - 2099 simulated with the AGMERRA climate data	
	and the RCP 2.6 and RCP 8.5	194
D.5	Regional trends in the relative decrease in maize and wheat yields due to water erosion,	
	as well as share of grid cells where water erosion reduces crop yields for 2010-2099	
	simulated with RCP 2.6 and RCP 8.5	195
D.6	Maize and wheat yield reduction due to water erosion for 2090-2099 simulated with	
	different field management scenarios per world region	196

D.7	Grid cells where maize and wheat yields are reduced due to water erosion per field man-	
	agement scenario for the years 2090-2099	97
D.8	Regional trends in water erosion rates in maize and wheat fields between 2010-2099	
	simulated with the RCP 2.6 and RCP 8.5	98
D.9	Regional changes in average precipitation plotted against regional changes in average	
	water erosion between the decade 2010-2019 and the decade 2090-2099 in maize fields 1	99
D.10	Regional changes in average precipitation plotted against regional changes in average	
	water erosion between the decade 2010-2019 and the decade 2090-2099 in wheat fields 2	:00
D.11	Regional gains and losses in annual average maize and wheat production simulated with	
	the RCP 2.6 and RCP 8.5 scenario for the decade 2090-2099 compared to average pro-	
	duction for the years 1980-2010 simulated with the AGMERRA climate data 2	01
D.12	Global maize and wheat production and production losses due to water erosion per field	
	management scenario simulated with the RCP 2.6 and RCP 8.5 scenario, calculated for	
	the decade 2090-2099	202

List of Tables

4.1	Equations to calculate the erosivity factor in each water erosion equation available in EPIC. 85
4.2	Set of rules for field size and slope length estimation for each dominant slope class 88
4.3	Tillage management scenarios for maize and wheat cultivation
4.4	Summary of management assumptions and erosion equation selected for the baseline
	scenario
5.1	First-order sensitivity indices (SI) ranking for the five most sensitive input parameters
	for each water erosion equation
5.2	Total-order sensitivity indices (SI) ranking for the five most sensitive input parameters
	for each water erosion equation
6.1	National maize production (FAO, 2020) and absolute and relative maize production
	losses for the countries with the highest absolute losses
6.2	National wheat production and absolute and relative wheat production losses for the
	countries with the highest absolute losses
A.1	Input parameters directly connected to the water erosion equations in EPIC

List of Abbreviations

- AGMIP: Agricultural Model Intercomparison and Improvement Project
- C: Carbon
- CAP: Common Agricultural Policy
- CN: Curve Number
- DEM: Digital Elevation Model
- EPIC: Environmental Policy Integrated Climate Model
- FAO: Food and Agriculture Organization of the United Nations
- GCM: General Circulation Model
- GGCM: Global Gridded Crop Model
- GLADIS: Global Land Degradation Information System
- GLASOD: Global Assessment of Human-induced Soil Degradation
- IIASA: International Institute for Applied Systems Analysis
- IPBES: Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services
- IPCC: Intergovernmental Panel on Climate Change
- ISI-MIP: Inter-Sectoral Impact Model Intercomparison Project
- ISRIC: World Soil Information Center
- ITPS: Intergovernmental Technical Panel on Soils
- LDN: Land Degradation Neutrality
- MUSLE: Modified Universal Soil Loss Equation
- N: Nitrogen
- P: Phosphorous

- RCP: Representative Concentration Pathway
- RUSLE: Revised Universal Soil Loss Equation
- SDG: Sustainable Development Goals
- SOM: Soil Organic Matter
- UNCCD: United Nations Convention to Combat Desertification
- UNEP: United Nations Environment Programme
- USDA: United States Department of Agriculture
- USLE: Universal Soil Loss Equation

Chapter 1

Introduction

1.1 The pressure on soil resources through the intensification of agriculture

Soil is one of the most important natural resources and essential for agriculture. A healthy soil is crucial for providing nutrients and water to crops. However, soil resources have become increasingly under pressure by the intensification of agriculture. The overuse of soil or land can lead to its degradation and reduces its value for agriculture. Between the 1960s and 2010s, global cereal production nearly tripled (Figure 1.1a). To sustain the productivity of soil during the ongoing intensification of grain cultivation, fertiliser use increased in a similar rate in the second half of the 20th century (Figure 1.1b). The intensification of cereal production in past decades triggered by improved crop varieties, irrigation technology, heavy machinery and increasing use of fertiliser, herbicides and pesticides resulted in major environmental impacts negatively affecting biodiversity, water and soil resources (Tilman et al., 2001; Vitousek et al., 1997).



Figure 1.1: (a) Increase in global maize, wheat and total cereal production between the years 1961–2018 (FAO, 2020). (b) Increase in global fertiliser use between the years 1961-2002 (FAO, 2020).

The intensification of agriculture allowed for an enormous growth in global population, which increased from 2.5 billion in 1950 to 7.8 billion in 2020 (UN, 2020). The raising population in turn resulted in a raising demand in food and forage leading to further intensification of agriculture and pressure on soil resources. This development was accompanied by economic growth leading to a rise in per capita food consumption. Forecasts for the year 2050 predict that the global population will grow to 9.7 billion, the world GDP will be 2.5-fold and per capita income 1.8-fold compared to the year 2012 (Melorose et al., 2015; Alexandratos and Bruinsma, 2012). The *United Nations Food and Agriculture Organization* (FAO) estimates an increase in food production of about 40 - 53% by 2050 compared to 2012, with major differences across regions (FAO, 2018). Whilst future agricultural supply is driven by many factors such as consumer preferences, intensification of agricultural production, expansion of agricultural land, food loss and waste; regional differences are largely driven by the uneven spread of population growth (Valin et al., 2014). Consequently, the increasing pressure on agriculture and soil will vary strongly around the world. Most of the regions where population growth continues to be fast are those with inadequate food consumption and high levels of undernourishment (Gerland et al., 2014). Parts of the regions with a projected high growth in food demand already have low levels of agricultural productivity due to infertile soils (Montanarella et al., 2016).

1.2 Land degradation

Although soil degradation is estimated to have been ongoing for at least 12,000 years, it increased exponentially in the last 200 years (Sanderman et al., 2017). Before the invention of synthetic fertiliser in the early 20th century, it was common practice among farmers to maintain the fertility of soils by returning nutrients through crop residues and manure and preventing soil erosion. Traces of ancient farming systems, which were sustainable for hundreds and even thousands of years, can be found on all continents (Olsson et al., 2019). In modern agriculture, farming practices to maintain the natural fertility of soils have become less common as the increasing availability of artificial fertiliser allows to replenish depleted soil nutrients (Erisman et al., 2008). However, availability of fertiliser is spread unevenly around the globe. The depletion of soil nutrients in sub-Saharan Africa due to decades of farming without adequate fertiliser and manure is a primary factor limiting current food production (Sánchez, 2010; Folberth et al., 2013). Whilst access to fertiliser is an important requirement for the intensification of agriculture, leaching and runoff of excess fertiliser cause various negative impacts around cropland, which is a common cause of deteriorating environmental quality in developed regions (Montanarella et al., 2016). Moreover, despite the possibilities to replenish soil nutrients, increasing field management intensity is associated with deterioration in important soil properties and supporting ecosystem services (Squire et al., 2015).

The threat of land degradation for both humans and the environment has been recognised by many governments and international organizations. According to the *Science-Policy Platform on Biodiversity and Ecosystem Services* (IPBES) land degradation has negative impacts on at least 3.2 billion people worldwide leading to various socio-economic implications, such as migration, and the situation is expected to worsen due to climate change (Montanarella et al., 2018). Land degradation is addressed in the *Sustainable Development Goals* (SDGs), which is a roadmap by the *United Nations* (UN) to move

towards a more sustainable use of the planet's resources (Keesstra et al., 2018). Goal 15 is of direct relevance to land degradation, with the objective to protect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat desertification and halt and reverse land degradation and halt biodiversity loss (Olsson et al., 2019). SDG 15.3 addresses Land Degradation Neutrality (LDN), which is related to land restoration and halting land degradation. The concept of LDN was introduced by the United Nations Convention to Combat Desertification (UNCCD) and is defined as 'a state whereby the amount and quality of land resources necessary to support ecosystem functions and services and enhance food security remain stable or increase within specified temporal and spatial scales and ecosystems' (Cowie et al., 2018). Pursuit of LDN requires effort to avoid further net loss of the land-based natural capital relative to a reference state, or baseline (Olsson et al., 2019). To reduce land degradation on arable land, several policy programs encourage sustainable land management practices. The EU's Common Agricultural Policy (CAP) ties agricultural subsidies to implementation of best management practices and environmental protection, for example, by providing crop insurance for lower fertiliser application rates (Smith et al., 2016). The post-2020 CAP includes as one of its main objectives efficient soil management, combined with measures to reduce soil erosion and increase soil organic carbon in European agricultural soils (Sartori et al., 2019). In the USA, the Farm Bill extends soil conservation compliance requirements in order to qualify for crop insurance subsidy (Islam and Reeder, 2014). Other examples include the USA's Conservation Reserve Program (set aside marginal lands, steep slopes); China's Conservation Tillage Program; and several policies in Africa for integrated land management to help protect vulnerable soils (Smith et al., 2016).

1.3 Land degradation by water erosion

Whilst many land degradation processes influence agriculture, water erosion is widely considered as the most serious threat to global soil resources on arable land and agricultural productivity (Den Biggelaar et al., 2004a; FAO and ITPS, 2015; Kaiser, 2004; Panagos et al., 2018; UNCCD, 2017).

Water erosion and wind erosion are combined under the term soil erosion, which describes the removal, transport and deposition of soil particles by water or wind from their original location to another site. As such, soil erosion impacts the residence time, flux rates, storage and distribution of important elements in soils that sustain the life-support system of the Earth, including carbon (C), nitrogen (N) and phosphorus (P) (Cherlet et al., 2018). The removal of topsoil can reduce crop yields due to the loss of nutrients in the topsoil, degradation of the soil structure, and decreasing plant-available water capacity (Våje et al., 2005). Soil erosion does not only affect the productivity of agriculture but also leads to negative off-site effects. The runoff of nutrients, pesticides and organic carbon, which are included in the topsoil, leads to the pollution of surface waters, eutrophication and the emission of greenhouse gases (Chappell et al., 2016; Ravishankara et al., 2009; Tilman et al., 2001; Vitousek et al., 1997). Other off-site effects caused by sediment runoff are the increased sedimentation of lakes and rivers affecting fisheries, wildlife habitats and biodiversity, increased risk of flooding, damage of recreational activities,

and destruction of infrastructure such as roads, railways, waterways and harbours (Clark, 1985; Pimentel et al., 1995; Telles et al., 2011).

The soil degradation assessment in the GLASOD project showed that the vast majority of global soil degradation originated from soil erosion through wind and water (Oldeman et al., 1991). The latter has been shown to be responsible for more than half of the world's soil degradation. In the report on the *Status of the World's Soil Resources*, water erosion was named along the loss of organic carbon and nutrient imbalances as the most severe threat to soil resources (FAO and ITPS, 2015). In the *Guidelines for Sustainable Soil Management*, soil erosion is named as the most significant threat to global soils and the ecosystems services they provide (FAO, 2017). The threat of soil erosion for society has been recognised by many policy-makers and thus soil erosion is specifically addressed in several environmental and agricultural policy programs addressing land degradation (Alewell et al., 2019; Panagos et al., 2018; Sartori et al., 2019).

Regions most vulnerable to water erosion are mountain regions, due to steep slopes, the tropics and subtropics, due to abundant high energy precipitation, and arid regions, where precipitation events are rare but often intense and the vegetation cover is sparse. Due to their environmental characteristics, South America, sub-Saharan Africa, South and East Asia have been identified as the most vulnerable regions to erosion on agricultural land by several prior studies (Borrelli et al., 2017; Pimentel et al., 1995).

Water erosion is a natural process, but agricultural intensification and expansion accelerates water erosion dramatically. It has been estimated that agricultural activities are responsible for 75% of global soil erosion (Pimentel, 2006). In several regions worldwide, agricultural expansion due to population pressures and increasing demands for food and fodder led to the cultivation of marginal land susceptible to water erosion (Turkelboom et al., 2008; Willcocks and Twomlow, 1993) or soils prone to degradation (Wildemeersch et al., 2015). Although regional examples have also shown that increases in crop production due to rising population do not necessarily lead to more soil erosion if the land is managed sustainably (Tiffen et al., 1994), recent estimates suggest that the area of water erosion has increased due to deforestation and expansion of cropland (Borrelli et al., 2017). However, its actual extent and impact on crop production and environmental quality is highly uncertain.

1.4 Processes leading to water erosion

Water erosion is controlled by factors influencing the detachment and the transport of soil particles including rainfall and water runoff, soil properties, topography and land use. Soil runoff due to water erosion is initiated as soon as the velocity of water flow is big enough to reach the energy for the detachment of soil material. Subsequently, the dislodged material is transported to another location. During strong rainfall events the detachment of soil material through the energy of raindrops and the dispersion of disintegrated soil particles can also cause sealing and crusting of the topsoil (Oldeman, 1994).

Soil runoff due to water erosion can be divided into different stages depending on its severity,

according to Roose (1996): The initial phase of water erosion is called sheet erosion, which describes the evenly distributed removal of thin layers of the topsoil by overland flow or raindrop impact. However, sheet erosion is hardly detectable on a yearly basis, since other processes such as the respiration of swelling clay soil from rewetting or land management practices like tillage can have a higher impact on annual variability of topsoil depth. Nevertheless, annual repetition of sheet erosion can contribute to the removal of most of the surface horizon in a few decades. With ongoing runoff, sheet erosion can combine with more concentrated water flow along rills and grooves and lead to linear erosion of the topsoil, which results into the removal of several cm of the topsoil. The formation of rills is controlled by the cohesive strength of the soil and the shear forces exerted on the soil (Merritt et al., 2003). If the shear stress in a rill is high enough water flow within a rill may detach significant amounts of soil. Concentrated linear water erosion can lead to the formation of gullies, which increase the loss of soil up to more than 50 cm. In contrast to rill erosion, the deep channels caused by gully erosion cannot be obliterated by cultivation (Merritt et al., 2003).

The properties of the soil material determine the susceptibility of the soil to water erosion. The different sizes and weight of soil particles lead to a selective removal of soil material through water (Roose, 1996). As a result, erosion can lead to skeletonization of the soil surface as light material such as organic matter is removed, whereas heavier material such as sand and gravel is left behind. The selection process by water flow was described by Hjulstrom (1935), who studied the impact of water flow velocity and soil grain diameter on deposition, transport and erosion of soil material. With increasing diameter of the soil particle its weight increases, and thus more water flow velocity is necessary to gain the energy for the deposition and transport of the soil particle. Hence, coarse material such as gravel is less fragile to erosion than lighter material like fine sand. However, some material even lighter than sand such as clay is less susceptible to removal through water and erosion due to its cohesive forces. Cohesive soil materials have the properties to stick together and thus more energy is needed to detach them from the surface.

Another important property of soils is their infiltration capacity, which is determined by the density of soil particles and the available pore space. With increasing infiltration, the amount of water on the soil surface will be reduced resulting into less runoff. Thus, runoff will increase on soils, which are very dense on the surface and subsequently more erosion can occur (Favis-Mortlock and Mullan, 2011). The density of the soil surface can be modified by unsustainable land management with the use of heavy machinery or high livestock density.

The slope gradient of a field is a key driver for the transformation of rainfall into runoff and thus water erosion (García-Ruiz et al., 2015). With increasing slope steepness, the infiltration rate of the soil decreases, which increases the amount of water runoff. In addition, increasing slope steepness increases the volume and speed of water runoff. Similarly, the length of a slope determines volume and speed of water runoff, but relationships between slope length and runoff energy and capacity can be variable. Although common empirical erosion models suggest an increase in erosion with increasing slope length (Liu et al., 2000), experiments found that shortening slope length does not always reduce soil loss as

soil properties and field management can alter the length effect on soil loss (Bagarello and Ferro, 2010; Loch, 1996).

Soil surface roughness and soil cover through vegetation and organic matter have a strong impact on water erosion. A protective layer of plants and organic matter can protect bare soils from incoming rainfall and thus reduce erosion. Moreover, a dense vegetation cover can reduce water erosion by increasing the soils' infiltration capacity (Istanbulluoglu and Bras, 2006). In addition, soil coverage increases the surface roughness and thus reduces water runoff. The latter can be also influenced through tillage management. Generally, the impact of soil roughness and cover on water erosion on cropland is largely determined by field management.

1.5 The impact of agriculture on water erosion

Naturally, the susceptibility of soil to water erosion depends on the degree of its exposure to water energy. Agriculture commonly increases the exposure of soil to water energy due to a decline in the natural vegetation cover. Other forms of land degradation caused by agriculture, such as soil compaction due to intensive land use by heavy machinery or high livestock density, can exacerbate the erosion process (Fleige and Horn, 2000). Due to the influence of humans on vegetation cover, soil roughness and soil permeability, the exposure of soil to water energy has increased in many areas. A review of soil removal rates from different landscapes illustrates that highest erosion rates are found on arable land (Montgomery, 2007). The same study also found substantially lower erosion rates in fields where soil conservation methods are applied in comparison to conventionally managed fields.

Land use decisions by farmers can both increase and decrease the degree of water erosion. More specifically, the control of soil cover and topsoil characteristics through crop, residue and tillage management enables farmers to influence the exposure of soil to water erosion. Usually, when growing row crops such as maize and soybeans, the soil surface between the crop rows is bare, which increases the vulnerability of soil to water energy (Kort et al., 1998). Monocultures often leave the soil bare during fallow periods in the winter or in the spring before the vegetation cover of crops is dense enough to protect the soil from incoming rain. Moreover, organic matter, which stabilises the soil structure, is usually low in intensive cultivated fields as heavy fertiliser use reduces its significance to sustain soil fertility (Corsi et al., 2012). Bare soils in general are very susceptible to water energy and should be avoided on croplands vulnerable to water erosion. During the fallow period cover crops such as grasses or legumes can be planted to add protection to the soil, and to fixate nutrients and sequester carbon (Alvarez et al., 2017; Corsi et al., 2012; Poeplau and Don, 2015). Furthermore, row crops can be combined with strips of other crops or weeds in order to increase the density of the vegetation cover and thus prevent erosion (Claassen et al., 2018).

Tillage refers to the mechanical treatment of the soil in order to prepare the land for growing crops, which affects the soil characteristics such as soil moisture, soil temperature, infiltration and evapotranspiration processes (Busari et al., 2015). Tillage can be performed in different ways depending on how intensive the topsoil is manipulated. Conventional tillage is the most intensive form of soil disturbance using agricultural machinery such as harrows and ploughs in order to mix harvest residues, organic matter and nutrients in the topsoil and break down soil aggregates to prepare the seedbed for crop planting (Claassen et al., 2018). Intensive tillage can accelerate erosion by breaking down soil aggregates through heavy machinery, which decreases the robustness of the soil (Zheng et al., 2018). In the mid-20th century, awareness that conventional tillage methods accelerate erosion increased and thus reduced tillage and no-tillage methods were tested in agriculture (Montgomery, 2007). With reduced tillage strategies, often more harvest residues are left on the field, which increases the surface roughness and thus slows water flow and soil erosion. The lower exposure of soils, as a result of more residues left on the field, reduces the susceptibility of the topsoil to erosion (Sharratt et al., 2015). In addition, retention of crop residue protects the soil from direct impact of raindrops and sunlight (Busari et al., 2015). Furthermore, the lower the depth of tillage the less the soil becomes disturbed, which can help to preserve and enhance the soils productivity or even restore previously lost productivity of the soil (Corsi et al., 2012). Moreover, the minimisation of soil disturbance enhances soil biological activities, soil air and water movement (Busari et al., 2015).

As outlined in the previous chapter, environmental characteristics of a field such as soil properties and topography are important factors for the susceptibility of arable land to water erosion. Farmers can alter relevant field characteristics and thus influence water erosion. In hilly landscapes, adaptation measures to reduce the effect of slopes steepness on water erosion can be critical to maintain agricultural productivity. The most effective anti-erosion measure on slopes is terrace farming, which is the practice of creating steps of flat areas in hilly landscapes in order to prevent water runoff (Chen et al., 2020). Another often used anti-erosion measure on hilly cropland is ploughing along the contours of the slope to break the water flow and thus decrease erosion (Zhang et al., 2004).

1.6 The impact of water erosion on crop productivity

Water erosion affects a variety of soil functions relevant for crop growth such as nutrient levels, pH, water-holding capacity, texture, infiltration rates and soil organic matter (den Biggelaar et al., 2001). The most frequently reported causes of yield reductions as a result of water erosion are (i) nutrient deficit, (ii) water deficit, and (iii) the reduction of soil depth leading to a limitation of root growth by a clayey subsoil or by a pan or bedrock (Bakker et al., 2004).

The impact of erosion on root growth and crop yields increases with progressing erosion as root growth becomes increasingly more difficult for the plant. Thus, erosion may not affect crop yields until an advanced degradation of the soil layer, when continuous soil loss has reduced the soil depth to such an extent that the development of the root system becomes restricted. Therefore, in some cases the impact of erosion on crop yields may increase over time with equal erosion rates (Bakker et al., 2004).

Water deficits due to erosion are most likely during droughts. Water availability for crops is affected by (i) reduced soil water storage capacity due to decreased soil depth, (ii) reduced soil water holding capacity due to loss of soil organic matter and soil structure, and (iii) reduced soil moisture availability due to the transport of clayey soil material to the surface (Bakker et al., 2004).

Important plant nutrients such as nitrogen, phosphorus and potassium are most abundant in the topsoil and thus are susceptible to surface runoff. Organic matter is an important natural component for soil fertility, which can be found in a high concentration near the soil surface in form of decaying leaves and stems. The high susceptibility of soil organic matter availability by soil erosion is demonstrated by studies, which show that removed soil through wind and water is richer in organic matter than the soil left behind (Pimentel et al., 1995; Lal, 2002). As eroded soil is often deposited at another field site, soil is often not lost but moved from one part of the landscape to another (Duan et al., 2016). However, the deposition of clay on the topsoil due to erosion can also lead to a reduced nutrient availability as clay particles absorb nutrients (Bakker et al., 2004).

Soil nutrients lost through erosion can be replaced by nutrients from synthetic or organic fertiliser. Thus, the amount of fertiliser applied to a field influences the impact of water erosion on crop yields. One study demonstrates that soils with severe erosion may produce 15-30% lower maize yields than uneroded soils, whereas the reduction ranges from 13 to 19% with fertilisation (Pimentel et al., 1995). Moreover, the timing of fertiliser application in respect to plant growth stages or rainfall events can be important in the reduction of water erosion impacts (Luo et al., 2020). The replacement of soil nutrients with fertiliser can lead to significant economic and environmental costs.

1.7 The impact of climate change on water erosion

Whilst many studies focus on the impact of climate change on crops, the impacts of climate change on soil resources are hardly explored (Montanarella et al., 2016). In a recent special report on *Climate Change and Land* by *The Intergovernmental Panel on Climate Change* (IPCC), the authors stated that "climate change exacerbates the rate and magnitude of several ongoing land degradation processes and introduces new degradation patterns (high confidence)" (Olsson et al., 2019). Water erosion was named as one of the land degradation processes most directly linked to climate change due to a projected increased frequency, intensity and/or amount of heavy precipitation. In addition, changes in vegetation cover and soil properties due to climate change may increase the risk of water erosion (Li and Fang, 2016). Consequently, the loss of soil productivity through water erosion and other land degradation processes in combination with climate change can multiply threats for natural resource-based livelihoods (Webb et al., 2017).

The impact of climate change on water erosion is the result of a complex web of relationships between climate, biomass, soil resources and the way farmers manage their fields (Figure 1.2). Direct climate change impacts on water erosion are mainly determined by changes in precipitation amount and intensity (Li and Fang, 2016; Nearing et al., 2004). Increasing rainfall amounts can lead to an increase in soil moisture and saturation of the soil generating more water runoff and soil loss. Excess runoff due to soil saturation is common after a long period of low-intensity rainfall, which is often the case in



Figure 1.2: The main links between climate change, erosion, crops and field management. The diagram illustrates how climate change affects the interactions between crops and water erosion, and what opportunities farmers have to influence these relationships. The plus and minus signs indicate a positive or negative correlation between the respective components.

humid areas (Smith and Goodrich, 2006). Additional soil removal in saturated soils can be caused by subsurface runoff transporting soil particles underground (Li and Fang, 2016).

Changes in rainfall intensity generally affect rainfall energy during storm events, which directly affects the detachment of soil particles via the impact of raindrops (Van Dijk et al., 2002). Moreover, highly intense rainfall often exceeds the soils infiltration capacity and thus can cause excess runoff and a much greater transport of suspended sediment load (Mohamadi and Kavian, 2015). These events are most common in semi-arid and sub-humid zones and were observed in Beijing, the Mediterranean, Iran, France, and Western USA (Li and Fang, 2016). As changes in rainfall amount and rainfall intensity have different effects on soils, changes in the degree of both parameters together are likely to have a greater impact than changes in both separately (Nearing et al., 2004).

Rainfall also influences water erosion indirectly by controlling the availability of water supply for plants and thus the development and distribution of the soil's vegetation cover. Other indirect impacts of precipitation on water erosion are linked to the impact of land use on water erosion, which can be affected through spatio-temporal shifts in precipitation, which may influence planting dates and the length of the

growing season. The timing of rainfall events also may have significant impacts on changes in wetting and drying of soils and thus their infiltration capacity affecting water erosion (Lado et al., 2004). Shifts in precipitation patterns can also directly affect water erosion if stronger precipitation is occurring at the beginning of the growing season or winter season, in which soils are less protected by dense vegetation covers (O'Neal et al., 2005; Leek and Olsen, 2000; Mullan et al., 2012).

The indirect impact of climate change on water erosion through changes in the protective vegetation cover are strongly influenced by temperature changes and water availability for plants. Projected temperature changes can both increase and decrease the vegetation cover and thus influence water erosion rates positively and negatively (O'Neal et al., 2005). Moreover, higher temperature can also reduce the time soils are protected through crop residues, by increasing microbial activity and their decomposition rates, which can be further accelerated by increases in soil moisture (Nearing et al., 2004). Higher temperatures can reduce soil protection through snow cover and increase the ratio of rain to snow and snowmelt (Li and Fang, 2016). Changes in temperature affect evapotranspiration, which influences soil moisture and infiltration capacity and thus the partitioning of water into surface and subsurface runoff (Nearing et al., 2004). Changes in solar radiation have similar indirect effects on water erosion as temperature changes, since solar radiation also affects plant growth, evapotranspiration, soil moisture and other parameters influenced by surface temperature.

Climate change can reduce water erosion most directly through decreasing precipitation amounts and intensity (Pruski and Nearing, 2002a). In addition, increased vegetation growth due to changing precipitation, temperature, solar radiation and CO_2 fertilisation can reduce water erosion (Chaplot, 2007; Olsson et al., 2019). Moreover, temperature increases can decrease soil moisture and thus increase the soil's water infiltration capacity.

Simultaneous positive and negative impacts of climate change on water erosion complicate research on climate change impacts on water erosion. For example, increases in temperature and soil moisture can lead to increased plant growth and thus soil cover, but also faster rates of residue decomposition due to an increased microbial activity and thus reduced soil cover (Nearing et al., 2004). Decreasing precipitation amounts can lead to lower water runoff but may also decrease the soils vegetation cover and thus the protection against water erosion (Pruski and Nearing, 2002a). Anthropogenic or socio-economic factors add further complexity to climate change impacts on water erosion. Climate change may alter field management such as planting and harvesting dates, type of cultivars, crop rotation and tillage techniques, which can have a significant impact on soil cover and thus water erosion (Nearing et al., 2004). Although climate change impacts on water erosion can be very variable, numerous studies agreed that increasing precipitation leads to increasing water erosion (Li and Fang, 2016). Where precipitation decreases, changes in interactions of plant biomass, runoff, and water erosion are more complex (Pruski and Nearing, 2002a).

Due to their complexity, the consequences of climate change on water erosion have not been investigated in depth. Thus, the impact of climate change on soil functions has been considered to be the largest source of uncertainty in any projections of the trends in key ecosystem services provided by
the soil (Montanarella et al., 2016). As the quality of soil resources is one of the main determinants for agricultural productivity, further research on the links between climate change and soil degradation processes such as water erosion is necessary to improve the understanding of climate change impacts on agriculture.

1.8 The impact of climate change on crops

As the indirect effects of climate change on water erosion through changes in land use and vegetation cover are of great importance, the effects of climate change on crops are relevant for future water erosion.

Observed climate change is already affecting crop production through increasing temperatures, changing precipitation patterns, and greater frequency of some extreme events (Mbow et al., 2019). Regional effects of climate change are spread unevenly. Observations and projections of climate change impacts on the most important staple crops show that yields in many lower-latitude regions are affected negatively by climate changes, while in many higher-latitude regions, yields are affected positively (Mbow et al., 2019; Rosenzweig et al., 2014).

Increasing temperature can both increase and decrease crop yields, as these are determined by a minimum and maximum temperature within which the phenological development of a plant takes place, as well as by an optimum temperature at which the phenological phases are accelerated. The development of plants slows as temperature increases above the optimum and ceases when plants are exposed to their maximum temperature (Hatfield et al., 2011). As average temperatures in many tropical agricultural regions are already close to the high-temperature thresholds for suitable production of major staple crops, even moderate temperature increases in lower latitudes can have negative impacts on crop yields (Rosenzweig et al., 2014). Additionally, air temperature has an important impact on the amount of suitable days for a crop to grow. The time period of crop growth can be described in thermal time, which defines the time until a crop reaches maturity after a specific number of 'growing degree days' expressed by the sum of daily temperature values (Su et al., 2013). When daily temperatures increase, the rate of phenological development increases and the plant reaches maturity earlier, which results into less days where the plant can intercept light for biomass accumulation and thus yields will decrease (Ritchie and Nesmith, 1991). The definition of 'growing degree days' includes a base temperature below which crop growth does not continue. Consequently, temperature increase can lead to more days suitable for crop growth and thus create new potential regions where certain crops can grow (Zabel et al., 2014).

Crop cultivation systems are closely linked to hydrological systems through the availability of water resources, which are strongly dependent on precipitation and evapotranspiration patterns controlled by the climate system. It is rare that crop growth is limited solely based on heat stress due to high temperature. Crop damage due to increased temperature is rather connected to water stress for crops caused by increased evapotranspiration, which leads to a reduced availability of soil moisture (Carter et al., 2016). In addition, too much water as a result of increasing extreme rainfall events can also prevent or decrease crop growth. Increasing strong rainfall can wash away important nutrients from the topsoil for the crop, which limits its growth (Burt et al., 2016). Furthermore, increasing rainfall intensity can lead to waterlogging or flooding of the soil, which can damage crops due to a lack of oxygen and salinisation (Saraphirom et al., 2013; Rosenzweig et al., 2002). Therefore, changes in precipitation amounts, intensity and patterns can affect crops both positively and negatively.

Water availability for a crop can be also influenced through the impact of solar radiation on evapotranspiration and thus, changes in the degree of solar radiation affect crop growth (Carter et al., 2016). An increase in solar radiation will increase the amount of energy to vaporize water, which results into a loss of water for the plant. Furthermore, the degree of solar radiation influences crop growth through the availability of light, which is an important component for the plant to gain energy during photosynthesis. Therefore, increasing cloud cover will decrease the amount of solar radiation and thus decrease the productivity of the plant.

The effect of increasing atmospheric CO_2 on crop growth is largely positive. As crops use CO_2 as an input to gain energy during photosynthesis, an increase in the concentration of CO_2 in the atmosphere results in an increase of photosynthesis when other growth factors such as water and light are not limited. Thus, atmospheric CO_2 can have a fertilising effect for crops. Importantly, this effect is different in plants that differ in their method of carbon fixation during photosynthesis. Experiments have shown that C3 crops such as wheat increased significantly with increasing CO_2 due to increasing photosynthetic activity (Slingo et al., 2005). Crop growth of C4 crops such as maize may not response directly to increasing atmospheric CO_2 , but yields might increase indirectly through increasing water use efficiency via reduction in stomatal conductance (Long et al., 2006b). However, recent studies highlight that although increasing CO_2 affects crop growth positively, it may lower concentrations of important nutrients in crops (Myers et al., 2014). More research is needed to fully understand the effect of increasing atmospheric CO_2 on crops and the impact on human nutrition.

Whilst changes in temperature, precipitation, solar radiation and atmospheric CO_2 mainly determine the impact of climate change on crops, changes in other climate parameters such as wind speed, relative humidity, and atmospheric ozone also influence crops, but are less explored (White et al., 2011). Moreover, climate change impact on crops are usually complex and are often the result of a combination of several stress factors. For example, in Malawi, crops drying before maturity and damage due to floods and watershortages led to food shortages in the 1990s and early 2000s (Mkwambisi et al., 2010). In Swaziland, changing rainfall patterns led to the spread of the parasitic weed *Striga asiatica*, which reduced maize production (Stringer et al., 2007). Excessively wet soil can reduce crop yields through an increased risk of plant disease and insect infestation or by delaying planting or harvesting due to inability to operate machinery (Rosenzweig et al., 2002).

1.9 The use of crop models in impact assessments

Given that humans have depended on agriculture for more than 10,000 years, the impact of environmental change and land management on agricultural production have long been a focus of science. Since the 1960s, numerical crop models have been used to simulate the complex biological and physical interactions in agricultural systems to understand the impact of environmental drivers and farming techniques on crop yields (De Wit, 1965; Monteith, 1965). With the emergence of global-scale phenomena that affect agricultural productivity (most prominently climate change), crop models are increasingly being applied at the global scale to explore regional variabilities of crop yield changes in response to environmental factors and field management (Mueller et al., 2017). In addition, crop model outputs frequently contribute to integrated assessment modelling studies. In these studies, models from different disciplines are combined to address complex global problems, such as the growing population and its demands for more food, water, and energy in the face of climate change, the limited arable land for expanding food production, and increasing pressures on natural resources and biodiversity (Jones et al., 2016). Moreover, crop models are a transparent and consistent tool to inform agricultural and environmental policies, which rely on large-scale indicators to represent administrative units ranging from regional to national scale.

As such, crop models are an important tool for researching complex global problems. However, not all biophysical interactions in crop cultivation systems are understood and thus not all processes relevant for crops are considered in crop models. Moreover, the complexity of modelling crop cultivation systems increases with the spatial focus of a study. Crop modelling studies that assess the impact of environmental change or management scenarios on agricultural productivity at the regional or global scale often neglect land degradation processes (Balkovič et al., 2018). Thus, such studies neglect a crucial driver of long-term agronomic trends. In addition, there is a growing need for large-scale indicators on land degradation impacts and patterns to inform policy programs (Montanarella et al., 2016). As water erosion can produce high costs for farmers and negatively affect infrastructure and environmental quality, many agricultural and environmental policy programs aim to reduce water erosion (Alewell et al., 2019). Models can help to analyse the effectiveness of field management scenarios to reduce water erosion and to project future water erosion rates and impacts under land use and climate change scenarios (Borrelli et al., 2020). Whilst there are currently efforts to improve models to monitor water erosion rates over large scales (Borrelli et al., 2021), more research is necessary to understand interactions between agriculture and water erosion and their impact on current and future crop production in different global regions. As agriculture influences human well-being and interacts with environmental quality, understanding these relationships and how they are influenced by regional characteristics, will be crucial to solve many current and future global challenges.

1.10 Thesis aims and objectives

The aim of this thesis is to better understand the effects of water erosion on current and future grain yields. In addition, this thesis aims to improve the representation of the interactions between water erosion and crops in large-scale crop modelling studies to improve crop yield projections and water erosion assessments. The global gridded crop model (GGCM) EPIC-IIASA is used for the analysis.

This model is used to generate large-scale and global crop yield projections to assess the impact of field management and climate change scenarios on crop production. Maize and wheat are used to represent cereal crops affected by water erosion.

The three main objectives of this thesis are to:

- 1. Analyse the robustness of water erosion rates simulated with EPIC-IIASA on global maize and wheat croplands.
- 2. Explore the impact of water erosion on global maize and wheat yields.
- 3. Explore the combined effects of climate change and water erosion on global maize and wheat yields.

The first objective aims to develop and test a modelling approach for estimating water erosion rates on global maize and wheat cultivation areas based on various field management scenarios. This includes identifying the most sensitive model input parameters controlling the simulation of water erosion, quantifying the uncertainty of simulated water erosion rates stemming from different field management scenarios and water erosion estimation methods, and evaluating the agreement of simulated water erosion rates with field data across different environments. The modelling approach evaluated in the first part of this thesis will be used for the second objective to simulate the impact of water erosion on global maize and wheat yields using historic climate data. The model outputs will be analysed to explore global differences in water erosion impacts on maize and wheat yields due to different environmental characteristics and field management. To address the third objective, the previous modelling approach will be applied with contrasting climate change scenarios. Based on the model outputs, regions where maize and wheat yields are susceptible to both climate change and water erosion will be identified, and the impact of climate change on water erosion will be analysed.

1.11 Thesis overview

The thesis is structured as follows. Chapter 2 reviews the literature on water erosion assessments. It highlights the advantages and disadvantages of common methods to assess water erosion patterns and impacts. It identifies simple water erosion models as the most suitable method for large-scale water erosion assessments. Further, it describes knowledge gaps on current and future water erosion patterns and impacts that can be addressed using the biophysical crop modelling framework developed in this study.

Chapter 3 reviews the literature on crop models. It provides an overview of the types of crop models and their applications, and presents the results of previous modelling studies on the impacts of climate change on crops, including their uncertainties. It finds that soil processes are often neglected in crop yield projections, although they can significantly influence the impact of climate change on crops and the effectiveness of climate change adaptation strategies.

Chapter 4 introduces the modelling framework of this study. It describes the relevant equations

used in the EPIC model to simulate the interactions between crops and water erosion. Further, it lists the model inputs used in this thesis and the methods for aggregating, analysing and evaluating the model outputs.

Chapter 5 - 7 present and discuss the results of this thesis. In Chapter 5 the modelling framework of this thesis is evaluated by presenting and discussing the uncertainties, sensitivities and robustness of the simulated water erosion rates using EPIC-IIASA. Chapter 6 looks at the model results on the impact of water erosion on global maize and wheat yields. It analyses the major determinants influencing water erosion impacts on maize and wheat yields and draws conclusions on the role of water erosion impacts for the global production of both crops. Chapter 7 analyses the combined effects of climate change and water erosion on global maize and wheat yields. It identifies regions where maize and wheat production is susceptible to both climate change and water erosion and analyses potential changes in future water erosion due to climate change.

Chapter 8 summarises the thesis results and draws conclusions on the contributions of this thesis to global water erosion assessments and crop yield projections. It also considers the future research needs to improve estimates of large-scale water erosion and to account for soil degradation in global crop yield projections.

Chapter 2

Literature review of water erosion assessments

2.1 Introduction

Information on the state of soils is necessary to inform agricultural and environmental policy programs, such as the European Unions's Common Agricultural Policy (CAP) (Keesstra et al., 2016; Panagos et al., 2016). In addition, reducing soil erosion is an important matter for achieving several SDGs (Keesstra et al., 2016). However, Humphries and Brazier (2018) believe that soil-related policies are currently secondary, being the result of other environmental objectives. One important reason for this is the difficulty of assessing the state of soils on a scale that is relevant to policy. The difficulty of this task is shown by the various attempts to map global land degradation, which led to very different results.

Among the many land degradation processes that are commonly assessed, recent efforts have been made to improve the assessment of water erosion at regional and global scales using models (Borrelli et al., 2021). The recent studies have shown that modelling tools can provide information on the status of soil under different land use and can test the impacts of external factors such as climate and management on soils on policy-relevant spatial and temporal scales. According to Alewell et al. (2019), the recent modelling efforts have shown that large-scale modelling is inevitable to increase the social and political visibility of soil resource issues.

Despite the recent improvements, large-scale assessments on water erosion rates on arable land are still rare or do not provide detailed information on varying water erosion rates for different crops and field management strategies. This lack of information on water erosion also makes it difficult to quantify its impact on crop yields and pollution around croplands on a large scale, and how this might change in the future due to climate change. There is also a need for large-scale estimates of water erosion in other soil-related research areas. For example, earth science system models, climate change and carbon mitigation scenarios, and hydrology and flood forecasting models could be improved if soil erosion were considered (Alewell et al., 2019).

This literature review presents existing global assessments of land degradation to illustrate the complexity of evaluating the current state of soil resources on a large scale. Moreover, these assessments highlight the role of water erosion in global soil degradation. Subsequently, this literature review will present the latest regional and global water erosion assessments and the advantages and disadvantages of different methods to explore water erosion rates and impacts. Finally, this literature review presents the current state of knowledge as well as knowledge gaps regarding the impacts of climate change on water erosion.

2.2 Global land degradation assessments

Several global assessments suggest that large parts of the earth surface are already affected by land degradation. However, the degree of global land degradation named by a study strongly depends on the indicators used to describe land degradation. Commonly used methods to analyse global land degradation are remote sensing, biophysical models, inventories of land use/condition and expert opinions (Gibbs and Salmon, 2015). Due to the complexity of land degradation a complete assessment cannot be performed with a single method, but a combination of studies and methods can give an approximation (Vogt et al., 2011). Various indicators exist to measure the state of soil, vegetation and biomass or potential land degradation under specific land use and field management (Olsson et al., 2019). It is important to distinguish between studies monitoring the current state of land degradation and studies analysing potential land degradation. Studies focusing on land degradation processes often focus on their extent rather than quantifying their impact, e.g. changes in crop yields due to land degradation.

The Global Assessment of Human-induced Soil Degradation (GLASOD) carried out by the World Soil Information Center (ISRIC) and the United Nations Environment Program (UNEP) from 1987-1990 was the first major global assessment of land degradation focusing on soil resources (Oldeman et al., 1991). As that study focused on soil degradation, the loss of vegetation and biodiversity were neglected in GLASOD. In that project, the degree of soil erosion by wind and water, chemical soil deterioration, and physical soil deterioration was assessed based on expert judgement by soil scientists on loosely defined physiographic units on a global scale. The outcome of that study estimated that 38% of global land was lightly affected by land degradation made by humans, 46% moderately and 15% of global land was affected in a high degree. The most important degradation process was erosion by water and wind, which accounted for 84% of degradation. The most degraded area was Europe (25%), followed by Asia (18%) and Africa (18%) (Nkonya et al., 2011).

The most comprehensive assessment of global land degradation was conducted by the *Food and Agriculture Organization of the United Nations* (FAO) in collaboration with the *International Institute for Applied Systems Analysis* (IIASA) presented in the *Global Land Degradation Information System* (GLADIS). GLADIS gives insights into the global status of land resources and degradation pressures acting on them derived from a combination of biophysical and socioeconomic determinants (Nachtergaele et al., 2011). GLADIS provides classes of land degradation based solely on biophysical indicators,

which assess the status and degradation of soil, water, biomass and biodiversity. Moreover, the assessment includes past and present trends and thus recognises causes and impacts due to biophysical and socio-economic factors, and includes a prognosis of future land degradation trends. The results show that 32% of global land is in areas with high provision of biophysical goods and services, but with medium to strong degradation processes. Further, 27% of the global population live in areas with a low provision of biophysical goods and services and a medium to strong degradation (Nachtergaele et al., 2011). The special definition of the land degradation index adding social and economic indicators to biophysical factors gives a more comprehensive insight into the impact of land degradation in certain regions and thus provides a wide range of information. As such, the assessment also indicates that 42% of the very poor live in degraded areas, as compared with 32% of the moderately poor and 15% of the non-poor (Nkonya et al., 2011).

In the report on the *Status of the World's Soil Resources* by *The Intergovernmental Technical Panel* on *Soils* (ITPS) about 200 soil scientists reviewed the scientific knowledge from more than 2000 peerreviewed scientific publications to provide a global perspective on the current state of the soil, its role in providing ecosystem services, and the threats to its continued contribution to these services (FAO and ITPS, 2015). The land degradation processes considered in the report are erosion, compaction, acidification, contamination, sealing, salinisation, waterlogging, nutrient imbalance (i.e., both nutrient deficiency and nutrient excess), and losses of soil organic carbon and of biodiversity. Soil erosion has been identified as the most severe land degradation process leading to deteriorating water quality in developed regions and to lowering of crop yields in many developing regions (Montanarella et al., 2016). The authors concluded that nitrogen and phosphorus fertiliser use in infertile tropical and semi-tropical soils – the regions where the most food insecurity among us are found – needs to be increased, while global use of these products needs to be reduced.

The most recent assessment of global land degradation was carried out by Pravalie et al. (2021) based on a collection of geospatial data from previous studies on aridity, water erosion, vegetation decline, soil salinisation and soil organic carbon decline. The authors concluded that aridity is by far the largest singular pressure for agricultural systems, affecting 40% of the arable lands' area, and that water erosion is the second major degradation process, affecting 20% of global arable systems. However, Pravalie et al. (2021) only considered water erosion rates larger than 10 tha^{-1} , which have been considered as not sustainable to maintain agricultural productivity. Moreover, aridity is usually not considered in land degradation assessments, as dryness is a climatic impact rather than a land degradation processes. Pravalie et al. (2021) concluded that simultaneous pressure of multiple land degradation processes is rare. According to their study, the most common combination of aridity and water erosion affect 7% of cropland.

Despite several attempts at mapping the spatial and temporal scale of degraded land, it remains a challenging task due to the difficulty of defining land degradation, determining baselines or measuring relevant processes (Herrick et al., 2019). Therefore, views on assessing land degradation are diverging and thus no reliable global maps of the extent and severity of land degradation exist. The most promi-

nent estimates of the global extent of land degradation are very variable, ranging from 1–6 billion ha of ice-free land surface (up to 66%) (Gibbs and Salmon, 2015). To support the goal of governments and international organisations of reducing land degradation, it is necessary to improve indicators and baseline datasets of land degradation at global scale to better monitor progress (Pravalie et al., 2021).

2.3 Water erosion assessments

2.3.1 Large-scale water erosion assessments

First attempts to address water erosion at the global scale were made in the 1930s (Jacks and Whyte, 1939). Since then, many studies estimated the amount of soil loss due to water erosion worldwide (García-Ruiz et al., 2015). The majority of erosion studies is concentrated in North America and Europe. The earliest estimates of global soil loss are based on extrapolation of erosion plot data to the global scale or they are derived from sediment yields from rivers (Myers, 1993; Pimentel et al., 1995; Walling and Webb, 1983). More recent global estimates are derived from models (Borrelli et al., 2017; Doetterl et al., 2012; Van Oost et al., 2007). Doetterl et al. (2012) summarised existing estimated agricultural soil loss amounts on a global scale based on both field measurements and model outputs, which range between 24 and 120 Pg yr⁻¹. The wide range of estimates indicates considerable uncertainty in global water erosion estimates.

Large-scale water erosion assessments based on field experiments

As models have only recently become the preferred method for large-scale water erosion assessments, most earlier assessments have been based on extrapolation from field experiments, contributing to the majority of available regional and global water erosion estimates. The most cited value of global soil loss on arable land is 75 Pg yr⁻¹, which was estimated by Pimentel et al. (1995) based on plot measurements, but has been criticised repeatedly due to the uncertainty of extrapolating plot measurements to larger scales (Crosson, 1995). Cerdan et al. (2010) reviewed erosion rates from 81 experimental sites in 19 European countries and estimated a mean annual erosion rate on arable land in Europe of 3.6 t ha⁻¹, although aggregation of the data by land use showed large variations. Boardman (1998) questioned the usefulness of an average rate of soil erosion for Europe or other continents as the rates vary too much in time and space. Further, he suggested that plot experiments are a poor basis for regional generalisation as they only represent data from a very small area.

Although the usefulness of field data for large-scale water erosion assessments has been criticised, they are often used to assess the severity of regional soil erosion problems. Several previous researchers used a collection of field measurements of soil erosion rates to compare them to tolerable soil erosion rates (Verheijen et al., 2009; Benaud et al., 2020; Montgomery, 2007). Montgomery (2007) collected field data worldwide derived from various field measurement methods and estimated median and average water erosion rates of 1.5 and 3.9 mm yr⁻¹ on cropland under conventional agriculture. Assuming a soil bulk density of 1200 kg m⁻³ these values equal to median and average water erosion rates of

18 and 47 t ha⁻¹. These rates exceed suggested tolerable rates of 1 t ha⁻¹ (Benaud et al., 2020) and previous estimates of average annual soil formation rates of 0.3 t ha⁻¹ – 1.4 t ha⁻¹ (Evans et al., 2019; Montgomery, 2007; Verheijen et al., 2009). Even the more generous tolerable soil loss rates defined by the U.S. Department of Agriculture (USDA) of up to 11 t ha⁻¹ (Schertz and Nearing, 2006) are below the values derived by Montgomery (2007).

Benaud et al. (2020) collected soil erosion measurements across the UK derived from different field experiment methods and concluded that mean erosion rates are low, while 16% of observations exceed soil loss rates of 1 t ha⁻¹. Although Benaud et al. (2020) collected a large amount of data from across the UK, they concluded that understanding whether or not the UK has a soil erosion problem remains unclear as i) existing studies are skewed towards locations with a known erosion likelihood and ii) the definition of a tolerable erosion rate remains problematic in the absence of a comprehensive and methodological consistent national-scale study of historic and current soil erosion rates.

A meta-analysis of more than 4000 erosion measurement sites by García-Ruiz et al. (2015) showed an extraordinarily high variability in erosion rates, with almost any rate possible irrespective of slope, climate, scale, land use/land cover and other environmental characteristics. Despite a slight positive correlation between erosion rates and slope and annual precipitation, there is much variability in erosion measurements that is at least partially related to the experimental conditions. In addition, Cerdan et al. (2006) noted that large-scale studies based on data from experimental plots tend to estimate higher amounts of soil loss because the locations of erosion measurements are often biased towards bare soil cover, steep slopes or extreme precipitation events.

As such, the usefulness of field data to derive conclusions on large-scale erosion rates is questionable due to their uneven distribution and variable quality. Boardman (2006) concluded that 'for most areas in the world the erosion data is woefully inadequate'. Stroosnijder (2005) challenged the adequacy of erosion measurements and warned of a scientific and technical crisis because 'there are insufficient empirical data of adequate quality, a lack of funds to improve that situation, a lack of development of new technologies and equipment, and a lack of skilled personnel'. Benaud et al. (2020) concluded that 'existing monitoring approaches have been resource intensive in nature and/or biased towards a single erosion process, hindering their suitability for future, holistic national-scale monitoring programs.' The uneven distribution of erosion measurements for different global regions reduces their usefulness for robust global erosion estimates.

Large-scale water erosion assessments based on models

Models can be used as a tool to overcome the lack of data coverage to estimate erosion rates and account for the variability of soil, topography, climate and field management around the world. Model outputs overcome the lack of homogeneity in available data and thus can be useful to identify soil erosion hotspots determined by the combination of erosion controlling factors in a more transparent way.

The Universal Soil Loss Equation (USLE) and its modification have been used with high-resolution global input data to estimate water erosion (Borrelli et al., 2017; Doetterl et al., 2012; Van Oost et al.,

2007). These studies resulted in a range of annual global soil removal estimates and water erosion rates on cropland of between 13 - 22 Gt and 11 - 13 t ha⁻¹. These estimates are less variable and more closely compared to field measurements from various sources than earlier estimates, which suggests a higher robustness.

However, the use of models in large-scale erosion assessments has been criticised heavily (Evans and Boardman, 2016b). The most commonly used models for large-scale assessments are USLE-type models due to their simplicity. However, their use is criticised as they are based on plot data and thus not suitable for extrapolation to large fields. In addition, they were not developed for environmental conditions different than the United States' Midwest.

Nevertheless, due to their simplicity and the growing availability of global environmental datasets that can be used as sources for model input data, USLE-type models are widely seen as the most reliable tools to derive erosion estimates on large and global scales (Alewell et al., 2019). Moreover, unlike field observations, models can provide predictive estimates of water erosion and are therefore useful in land use and climate change impact assessments and scenario analysis.

2.3.2 Common water erosion modelling techniques

Common factors in models used to determine water erosion are land use, slope, precipitation amount and intensity, runoff and peak runoff rates, runoff shear stress, soil cohesion, and surface roughness (de Vente and Poesen, 2005). Most models focus on plot to catchment scale and simulate only a selection of erosion processes (sheet-, till-, ephemeral gully erosion), whereas permanent gullies, mass movements and riverbank erosion are most often not considered (De Vente et al., 2013). Furthermore, models focus either on simulating single erosion events or on assessing average erosion in hourly, daily, monthly or annual time-steps (Karydas et al., 2014). As such, water erosion models differ in complexity, data requirements and equations used to formalise processes and the outputs they provide (Abdelwahab et al., 2018). Therefore, the choice of model strongly depends on the data and calibration requirements, temporal and spatial scale and the erosion process considered (De Vente et al., 2013).

Most reviews classify water erosion models into empirical, conceptual and physics-based models (Merritt et al., 2003; de Vente and Poesen, 2005). Empirical water erosion models delineate a mathematical algorithm out of the relationship between the confounding measurable parameters, which control water erosion, and measured eroded sediments (Alewell et al., 2019). Empirical models are the simplest type of model and thus their computational and data requirements are usually less than for conceptual and physics-based models (Merritt et al., 2003). However, empirical models are often criticised for ignoring the inherent non-linearities in the catchment system (Wheater et al., 1993). The most widespread empirical water erosion model is the Universal Soil Loss Equation (USLE) (Wischmeier and Smith, 1978).

Conceptual models typically represent a catchment as a series of internal storages and usually incorporate the underlying transfer mechanisms of sediment and runoff generation in their structure through defined flow paths (Merritt et al., 2003). Thus, conceptual models include a general description of catchment processes, which distinguishes them from empirical models (Sorooshian, 1991). Often used conceptual models are SWAT (Neitsch et al., 2011) or ANSWERS (Beasley et al., 1980).

Physics-based models such as the WEPP (Lane et al., 1992) and the EUROSEM (Morgan et al., 1998) include physical equations describing streamflow and sediment and associated nutrient generation in a catchment (Merritt et al., 2003). Although parameters used in physical-based models are measurable, they usually need to be calibrated against observed data due to their large number and the heterogeneity of important catchment characteristics (Merritt et al., 2003). Although most reviews on water erosion models classify models into physics-based models, conceptual models, and empirical models, most models do not fall strictly into one category, as even the physics-based models retain some empirical elements in the model algorithms (de Vente and Poesen, 2005).

The complexity of a model can be further distinguished between spatially distributed models and lumped models. Spatially distributed models reflect the spatial variability of processes and outputs and thus are capable to produce erosion patterns in a single run, whereas lumped models only produce a single value in a single run (Karydas et al., 2014). The WEPP and EUROSEM models are polygon-based models adapted to catchment scale and include special elements such as channels and ponds, which allows to identify potential sources and sinks of water, sediments and associated chemicals within a catchment (Jetten et al., 2003). USLE-type models are lumped models assuming a spatially homogenous uniform hillslope, although it is possible to apply them to more complex terrain using a raster approach to generate spatially distributed outputs (Jetten et al., 2003). Thus, spatially distributed models can be further distinguished between models based on independent raster, where soil loss is estimated at a specific location without contribution from a neighbouring location, and pathway type models, where topological relations are considered (Karydas et al., 2014).

2.3.3 The Universal Soil Loss Equation

The Universal Soil Loss Equation (USLE) (Wischmeier and Smith, 1978) and its modifications are the most widely known modelling approach to assess soil erosion and have been applied in more than 100 countries (Alewell et al., 2019). USLE was developed with an empirical approach based on a statistical analysis of more than 10,000 plot-years of soil loss data (Wischmeier and Smith, 1978). USLE simulates mean annual soil loss due to sheet and rill erosion on hillslopes determined by rainfall erosivity, soil erodibility, topography, crop system and management practice.

In the 1990s, the USLE was further developed to the Revised Universal Soil Loss Equation (RUSLE), by updating the main components of the USLE (Renard et al., 1997). A detailed description of the components of the USLE and its modification is included in Chapter 4.2.5. The RUSLE was further developed to the RUSLE2, which used process-based equations to simulate the detachment, transport and depositions of soil particles (USDA-ARC, 2013). As USLE and RUSLE were designed to predict long-term average annual soil loss and do not consider runoff explicitly, its ability to predict event soil losses is limited (Kinnell, 2010). In a modified version of the USLE (Modified USLE, MUSLE), the erosivity factor based on rainfall energy was replaced with an erosivity factor based on runoff energy,

which improves the equations applicability to individual storm events (Williams, 1975).

Alewell et al. (2019) states that the main reasons USLE type modelling is so widely used throughout the world are i) its high degree of flexibility and data accessibility, ii) a parsimonious parametrisation, iii) extensive scientific literature and iv) comparability of results allowing to adapt the model to nearly every kind of condition and region of the world. According to Risse et al. (1993), 'USLE has been used throughout the world for a variety of purposes and under many different conditions simply because it seems to meet the need better than any other tool available'. In addition, Renard et al. (1997) stated that more than four decades of using the USLE has proven that it is valuable as a conservation-planning guide in the USA, providing farmers and conservation planners with a tool to estimate rates of soil erosion for different cropping systems and land managements.

2.3.4 Uncertainties in water erosion models

The main causes of uncertainty in water erosion models are: i) low input data quality, ii) incomplete description of erosion, sediment transport and deposition processes or model structure, and iii) errors in measured erosion rates for model calibration and evaluation (De Vente et al., 2013).

The significance of uncertain input data for simulating water erosion varies with the model and study location as they influence the sensitivity and importance of factors describing water erosion. In most models and locations, slope steepness and length are important factors for determining water erosion. Thus, the spatial resolution of Digital Elevation Models (DEMs), which are often used to derive slope information for model inputs, has a strong impact on simulated erosion (Mondal et al., 2017; Liu et al., 2011). Also, the temporal resolution of climate data such as rainfall intensity, duration and seasonality, is often named as important input data for the robustness of water erosion estimates (Shen et al., 2012).

Land use information for simulating water erosion can be both difficult to obtain without on-site observations and difficult to simulate. Although some models simulate crop growth and seasonal differences in vegetation cover, several erosion models assume constant land use throughout the simulation period (De Vente et al., 2013; Borrelli et al., 2017). If constant land use is assumed, feedbacks of climate change to soil cover cannot be considered in water erosion models. In addition, land cover data is often obtained from regional land use maps, which is not always a good indicator of the soil's protection from raindrop impact and surface runoff, which is mainly determined by the percentage soil cover and root density (De Vente et al., 2013).

Furthermore, data on soil conservation measures and other field management techniques that influence the soil's protection are often not readily available. Even if detailed data on farming techniques is available, explaining the effectiveness of soil conservation measures in reducing soil runoff remains difficult (Chen et al., 2020).

The requirement for detailed input data is usually highest for process-based models due to their complexity, which may increase the negative impact of uncertain input data on the robustness of model results. Moreover, it can be challenging to obtain detailed input data in data scarce study regions or over large scales (Panagos et al., 2016). For example, polygon-based models such as the WEPP and

EUROSEM models require detailed data to include special elements such as channels and ponds to identify potential sources and sinks of water sediments and associated chemicals within a catchment (Jetten et al., 2003). In addition, detailed field data is needed to evaluate spatial soil erosion and soil deposition patterns. The evaluation of spatial patterns in models is important for the evaluation of their robustness. For example, previous studies predicted acceptable soil loss rates but an incorrect pattern of the source and sink areas and thus they 'predicted the correct result for the wrong reasons' (Jetten et al., 2003). However, even when spatial datasets are available, simulated spatial heterogeneity is difficult to evaluate as the available field data does not always provide the necessary detail and thus spatial relationships in model structure may increase modelling uncertainty (Brazier et al., 2007).

If all the necessary input data is available, process-based models may simulate a more robust estimate of water erosion than less complex model. However, some water erosion processes are hardly understood and thus their representation in models may reduce the robustness of model outputs (Parsons, 2019). Therefore, complex models are exposed to more potential error sources due to their model structure than simpler models (Brazier et al., 2000).

According to Alewell et al. (2019), the two main problems with complex models are i) the unavailability of the complex set of input parameters especially at large temporal and spatial scales and ii) over-parametrisation which results in a non-uniqueness of fit and/or insensitivity of model output to some of the input parameters. Moreover, several studies concluded that more complex models do not perform better than the USLE (Alewell et al., 2019; Jetten et al., 2003; Kinnell, 2010). Nevertheless, as the robustness of complex models is largely determined by the availability of input data, their applicability may improve in the future due to increasing availability of high-resolution data through improving remote sensing techniques (de Vente and Poesen, 2005).

Although USLE is a simple model it also suffers from errors due to the lack of input data or errors due to unknown water erosion processes. For example, the robustness of the topographic factor of the USLE has been questioned as the relationship between slope length and soil erosion is not clear (Bagarello and Ferro, 2010). Also the simplicity of the USLE is repeatedly criticised. Although there is a lack of measured rainfall intensity and sub-hourly data over long-periods, the simple empirical equation implemented in USLE, which correlates the rainfall erosivity factor with available daily/monthly/annualy rainfall, is often criticised (Oliveira et al., 2013). In addition, USLE-type models are often criticised that they do not include gully erosion, bank erosion and mass movements, despite a potentially large contribution to total sediment yield by these processes (De Vente et al., 2013). Evans and Boardman (2016b) criticised that USLE is mainly driven by rainfall and slope factors whereas in Britain the occurrence of erosion is largely determined by soil type and land use. Parsons (2019) mentioned that USLE does not account for interactions between each factor, although previous observations show that soil erodibility is not independent of rainfall intensity. He further points out that USLE is developed from plot data characterised by slopes between 2 - 18 % and fields between 30 to 300 ft (9 - 90 m) long and thus its application on fields with different characteristics is uncertain. As USLE is based on data from small plots, several authors criticised the use of USLE on large scales (Evans, 2013; Evans and Boardman,

2016b). However, this criticism disregards the suggested application of USLE for large-scale applications by its developers, who proposed to 'break down the drainage area into a series of tracts having relatively homogeneous land use and treatment. The erosion equation is then used to approximate the average annual rate of soil movement from each tract' (Wischmeier and Smith, 1965). However, using a raster scheme assumes that cells are independent from each other and thus neglects possible sediment redistribution.

Apart from the RUSLE2 modification, sediment deposition is not considered in USLE-type equations (Kinnell, 2010). Models that do explicitly simulate sediment transport and deposition are based on complex interactions between erosion, deposition and topography and thus require very detailed input data (De Vente et al., 2013). Moreover, modelling sediment redistribution is further complicated as the process is not fully understood (Trimble and Crosson, 2000). In some cases environmental change is primarily reflected in reorganisation within the basin rather than in a direct change in sediment yield at the basin outlet (de Vente and Poesen, 2005). Several studies showed that changes in land use or climate may alter erosion rates in upslope areas but did not significantly change sediment export at the basin outlet (Trimble, 1999; Walling, 1999; Prosser et al., 2001). As USLE-type equations do not consider sediment deposition, they provide gross erosion rates and thus may overestimate the impact of environmental changes on erosion. Furthermore, this hampers the evaluation of model outputs with field measurements, which mostly provide net erosion rates.

As measured water erosion rates are usually used for model calibration and evaluation, their robustness can be crucial for the quality of model outputs. Due to the high heterogeneity of the water erosion process and experimental setups to measure water erosion, measurements are very prone to errors and some methods are more uncertain than others (García-Ruiz et al., 2015; Rüttimann et al., 1995; Parsons, 2019). Furthermore, the observed data needs to be appropriate for the setup of the modelling study. Water erosion measurements at specific temporal and spatial scales might not be representative for other scales. Scaling soil erosion rates up or down may lead to errors as soil erosion processes vary with scale (Prasuhn et al., 2013). For example, Parsons et al. (2004) argues that the quantity of soil eroded is not directly proportional to the area over which erosion is measured. The consideration of temporal scales in measured erosion rates is important as water erosion can be very variable between days, seasons and years and thus event-based or seasonal measurements are not representative of mean annual water erosion (De Vente et al., 2013). Therefore, it is crucial to select data for model calibration and validation that correspond to the spatial and temporal scale at which the model is applied (de Vente and Poesen, 2005). Moreover, a direct comparison between simulated erosion rates on fields or larger scales against erosion assessments based on field studies is difficult as erosion measurements have been often reported to be skewed to higher risk sites and thus field studies often do not include all erosion risk sites equally (Benaud et al., 2020; Cerdan et al., 2010).

In general, the large spatial and temporal variability of soil erosion and the uncertainty of input parameters used in models to predict soil erosion hampers the calibration and validation of soil erosion models, which will not be solved by constructing even more complete, and therefore more complex, models (Jetten et al., 2003). The simplicity of USLE has the advantage of increasing transparency and objectiveness of model evaluation and thus facilitates the interpretation of model results by policy makers (Alewell et al., 2019; Panagos et al., 2016). Moreover, modelling studies usually do not focus on accurately predicting absolute erosion values, but rather to predict relative differences, trends over times and systems reactions to processes and management practices (Alewell et al., 2019). Despite the existing uncertainties, models are a useful tool to test hypotheses on process understanding, relative spatial and temporal variations, scenario development and controlling factors (Oreskes et al., 1994).

2.3.5 Common methods for measuring water erosion based on field experiments

As modelling water erosion can be challenging and includes numerous sources of uncertainty, several authors prefer field-based water erosion assessment methods over modelling (Evans and Boardman, 2016b,a). However, different methods to measure erosion in the field have advantages and disadvantages, including many different uncertainties. In other words, all water erosion assessment methods are subject to criticism (Parsons, 2019; Trimble and Crosson, 2000; Parsons and Foster, 2011). Similar to the varying water erosion estimates generated by different types of models, also the approach chosen to measure water erosion influences the measurement results strongly. For example, Brazier (2004) observed that different methods to measure erosion yielded different erosion rates at a location. Due to the complexity of measuring water erosion there is often only a weak correlation between environmental factors and erosion rates, which hampers the use of observed data in parametrisation and validation of models (García-Ruiz et al., 2015).

The most common methods to measure erosion on cropland are plot experiments, radioisotopic tracer experiments and volumetric surveys. Other methods such as topographic surveys (e.g. laser scanning) are used less frequently to assess erosion on arable land as their resolution capability raises questions about their accuracy to detect finer traces of erosion and erosion on vegetation-covered land (Hsieh et al., 2009). In addition, this method does not distinguish between ground surface lowering due to erosion and other potential reasons on cropland such as soil compaction (Parsons, 2019). Other common methods such as suspended sediment yield measurements and bathymetry do not represent spatially explicit observations of on-site erosion and may include sediments from various sources such as bank erosion or road erosion. Thus, they are of limited value for water erosion assessments focusing only on agricultural land.

Most erosion measurements come from plot experiments introduced in the USA in the 1920s (Hudson, 1993). An overview of this method is provided by Cerdan et al. (2010), Hudson (1993), Mutchler et al. (1994), De Ploey and Gabriels (1980) and Zachar (1982). Soil loss due to water erosion can be quantified from sediment runoff collected from plots varying in sizes and in experimental design (total collection of sediment, fractioned collection of sediments using multislott divisors, measurement of discharge and sediment concentration by tipping buckets and Coshocton wheels). Plot scale studies are useful to build an empirical understanding of soil erosion under different environmental conditions and land use. As such, measurements of soil runoff from plots are useful to compare the relative changes in erosion under different topographic, climate and controlled field management conditions (Auerswald et al., 2009; Cerdan et al., 2010). Plot experiments were used for the development of the USLE model (Brazier, 2004).

However, the accuracy of erosion measurements derived from runoff plots has been questioned due to the complexity in designing plot experiments and collecting soil runoff data. In addition, the spatial extent of plot scale studies has a strong impact on measured erosion rates and thus may create bias (Parsons et al., 2006). In small plots, sediment mobilisation is dominated by rain splash, rill, and interrill erosion and thus sediment yield is relatively low. On larger areas sediment yields are expected to increase due to an increase in active erosion processes and flow shear stress due to concentration of flow (Vandekerckhove et al., 2000). However, with increasing basin areas also sediment storage, e.g. at footslopes, is expected to increase (de Vente et al., 2007). Cerdan et al. (2010) point out that plot erosion rates cannot be directly extrapolated to larger areas as plots do not entirely capture the spatial variability of erosion processes and rates as they occur in the landscape scale. Although the experimental settings of plots are often very resource intensive, erosion plots can only imitate natural conditions (Boix-Fayos et al., 2006).

Radioisotopic tracer experiments use the caesium radionuclide (¹³⁷Cs) released from nuclear weapon tests and nuclear power plant accidents as a tracer to measure long-term mean soil erosion rates in the period since the time of ¹³⁷Cs fallout (mainly in 1960s) until the time of soil sampling for ¹³⁷Cs determination (Fulajtar et al., 2017; Mabit et al., 2014b; Zapata, 2002). ¹³⁷Cs radionuclides from nuclear fallout are deposited in the topsoil and bind to soil colloids, so that they are only moved with the soil particles by mechanical processes such as soil erosion. Its chemical mobility and uptake by plants is negligible (Mabit et al., 2014a; Zapata, 2002). If part of the topsoil contaminated by ¹³⁷Cs is removed by erosion, the ¹³⁷Cs concentrations in soil profiles can be used to trace soil movements (Walling et al., 2014).

The use of radionuclides to assess soil erosion patterns is based on the four main assumptions that i) the fallout is locally, spatially uniform, ii) the fallout is rapidly and irreversibly fixed onto soil particles; iii) the subsequent redistribution of fallout is due to the movement of soil particles; and iv) estimates of soil erosion can be derived from measurements of ¹³⁷Cs inventories (Walling and Quine, 1992). However, the validity of all four of these assumptions has been questioned (Parsons and Foster, 2011). The objections of Parsons and Foster (2011) were discussed by Mabit et al. (2013). One of the most important confirmation of the usefulness of the ¹³⁷Cs method given by Mabit et al. (2013) are the positive results of a comparison between erosion values obtained with various measurement methods and erosion values obtained by ¹³⁷Cs method. Mabit et al. (2013) conclude that the ¹³⁷Cs method is based on a set of presumptions, which should be met to produce useful results, and careful interpretation of the obtained values is needed. If the basic assumptions are met and the methodological recommendations are followed the ¹³⁷Cs method is useful to assess erosion on agricultural fields.

A major advantage of the ¹³⁷Cs method is that it provides long term mean erosion rates (representing the period since ¹³⁷Cs fallout in the 1960s until the time of sampling) and overcomes the problem of high

temporal variability of erosion. Erosion rates from short-term field experiments are stronger influenced by the high interannual variability of erosion, which can lead to unreliable estimates of erosion rates (García-Ruiz et al., 2015). Furthermore, the erosion rates are determined for a grid of ¹³⁷Cs sampling points, which can provide valuable information on the spatial distribution of erosion.

Volumetric measurements of rill erosion are used since approximately the 1940s in the USA (Kaiser, 1978) and the 1950s in Europe (Lobotka, 1955), usually at field scale (Boardman, 1990, 2003; Boardman and Evans, 2020; Brazier, 2004; Evans, 2002, 2013; Herweg, 1988; Zachar, 1982). The volume of erosion rills is calculated by multiplying their lengths and profile cross-sectional areas, which are usually measured along several traverses or estimated from terrestrial and aerial photos (Evans, 1986; Watson and Evans, 1991). The erosion volume is expressed in volume units (m³ ha⁻¹) but can be converted to weight units (t ha⁻¹) by multiplying the volume with the soil bulk density. Volumetric measurements of rill erosion have been considered as a good method to monitor erosion in large fields (Boardman and Evans, 2020).

The most common criticism against volumetric measurements is that it neglects splash and sheetwash erosion as it assumes that rills and gullies are mainly causing soil redistribution within fields (Parsons, 2019). Neglecting sheet erosion in volumetric surveys likely leads to an underestimation of erosion in a field (Parsons, 2019). Erosion rates obtained with ¹³⁷Cs have been observed to be significantly higher than erosion rates derived from volumetric surveys, which highlight the significance of quantifying lessvisible erosion processes, such as sheet wash (Benaud et al., 2020). Volumetric surveys have been also criticised due to their high degree of subjectivity (Panagos et al., 2016). For example, Casali et al. (2006) reported that the error in measurements of the rill length can be ca. +/- 50 % due to their random irregularity. Boardman and Evans (2020) reported the difference in rill volumes ranging from -18% to +50% resulting from different numbers of traverses used for rill cross section measurements. In addition, a large difference between rill volumes derived from field measurements and rill volumes derived from photos has been reported (Watson and Evans, 1991). Therefore, volumetric measurements require careful planning and skilled surveyors to reduce the subjectivity and ensure consistency in obtained erosion estimates. If the robustness of obtained erosion volumes can be ensured, volumetric surveys can be very useful to monitor erosion over large areas on arable land.

2.4 Studies on the impact of water erosion on crop yields

While there are many studies that look at water erosion rates, there are few studies on the effects of water erosion on crops. A major reason is that assessing the erosion–productivity relationship is difficult due to the slow reduction in soil productivity, which may not be recognised until crop production is no longer economically viable (Bakker et al., 2004). Moreover, improved technology often masks erosion impacts (Littleboy et al., 1996), which hampers the observation of a decline in soil productivity over multiple years.

2.4.1 Methods to quantify the impact of water erosion on crop yields

The impact of soil loss on crop yields can be measured experimentally in the field. Water erosion impacts on crops are usually reported as absolute or relative reduction of crop yields per soil lost. Common methods are to artificially remove or add topsoil, comparison of yields in different landscape positions with different soil depths and comparison of yields in different plots with different degrees of water erosion (Bakker et al., 2004).

However, by reviewing studies about the relationships between crop productivity and erosion, Bakker et al. (2004) found a significant variability of the crop productivity response to erosion due to the choice of experimental methodology. Although many factors affect crop yields significantly, the most important factor explaining the observed variations between the studies reviewed is the experimental method used to assess the erosion–productivity relationship. Bakker et al. (2004) identified several sources of methodological uncertainties in existing studies, which may explain the variability in study results: Artificial changes in soil depth have a stronger impact on soil properties in comparison to a slow gradual runoff of soil, which is more common under natural conditions. Moreover, it is challenging to ensure homogenous conditions in different plots and transects in a comparison of crop yields influenced by different degrees of water erosion and thus the impact of other factors on crop growth and development cannot be excluded easily. In addition, soil depth change in natural environments can be also caused by other processes and thus it cannot be that easily related to water erosion alone. Bakker et al. (2004) suggested that the conclusions of past research investigating the relationship between erosion and crop productivity appear to be inconsistent.

2.4.2 Large-scale assessments of the impact of water erosion on crop yields

Panagos et al. (2018) summarised numerous studies analysing the impact of water erosion on crop productivity losses (Bakker et al., 2004, 2007; Den Biggelaar et al., 2004a; van den Born et al., 2000; De la Rosa et al., 2000; Lal, 1995; Larney et al., 2009; Oyedele and Aina, 1998). Most of these studies are focused on Northern America and Europe. Although the outcomes of existing studies are very variable, Panagos et al. (2018) estimated a crop productivity loss of 8% in intensively cultivated agricultural fields with annual erosion rates higher than 11 t ha⁻¹. Panagos et al. (2018) assumes that impacts of water erosion below a rate of 11 t ha⁻¹ will be compensated with fertiliser application in Europe based on the tolerant soil erosion rate suggested by the USDA.

The most comprehensive study on the impact of water erosion on crop productivity was conducted by Den Biggelaar et al. (2004a). Based on a review of field studies of water erosion impacts on various crops in 37 countries, from which half the studies were conducted in North America. Den Biggelaar et al. (2004a) found only minor differences of absolute crop yield declines due to water erosion between continents. However, in relative terms crop yield declines in Africa, Asia, Australia and Latin America were two to six times higher compared to Europe and North America. The high average yields in the most productive regions where mainly responsible for the low impact of water erosion on crops in relative terms. Den Biggelaar et al. (2004b) extrapolated the collected plot data across global soil types and estimated an average global annual reduction of 0.3% of yields from maize, millet, potatoes, sorghum, soybeans and wheat corresponding to an estimated annual loss of USD 523.1 million. However, due to missing data this estimate does not cover the entire globe. Due to a lack of data and the large variation of erosion impacts by soil order and crops, Den Biggelaar et al. (2004b) concluded that in general, little is known about these losses for many important crops in many developing countries. In summary, studies are too limited to generate a global picture of water erosion impacts on crops and to compare water erosion impacts across different regions and crops.

2.5 Studies on the impact of climate change on water erosion

Research on the impacts of climate change on water erosion has a long history, dating back to the 1940s, due to the close link of both processes (Li and Fang, 2016). Earlier research was focused on observed impacts of changing weather parameters (primarily precipitation) on water erosion (Leopold, 1951; Langbein and Schumm, 1958). In the 1990s several studies emerged, in which hypothetical climate change scenarios are used to simulate the impact of artificially modified climate parameters on water erosion (Favis-Mortlock et al., 1991; Favis-Mortlock and Boardman, 1995; Lee et al., 1996). With increasing development of climate and erosion models, the number of studies on the impact of climate change on water erosion substantially increased since the year 2000 (Li and Fang, 2016). The studies are primarily focused on the United States, Europe and China.

The most recent research on climate change impacts on water erosion is commonly based on a modelling approach linking water erosion models (e.g., USLE-based models, WEPP), hydrological models (e.g., SWAT) or crop models (e.g., EPIC) with General Circulation Models (GCMs) (Li and Fang, 2016). Another common approach is to focus on erosion changes caused by land use changes due to different driving forces including climate change (Maeda et al., 2010; Paroissien et al., 2015).

In a review of recent assessments of climate change impacts on water erosion, Li and Fang (2016) concluded that water erosion may both increase or decrease under climate change, depending on geographic locations, climate scenarios, precipitation patterns, topographic conditions, and land management practices. Although, this review found that changes in future water erosion vary widely around the world, in 136 records collected by Li and Fang (2016) erosion rates increased under future climate scenarios, while only 55 records showed a decreasing trend and 14 records indicated no change.

Pruski and Nearing (2002a) compared changes of water erosion, simulated with the WEPP model, with changes in relevant climate parameters at eight locations across the United States. They observed significant direct impacts of precipitation changes on water erosion, but often the key factor affecting water erosion were changes in biomass production induced by soil moisture, atmospheric CO_2 , temperature, and solar radiation. Overall, they concluded that water erosion can be expected to increase where precipitation increases are significant, but if precipitation decreases, water erosion may both increase or decrease depending on the complex interactions between plant biomass, runoff, and erosion. In a

separate study, the same authors simulated the responses of water erosion to precipitation changes using the WEPP model at three locations across the United States, which suggested that each +1% change in precipitation can effect a +2% change in runoff and a +1.7% change in erosion, if other factors stay equal (e.g., temperature, CO_2 levels, and solar radiation) (Pruski and Nearing, 2002b). In another study in the U.S. Midwest, Lee et al. (1996) estimated a similar relative response of water erosion to increasing precipitation. Their outputs from the EPIC model suggested a relative increase in erosion rates twice that of precipitation (e.g. 37 % increase in water erosion for 20% in precipitation) for current temperature and CO_2 . Using EPIC as well, Favis-Mortlock and Boardman (1995) simulated the impact of precipitation changes on water erosion in the UK and found an average increase of annual water erosion by about 12% due to an increase of precipitation of 10% and 2°C warming. In a separate study at the same locations, mean annual water erosion simulated with the EPIC model increased up to 30% under a similar climate change scenario (Favis-Mortlock et al., 1991). The differences to the prior study are likely due to changes of hydrological parameters and a different version of EPIC (Favis-Mortlock and Boardman, 1995), which indicates the difficulty of comparing case studies examining water erosion changes due to climate change.

Several studies have addressed the nonlinearity of changes in surface runoff and water erosion in response to changing climate parameters when accounting for the interactions of precipitation, temperature, and CO₂ concentrations with biomass (e.g., Gautam et al., 2015; Pruski and Nearing, 2002a). This complexity was comprehensively addressed by Nunes et al. (2013), who used a sequential modelling approach linking a storm-based erosion model and a regional climate change model with SWAT, to analyse the impact of projected lower rainfall amounts and higher storm intensities in Mediterranean watersheds with changing vegetation covers. Their results suggest that in many cases an increase in vegetation cover on cropland through warmer temperatures and CO₂ fertilisation is expected to mitigate the impacts of increased storm rainfall. The significant impact of biomass on water erosion changes was also considered by O'Neal et al. (2005), who accounted for the role of different field management practices on climate change impacts on water erosion in the United States. In 10 out of 11 locations water erosion simulated with the WEPP model under climate change scenarios increased by average 95% due to average precipitation change of +7%. In comparison to similar studies soil loss changes were greater and very variable due to the variation in management and planting dates in addition to changing climate parameters. Increased water erosion was caused sometimes by increased precipitation and sometimes by decreased crop cover from lowered maize yields due to extreme heat or drought. In other studies based on the WEPP model, water erosion rates increased even when precipitation decreased due to a drop in maize yields and soil residue cover caused by climate change (Southworth et al., 2000, 2002b,a). Hence, reduced rainfall does not necessarily lead to reduced water erosion. On the contrary, adaptation of soil conservation farming techniques can lead to reduced water erosion even when precipitation increases. For example, Klik and Eitzinger (2010) used the WEPP model to demonstrate the effectiveness of notillage and grass cover to decrease future water erosion under climate change projections in Austria.

Mullan et al. (2012) point out that the biggest limitations of studies on climate change impacts on

water erosion are the lack of considering land use and field management, next to the coarse temporal and spatial scale of climate change data. To address the significance of these limitations, the authors created different land use, management and rainfall intensity scenarios and simulated water erosion in Northern Ireland using the WEPP model and sub-daily and spatially downscaled rainfall projections. Model outputs suggested both a dramatic increase and decrease in future water erosion rates, depending on the specific scenario. Neglecting sub-daily rainfall and land use management scenarios would lead to a projected decrease in water erosion rates and thus hide the potential for increased water erosion under climate change in the study area. Due to the importance of sub-daily rainfall in their analysis, Mullan et al. (2012) stressed the need to look beyond average annual precipitation amounts to analyse future rates of soil erosion. Other studies underline this conclusions. For example, Routschek et al. (2014) suggest that projected changes in precipitation patterns including increasing rainfall intensities will lead to a significant increase in water erosion by 2050 in Saxony, Germany. Additionally, Zhang and Nearing (2005) attributed increased soil loss simulated with the WEPP model under selected climate scenarios in central Oklahoma in the United States to increased frequency of large storms represented by an increased variability in monthly precipitation. Further, Pruski and Nearing (2002b) addressed the different impacts of annual precipitation amounts and daily precipitation amounts (representing precipitation intensity) and suggested that changes in the latter had stronger impacts on water erosion changes. Although changes in precipitation intensity have been identified as highly relevant for climate change impact studies on water erosion, few studies include sub-daily rainfall data due to the coarse temporal resolution of projected climate parameters (Li and Fang, 2016).

In summary, model outputs about interactions between climate change and water erosion are highly variable between studies. The strongest agreement appears to be on a predicted increase in water erosion rates when precipitation increases significantly. However, the rate of increase varies between studies. Due to the many interactions between biomass and water erosion under changing climate parameters, general conclusions on the impact of climate change on water erosion are challenging to draw from present modelling studies. In addition, some interactions are neglected due to missing input data or lack of process representation in the model structure. Common addressed shortcomings are the lack of accounting for gully erosion in models, sub-daily rainfall or for changing rain to snow ratios with changing temperatures (Li and Fang, 2016). The latter can have a strong impact on the distribution of seasonal precipitation patterns and thus water erosion rates (Mukundan et al., 2013). In addition, most studies are based on field management scenarios, which do not account for farmers responses to a changing climate. As changes in crop cover are often decisive in climate change impacts on water erosion, this leads to a major uncertainty in predicted water erosion rates. In addition, shortcomings of models to simulate climate change impacts on crops, including crop responses to extreme heat or rainfall events and CO_2 fertilisation, add uncertainty to modelled water erosion rates. General conclusions about climate change impacts on water erosion are also hampered by the difficulty of comparing study results as often different models or model parameters are used. Furthermore, conclusions derived from modelled water erosion changes under climate change are linked to model inputs from a specific location describing environmental characteristics and land use. Transferring findings from one study locations to different locations with different circumstances is difficult as water erosion does not response linear to changing climate parameters. Therefore, extrapolating model outputs from field studies to larger scales is uncertain as well. As most studies were conducted in Europe, China and North America, little is known about climate change impacts on water erosion outside these regions.

2.6 Discussion

2.6.1 Models are the preferred method for assessing water erosion at large scales

Models are necessary to provide much needed large-scale indicators of water erosion using increasingly available datasets on topography, soils, climate and land use. Although empirical water erosion models such as USLE are often criticised as oversimplified models (e.g., Trimble and Crosson, 2000), its simplicity facilitates the application on large scales and data-scarce regions and improves the transparency of model outputs. Several authors agree that the simplicity of USLE-type models makes them most suitable for global water erosion assessments (Borrelli et al., 2017; Doetterl et al., 2012; Van Oost et al., 2007; Alewell et al., 2019).

More complex water erosion models demand detailed input data and calibration of model parameters, which becomes increasingly challenging with increasing scale of a study. This complexity is a major disadvantage for regional or global studies. Furthermore, even complex models with detailed data cannot fully simulate the real world as some water erosion processes are not fully understood.

Some authors criticise the use of water erosion models on large scales and prefer to use field experiments instead (Evans and Boardman, 2016b). Although field studies have many shortcomings as well, they may be more reliable to monitor erosion at specific locations if enough resources are available. However, major drawbacks of field experiments are their subjectivity as field methods are not fully standardised, as well as their high resource intensity as they demand a large amount of time and knowledge. Thus, they are not suitable to acquire the large amount of data necessary for large-scale and global assessments (Panagos et al., 2016). Moreover, water erosion models are needed to run scenarios (e.g. climate change impacts) and can be linked with other models (e.g. crop models) to quantify impacts of water erosion; thus, they are essential to address many pressing research objectives.

2.6.2 A better evaluation of the modelled water erosion estimates is needed

As USLE-type models are empirical models based on observations from the Midwest United States global water erosion estimates need to be evaluated. The robustness of USLE-type models decreases with increasing deviation of environmental characteristics that shaped the data used to develop the model. Previous studies evaluated their global estimates by comparing it to similar modelling studies or to a collection of field data derived from a mix of methods. However, previous studies neglected the varying quality of water erosion estimates for different environments in their evaluations. Therefore, there is a need for a more detailed robustness analysis of global water erosion estimates, which accounts for the

impact of different environmental characteristics on model outputs.

Furthermore, model outputs vary between different models as they differ in complexity, structure and parametrisation. Even the different models from the USLE-family generate different water erosion estimates. Model intercomparison projects, which have been very useful for the crop and climate modelling community (Rosenzweig et al., 2013, 2014; Mueller et al., 2017), may help to further address and quantify the uncertainties of water erosion models.

2.6.3 The impact of water erosion on crops is not adequately quantified

Whilst previous studies have focused on improving the spatial and temporal resolution of input data such as topography or precipitation (Naipal et al., 2015; Panagos et al., 2017), the important impact of farming techniques and different crops on water erosion rates has been neglected. One reason is that previous studies used static input data, which does not account for the daily interactions between climate, soil, crops and their management.

As common modelling methods used in large-scale water erosion assessments do not account for the daily interactions between crops and water erosion or different field management intensities (e.g. fertiliser use), they are of limited value to assess water erosion impacts on crop yields. Whilst some field studies exist, which quantify water erosion impacts on crop yields, large-scale assessments are rare. The few attempts to estimate water erosion impacts on crops on a continental scale are based on a limited amount of field experiments (Den Biggelaar et al., 2004a; Panagos et al., 2018).

Field experiments to assess water erosion impacts on crops are highly complex due to the slow reduction in soil productivity. Furthermore, they are highly subjective as no standardised way exists to measure this process. As the method used to measure water erosion impacts on crops can be the strongest driver influencing study results (Bakker et al., 2004), comparing the conclusions of different field studies or collating data from field studies to generate conclusions on larger scales is highly uncertain. Due to a lack of data, previous assessments on large-scale water erosion impacts on crops relied on the limited and very heterogeneous data currently available. However, the importance of the setup of field experiments can mask the impact of different environmental and management characteristics on water erosion rates. Moreover, the available data only represents a limited amount of environments, crops and field management conditions.

As a result, little is known about the effects of water erosion on crop production and its economic consequences. In addition, sound conclusions on geographically varying water erosion impacts cannot be drawn from existing studies and thus large-scale indicators to highlight regions most susceptible to water erosion impacts are lacking. To assess water erosion impacts on crops on a large or global scale a consistent method is needed to capture the effects of varying site characteristics on water erosion impacts.

2.6.4 The impact of climate change on water erosion is unclear

On smaller scales, process-based crop models are frequently used to analyse the relationship between climate change and water erosion on cropland. Whilst studies could demonstrate a close link between

changing climate parameters and water erosion, projecting changes in water erosion rates and impacts is challenging due to many reasons. Some of the most import reasons are the uncertainty in projected precipitation changes and their low temporal resolution. Moreover, disentangling the impact of single climate parameters on water erosion is difficult due to the many complex interactions between climate, soil and land use. Numerous modelling studies have already addressed these complexities, which led to important insights on the significance of land use and climate change impacts on crops for projecting water erosion changes.

As climate change impacts on crops vary around the globe it can be expected that also climate change impacts on water erosion vary in different global regions. Thus, hotspots should be identified where the susceptibility of crop production to water erosion may potentially increase due to climate change to concentrate efforts and research on soil conservation strategies at locations where they are most needed in the future. However, the conclusions of existing studies are limited to single fields, mostly located in Europe, China and North America. Moreover, as their study design has an overwhelming effect on the model outputs (e.g. models used, climate and field management scenarios), the different studies cannot be directly compared with each other. Thus, using the outcomes of existing studies to project differences of changing water erosion rates in different regions is limited.

Application of the models used to analyse the links between climate change and water erosion on larger scales would generate model outputs that describe the impact of climate change on interactions between crops and water erosion for different regions in a consistent way. This could provide useful indicators on the geographic variations of climate change impacts on water erosion and their variability under different climate and management scenarios. In addition, the model outputs can provide another layer to climate change impact assessments on crops by accounting for water erosion impacts on crops. This will help to identify regions vulnerable to both climate change and water erosion.

2.7 Conclusion

This literature review addressed several knowledge gaps about i) the uncertainty of water erosion estimates, ii) the impacts of water erosion on crop production, and iii) the impact of climate change on water erosion and its consequences for crop production. As there is a need for large-scale indicators on water erosion rates and impacts, and on land degradation more generally, these knowledge gaps need to be addressed.

Process-based crop models coupled with water erosion models are useful tools to simulate links between crops and water erosion and thus can address many of the literature gaps. The simplicity of empirical equations such as USLE-type models facilitates their application on a global scale and their linkage to process-based crop models.

However, global water erosion estimates based on USLE-type models include many uncertainties, which need to be addressed more thoroughly. This is considered in the robustness analysis of EPIC-IIASA's water erosion estimates, which is carried out as part of the first research objective of this thesis.

Improving confidence in water erosion estimates simulated using USLE in conjunction with GGCMs will provide a tool for analysing plant-water erosion interactions on a global scale. This will enable the simulation of new large-scale water erosion indicators targeted under the second and third research objectives, including (i) estimates of the impact of water erosion on crops for different global regions; and (ii) insights into the impacts of climate change on water erosion in different global regions.

Chapter 3

Literature review of crop models

3.1 Introduction

Crop models are used to simulate crop yields as a function of genotype, soil, weather and management strategies. Climate change and water erosion have an impact on the interacting processes in crop production systems and thus influence crop growth, crop development and crop yields. Crop models are frequently used to investigate the impacts of climate change on potential crop productivity via perturbations in the mean and variability of climate variables such as temperature and precipitation, as well as through the impacts of increasing atmospheric CO_2 concentration (Ewert et al., 2015; White et al., 2011). In addition, crop models are used as a tool to explore field management strategies for improving the resilience of crop production systems to climate change (Matthews et al., 2013; Webber et al., 2014). As not all interacting processes between the soil, biomass and climate are understood, crop models differ in their structure and complexity. Uncertainties in crop response to changing temperature, precipitation and atmospheric CO_2 are often discussed in crop modelling studies because of their relevance in climate change impact studies (e.g., Deryng et al., 2014; Osborne et al., 2013). More complex crop models also simulate interactions between the plant-soil-system, which are very important for exploring field management strategies for adapting to climate change.

Given the global importance of climate change and the central interest in agriculture, crop models are increasingly applied at the global scale (using global gridded input data infrastructures) to assess the impact of climate change on global crop production (Mueller et al., 2017). In addition, global gridded crop models (GGCMs) are used to address a wide range of questions beyond the impact of climate change on annual crop yields. For example, they are frequently used to assess weather-induced crop-yield variability, the impact of extreme weather events on crops, or yield gaps due to water and nutrient deficiency (e.g., Balkovič et al., 2014; Frieler et al., 2017; Jägermeyr et al., 2016; Leng and Hall, 2019). Moreover, model outputs of GGCMs are frequently used as inputs for global economic models to explore wider questions of the impacts of changing crop production in the face of population growth and economic development (Nelson et al., 2014). As such, GGCMs are an essential tool for addressing many pressing research questions.

However, GGCMs are largely driven by variable information on weather and atmospheric CO_2 concentrations, whereas assumptions on soil properties and/or management systems are usually static; thus, models do not reproduce all temporal dynamics and spatial patterns of observed crops yields (Mueller et al., 2017). As the contribution of weather variability to crop variability is lower in low-input systems than in high-input systems (Ray et al., 2015), negligence of other processes influencing crop yields such as the impact of pests, diseases, changes in management and soil quality through land degradation, reduces the significance of model outputs in some global regions. Accounting for land degradation depend substantially on soil parameters and soil process parameterisation, i.e. hydrology and nutrient turnover, which are often neglected domains in GGCMs (Folberth et al., 2019).

This literature review provides an overview of the types and uses of crop models and discusses their uncertainties. It will outline the focus of the latest large-scale and global climate change impact studies based on crop models, highlighting frequently neglected topics that need further scrutiny.

3.2 Crop model types

Crop models are mostly deterministic models. They are used to estimate crop yields as a function of input parameters and initial conditions. Deterministic crop models can be further divided into statistical, process-based and empirical models.

Statistical models can be used as a simple form of crop model, which are often based on regression equations. With the help of statistical crop models, the past relationship between changes in crop yields and determining parameters such as climate conditions can be used to receive estimates for future yield trends (Lobell et al., 2011). The advantage of this method is that no extensive input data and field calibration data is required, which can be helpful in regions where data is scarce (Lobell and Burke, 2010). However, predictions from statistical models are limited to the locations the data was derived from to develop the model (Ritchie and Alagarswamy, 2002). Moreover, statistical models do not predict responses to nonlinearities and do not explain the reason for certain changes (Jones et al., 2016; Challinor et al., 2009). Thus, statistical models are restricted to the climates and atmospheric conditions of today, and generally do not take into account interactions between crop genotype and environmental and management factors (Rötter et al., 2011). As such, they are of limited use for generating crop yield projections under different climate change scenarios and for adaptation decisions.

Process-based (or mechanistic) models are the most complex type of model because it is assumed that the individual parts of a system and their connections are understood and thus they can simulate an approximation of the real system. In process-based modelling crop yields are determined by equations representing a crop's physiological response to environmental variables (Estes et al., 2013). Although process-based models can simulate processes very close to the real system, they suffer from the major drawback of requiring a high amount of input data.

Unlike process-based models, empirical models are not based on the assumption that all underlying biological, physical and chemical processes of a system are understood. Empirical models are based

on observed relationships among experimental data, which are represented in a model. The simulated relationships fit the observed relationship and thus allow to predict what will happen in certain circumstances that fit the prior observed circumstances (Hudson, 1993). Due to the simulation of relationships with the help of empirical observations rather than mathematical descriptions, empirical models require less data to run and are thus easier to use on a wider scale. Common crop models often include both process-based and empirical elements.

3.3 Typical crop model routines to simulate crop growth

Common process-based crop models share the same fundamental approach of simulating the development of crops from germination to harvest as a response to climate, soil, phenological characteristics and field management. These determinants also influence the timing of different phenological stages such as flowering, grain-filling and the partitioning of the biomass into grain and other plant parts, which controls the amount of the yield. The many single factors used in the simulation of crop growth and yield can be divided into defining factors such as CO_2 , radiation, temperature, and crop characteristics, which determine the potential crop production, and factors limiting or reducing crop growth such as nutrient and water deficiency and the exposure to pollutants, pests, diseases and weeds (Jones et al., 2016; Bouman et al., 1996; Rabbinge, 1993). The range of factors involved in the simulation of crop growth and yields varies between crop models.

The most common way of simulating the development of crops is by calculating the timing of key phenological stages in the plants life cycle as a function of temperature (degree-days) and daylength (Boote et al., 2013). The potential rate of development is usually modified by limiting and reducing factors, which play an important role in delaying or accelerating specific crop development phases (Boote et al., 2013). Next to capturing the development characteristics of a crop, the final yield also depends on the separately simulated daily growth rate (Dodds, 2010). A common method to predict biomass accumulation is based on the work by Monteith (1977), who found a linear relationship between seasonal aboveground biomass accumulation of C-3 crops grown without stress in Britain, and intercepted solar radiation. The slope of this relation is the radiation use efficiency (RUE) in gMJ^{-1} , which is used by crop models as a constant to link plant growth to solar radiation (Boote et al., 2013; Dodds, 2010). Some crop models alternatively or additionally link plant growth to crop water use by using transpiration use efficiency (TUE) as a constant, which is the ratio of biomass produced per unit of water transpired by a crop. This method is based on the work by De Wit (1958) who correlated water use with plant growth. This relationship has been frequently used and modified, most prominently by adding vapour pressure deficit (VPD) as an environmental variable, which is addressing the increase in transpiration of a plant as the vapour pressure deficit of its surrounding air increases (Tanner and Sinclair, 1983). With the latter modification TUE is measured in $gkg^{-1}kPa$. Best estimates for crop growth rates can be achieved by using a combination of both the RUE and TUE methods (Dodds, 2010). Leaf area expansion plays a crucial part in estimating the interception of solar radiation; thus, this process is an important component in the simulation of plant growth rates. However, it is a difficult process to simulate as it is highly sensitive to temperature, atmospheric CO_2 concentration and the availability of water and nutrients (Boote et al., 2013). Common methods implemented in crop models are to use empirical leaf area curves specifying the course of growth and senescence of the leaf area during the growing season, which are influenced by crop development stages and crop stress. The simulation of leaf expansion can be either independent from the plants growth rate or calculated as a function of biomass accumulation (Dodds, 2010). Once the end of the growing season is reached a typical method to determine final yield is to use a prior defined harvest index, which is the ratio of the grain yield to the total above-ground biomass. Some crop models also simulate individual grains during the growing season in order to determine the final yield.

More complex crop models also simulate soil processes such as carbon, nutrient and water dynamics in the plant-soil-system and thus can provide outputs of soil components next to crop yield estimates (Boote et al., 2013). Outputs of those sub-models are further used to derive limiting and reducing factors to adjust potential biomass to actual biomass. Modelling the outputs and inputs of the water balance between crops and the soil is necessary in order to calculate important processes for biomass accumulation and crop development such as time-varying availability of water and nutrients for the plant (Jones et al., 2016). Therefore, most crop models are able to simulate hydrological processes such as infiltration, percolation, runoff and evapotranspiration, as well as carbon and nitrogen dynamics. In order to represent these dynamics, the physical and chemical processes in the soil need to be simulated as well as field management interventions, which have an impact on those processes. The many factors influencing those dynamics complicate adequate representation of water, carbon and nitrogen flows and thus common crop models differ in the way they include those processes. Furthermore, some processes such as responses to phosphorus, potassium, or micronutrients in the plant-soil-system are even more complex and are thus hardly implemented in crop models. Similar to this, routines to simulate the reduction of actual biomass due to damages on plants by insects, diseases, weeds and pollutants are absent in most crop models due to their complexity (Jones et al., 2016).

3.4 Crop models and their applications

The earliest crop growth simulation models were developed in the 1960s (De Wit, 1965; Monteith, 1965). Today, numerous crop models exist. The selection of crop models for a study depends to a large degree on the emphasis on different research questions, crops and regions (Ewert et al., 2015). The latter can be decisive for choosing crop models as models often perform best in regions, where the data was collected to support the development of the model (Challinor et al., 2017). Some crop models may be more suitable for climate change impact studies, whereas other models might be a better fit for a study on the impact of management strategies on soil productivity. For example, the crop models STICS (Brisson et al., 1998) and HERMES (Kersebaum, 2007) are frequently used to simulate crop N stress. APSIM (Keating et al., 2003) and DSSAT crop models (Dzotsi et al., 2010) have been proposed as useful crop models to

simulate the impact of P deficiency on crops (Ewert et al., 2015). The CROPGO model includes routines to account for soil salinity impacts on crops (Webber et al., 2010). CERES (Hoogenboom et al., 2019), EPIC (Sharpley and Williams, 1990) and APSIM (Keating et al., 2003) were the most frequently used crop models in a review of studies using crop simulation models to investigate different aspects of how climate change might affect agricultural systems. (White et al., 2011). Few crop models include sub-models to simulate water erosion and its impact on crops. The most prominent exceptions are EPIC and APSIM. Other crop models focusing on the sustainable use of soil resources are NRTM (Shaffer et al., 1983), CENTURY (Parton et al., 1987) and PERFECT (Littleboy et al., 1989), but they are limited in their ability to address crop management issues and accurate simulation of crop yields and are used less frequently in crop productivity studies (Keating et al., 2003; White et al., 2011).

Crop models are applied at field, regional and global scales. Most early crop models were developed for application at plot and field scale; thus, early applications of crop models in climate change impact studies were mainly site-based (Ewert et al., 2015). Recently, global gridded crop models (GGCMs) have been developed by combining crop or ecosystem models with global gridded input data infrastructures to directly use the outputs of global climate models to generate projections of future crop yields (Challinor et al., 2009). In addition, large-scale model outputs can easier be used as indicators for administrative units to inform environmental and agricultural policies (Ewert et al., 2011; Alewell et al., 2019). A major difference to field-scale model applications is that GGCMs cannot be calibrated and evaluated against field observations. Usually the evaluation of the performance of large- to global-scale models is based on a comparison to (sub-) national yield statistics, or to gridded yield data sets that are based on (sub-)national statistics (Mueller et al., 2017). Today, several prominent crop models are used next to ecosystems simulation models for climate change impact assessments at regional to global scales. Commonly used crop and ecosystems simulation models for large-scale climate change impact assessments include EPIC (Balkovič et al., 2014; Liu et al., 2007), DayCent (Stehfest et al., 2007), PEGASUS (Deryng et al., 2011), DSSAT (Deryng et al., 2014; Schauberger et al., 2017), GLAM (Osborne et al., 2007), LPJmL (Bondeau et al., 2007) and WOFOST (Boogaard et al., 2013).

Although crop models are now widely used to assess the impacts of climate change, most models were originally developed to support decisions about field management, such as choice of crop cultivar, irrigation timing and fertiliser application (Hertel and Lobell, 2014). Hence, crop models are often used for climate change adaptation studies. According to Matthews et al. (2013), crop models can contribute to i) determining where and how well crops of the future will grow; ii) crop improvement programmes; iii) identifying what future crop management practices will be appropriate and iv) assessing risk to crop production in the face of greater climate variability. The most frequently explored strategies in climate change adaptation studies based on crop models are changes in planting dates and cropping systems (cultivars or rotation), whereas new practices (e.g. type and distribution of fertilisers and tillage practices), interacting practices (e.g. planting dates and irrigation management) and improved crop characteristics are explored less frequently (White et al., 2011; Ewert et al., 2015). To thoroughly account for the impacts of different field management strategies on crop yields, models need to simulate the

interactions between farming techniques, such as tillage, crop and residue management, and processes affecting soil quality.

3.5 Global climate change impact assessments using crop models

Knowledge on the magnitude, rate, and pattern of climate change impacts on agricultural productivity are based on more than two decades of research (Rosenzweig et al., 2014). Common methods to analyse these impacts are based on GGCMs and statistical analysis of historical data. Studies comparing methods found that estimated climate change impacts using different methods yield similar results (Lobell and Asseng, 2017; Liu et al., 2016; Zhao et al., 2017). Crops most considered in climate change impact assessments are wheat, maize, soybean and rice and the dominant regions studied are the USA and Europe (White et al., 2011).

The results of GGCMs are compared in the *Agricultural Model Intercomparison and Improvement Project* (AgMIP) (Rosenzweig et al., 2013) and the *Inter-Sectoral Impact Model Intercomparison Project* (ISI-MIP) (Warszawski et al., 2014). Rosenzweig et al. (2014) summarised model outputs of seven GGCMs participating in AgMIP and indicated strong negative impacts of climate change on crop production, especially at low latitudes and at higher levels of warming. Model outputs under the RCP 8.5 scenario are summarised in Figure 3.1, which illustrates that highest negative yield changes for future maize, wheat, rice and soybean yields are at low latitudes. The contrast of climate change impacts between high-latitudes and low-latitudes is most pronounced for maize and wheat among the assessed crops with general agreement among models. Strong negative climate change impacts on crops at low latitudes can occur at moderate temperature increases (1 to 2 °C) due to the narrow gap between the high- temperature thresholds for suitable production of crops such as maize and wheat and the present temperature in tropical climates (Hatfield et al., 2011). This conclusion is confirmed by several other studies observing highest climate change impacts on major staple crops at low latitudes (e.g., Balkovič et al., 2014; Zhao et al., 2017). In addition, one study confirms these patterns by suggesting that areas in Russia, China, and Canada are projected to gain suitable plant growing days (Mora et al., 2015).

Most studies analysing projected climate change impacts on crop yields focus on the impact of temperature changes on crop yields. In addition, most studies account for future changes in field management by analysing climate change impacts under different fertilisation and irrigation management scenarios. Rosenzweig et al. (2014) summarised the simulated yield changes presented in Figure 3.1 based on model outputs with and without explicit nitrogen stress. The former considers current fertiliser application rates based on observational databases, whereas the latter scenario considers ideal fertiliser application that completely eliminates nitrogen stress for crops, which considerably reduces negative climate change impacts. Similarly, Balkovič et al. (2014) explored the impact of climate change on global wheat production under different irrigation and fertilisation intensification scenarios and concluded that the negative impacts of climate change can partly be offset by adequate intensification using nutrient additions and currently existing irrigation infrastructure. Some studies simulate climate change impacts



Figure 3.1: AgMIP median yield changes (%) for RCP8.5 (2070–2099 in comparison to 1980–2010 baseline) with CO_2 effects over all five GCMs x seven GGCMs (6 GGCMs for rice) for rainfed maize (35 ensemble members), wheat (35 ensemble members), rice (30 ensemble members), and soy (35 ensemble members). Hatching indicates areas where more than 70% of the ensemble members agree on the directionality of the impact factor. The bottom 8 panels show the corresponding yield change patterns over all five GCMs x four GGCMs with nitrogen stress (20 ensemble members from EPIC, GEPIC, pDSSAT, and PEGASUS; except for rice which has 15) (Left); and 3 GGCMs without nitrogen stress (15 ensemble members from GAEZ-IMAGE, LPJ-GUESS, and LPJmL) (Rosenzweig et al. 2014).

with more sophisticated field management scenarios. For example, Iizumi et al. (2017) addressed responds of crop yields to global temperature considering agronomic adjustments through socioeconomic changes by accounting for improved technologies, field management and historical patterns of fertiliser input in a GGCM. The model outputs show that global mean yields of maize and soybean will decrease with warming at the end of this century even when agronomic adjustments are considered, whereas those of rice and wheat increase with warming. However, complex future field management scenarios are usually highly uncertain. Therefore, some studies choose to present their results based on scenarios reflecting current conditions. For example, Zhao et al. (2017) used GGCMs, local point-based models, statistical regressions, and field-warming experiments to demonstrate that without CO_2 fertilisation, effective adaptation, and genetic improvement, each degree-Celsius increase in global mean temperature would, on average, reduce global yields of wheat by 6.0%, rice by 3.2%, maize by 7.4%, and soybean by 3.1%.

As indicated in the last example, atmospheric CO_2 fertilisation is regularly addressed in climate change impact studies and was found to be significant for crop yield projections (e.g., Levis et al., 2018; Ren et al., 2018; Tebaldi and Lobell, 2018). In a review about crop modelling studies that were conducted from 1995 to 2002, Tubiello and Ewert (2002) found that about half of the studies dealing with climate change considered responses to CO_2 . A more recent review mentioned that most papers ignoring effects of CO_2 were published prior to the year 2000 suggesting an increasing consideration of CO_2 fertilisation in climate change impact studies (White et al., 2011). For example, Levis et al. (2018) analysed the effects of CO_2 fertilisation on C3 and C4 crops using a mechanistic model embedded in an earth system model, and found that CO₂ fertilisation can mitigate yield losses of C3 crops such as wheat, rice and soybean worldwide. This is in line with studies reviewed by the IPCC (IPCC, 2014) and confirmed in recent studies (e.g., Ren et al., 2018; Reyer et al., 2017; Tebaldi and Lobell, 2018). However, the mitigation effect of CO₂ fertilisation varies for crops and regions. For example, in a regional analysis in the Middle East and northern Africa, Waha et al. (2017) found a significant correlation between crop yield decrease (legumes and maize are worst affected) and temperature increase regardless of whether the effects of CO₂ fertilisation or adaptation measures are taken into account. Similarly, Iizumi et al. (2018) compared outputs from a GGCM running with a factual and a non-warming counterfactual climate scenario as input respectively and concluded that between 1981 - 2010 climate change has decreased global mean yields of maize, wheat and soybeans by 4.1, 1.8 and 4.5%, even when CO_2 fertilisation and agronomic adjustments are considered. Rosenzweig et al. (2014) mentioned that CO_2 fertilisation does not compensate for increases in water demand and shortening of already-short growing periods for annual C3 crops in the tropics. However, Rosenzweig et al. (2014) also mentioned that responses to climate change impacts between models are more similar and much more negative across tropical and midlatitude bands once CO_2 effects are removed. However, relative yield changes with and without CO_2 effects are much closer in C4 maize than in the C3 crops indicating the higher significance of CO_2 effects for certain crops. Conclusions from field experiments suggest that the direct fertilisation effect on crop yields in model projections is overestimated suggesting that crop model parametrisation of CO_2 effects remains a crucial area of research (e.g., Ainsworth et al., 2008; Long et al., 2006b). One study also highlighted that increasing CO_2 lowers concentrations of important nutrients in crops; thus, threatening human nutrition (Myers et al., 2014).

Whilst changes in precipitation intensity, amounts and patterns are often decisive for observed seasonal crop yield changes (Huntington et al., 2017), they are less suitable to explain projected yield trends. Often, studies analysing projected climate impacts focus on temperature, which is more important than precipitation for yield trends, whereas precipitation plays a more important role in year-to-year variability of crop production than temperature (Lobell and Burke, 2008; Lobell and Field, 2007). Moreover, climate models perform the simulation of large-scale patterns of precipitation less well than for surface temperature (IPCC, 2014). Nevertheless, some studies focus on projected rainfall-related impacts for agriculture due to crop damage through heavy precipitation and flooding, or due to droughts through lack of rainfall. The latter is very important on rainfed cropland and is among the most severe climate related risk for crops (FAO, 2018); thus, it is subject to most studies researching precipitation related climate change impacts. For example, projected changes in precipitation where used by Al-Bakri et al. (2010) to asses their impact on rainfed wheat and barley in the Yarmouk basin in Jordan using the crop simulation model DSSAT. Their model outputs suggested that a reduction of rainfall by 10–20% reduced
the expected yield by 4–8% for barley and 10–20% for wheat, respectively. An increase in rainfall by 10-20% increased the expected yield by 3-5% for barley and 9-18% for wheat, respectively. The impact of projected precipitation changes on crop yields were analysed on a larger scale by Leng and Hall (2019) using a drought index and the outputs of 11 GGCMs participating in the AGMIP and ISI-MIP projects. They calculated an increase in yield loss risk by 9%–12%, 5.6%–6.3%, 18.1%–19.4% and 15.1%–16.1% for wheat, maize, rice and soybeans, respectively, in the ten largest producing countries by the end of the 21st century without considering the benefits of CO_2 fertilisation and adaptations. Whilst studies often highlight the positive effects of increasing precipitation on crop yields (e.g., Kang et al., 2009), few studies focus on the negative impacts of projected precipitation increases on crop yields. Often crop models fail to capture the negative impacts of heavy rain and extremely wet conditions (Balkovič et al., 2018). In one notable study based on the application of a modified version of the model CERES-Maize, Rosenzweig et al. (2002) estimated a doubling of crop damage within 30 years in the USA from increased heavy precipitation under climate change causing excess soil moisture and flooding. Rosenzweig et al. (2002) focused their discussion on crop losses through anoxic conditions, increased risk of plant disease and insect infestation, or delayed planting or harvesting due to inability to operate machinery caused by excess soil moisture and flooding. Water erosion impacts on crops are usually neglected in studies analysing yield changes due to precipitation changes. The relationships between crops, precipitation and water erosion in response to climate change are highly complex and are thus rather subject to field scale studies, which focus on disentangling these interactions to derive conclusions on potential changes in future water erosion rates (e.g., Mullan et al., 2012; Nunes et al., 2013; O'Neal et al., 2005).

In summary, most studies on projected climate change impacts focus on the impact of changes in temperature, precipitation and atmospheric CO_2 on crop yields. Although not all impacts of those changing climate parameters are considered or their effects are uncertain. Other important climate parameters are neglected in present studies such as solar radiation, wind speed, relative humidity, and atmospheric ozone (White et al., 2011). One important reason for their negligence is the low confidence in the projected changes of these climate parameters (IPCC, 2014).

3.6 Uncertainty and limitations of crop models

Due to uncertainty in certain model input data, model parameters and representation of biophysical processes, the robustness of model outputs varies for different locations, scales, environments, crops and scenarios. As all uncertainties cannot be addressed in a complex model study, it is common to identify and quantify the most important model uncertainties for a specific study (Gabbert et al., 2010). Uncertainty ranges are commonly determined by varying model outputs simulated with varying input values for the most sensitive parameters or through outputs simulated with different models. However, the latter can be challenging due to a wide variety of methodological approaches in different models (White et al., 2011; Asseng et al., 2013).

As crop model studies are based on numerous methods, model intercomparison projects (MIPs)

have been established to compare model outputs from different sites, crops and spatial scales and to develop standard protocols for modelling impacts. Prominent MIPs are AgMIP (Rosenzweig et al., 2013), ISI-MIP (Warszawski et al., 2014) and *Modelling European Agriculture with Climate Change for Food Security* (MACSUR) (Bindi et al., 2015). Additionally, these comparison projects play an important role in identifying and assessing the most significant uncertainties in crop model outputs (Challinor et al., 2017; Mueller et al., 2017).

Commonly discussed uncertainties due to limitations of crop models result from the need to strengthen the physiological assumptions of models (e.g. plant response to temperature), responses to CO_2 and genetic diversity of crop cultivars (White et al., 2011; Boote et al., 2013). Lacking representation of extreme weather events in crop models, such as increased frequency and intensity of extreme heat, drought or heavy rains, could lead to an overestimation of positive climate change impacts (Balkovič et al., 2018). In addition, several non-weather related processes influencing crop yields are not considered in most crop models such as impacts of pests, diseases and weeds (White et al., 2011). Further, very few crop models include effects of frost, snow, hail, flood, wind, and ozone on crop growth (Ewert et al., 2015). Also, soil degradation and soil microbial processes influencing nutrient cycles and nutrient availability for plants are often not taken fully into account in most crop models (Folberth et al., 2014).

These limitation are often the result of a lack of field data to evaluate models. Model outputs are usually compared against data from field experiments reflecting a wide range of crops and growing conditions defined by varying CO_2 and ozone concentrations, drought effects and deficit irrigation (Ewert et al., 2015; Boote et al., 2013). Global coverage of data from long-term experiments and the representation of all growing conditions in field experiments need to be improved to comprehensively evaluate crop model outputs (Ewert et al., 2015; Boote et al., 2015; Boote et al., 2013). In addition, model outputs other than crop yields and biomass, such as the representation of nitrogen cycling or carbon cycling, have not been adequately evaluated due to a lack of field data (Ewert et al., 2015). This uncertainty results in differences in the representation of processes between crop models. Generally, the consensus on the nature and magnitude of essential processes simulated in crop models needs to be improved (Challinor et al., 2017).

Another important source of uncertainty is information on climate and soil, which are often not available in the necessary spatial detail and thus adds uncertainty to crop model studies at all scales (Nendel et al., 2013). Climate change impact studies at field scale usually use downscaled climate change projections (Mullan et al., 2012; Vanuytrecht et al., 2016), adding bias and a significant source of uncertainty (Challinor et al., 2017). However, Asseng et al. (2013) found a greater uncertainty in simulating wheat yields under climate change due to variations among crop models than to variations among downscaled climate models. In large-scale analysis, soil data needs to be aggregated, which also has a strong effect on model outputs (Hoffmann et al., 2016). Although uncertainty of climate projections is often described as the major uncertainty in climate change impact studies, Folberth et al. (2016) found that soil-type-related yield variability can outweigh the simulated inter-annual variability in yields due to weather and can even determine if climate change impacts on crops are positive or

negative. Further, the lack of field management data causes uncertainty in global crop modelling studies. Whilst some spatially explicit dataset on field management and land use exist, such as crop-specific fertiliser rates (Mueller et al., 2012), planting dates and growing seasons (Sacks et al., 2010), irrigation area, amount and timing (Siebert et al., 2015), and crop-specific irrigation shares (Portmann et al., 2010), other management aspects are typically assumed to be static in space and time (Mueller et al., 2017). In addition, globally gridded datasets on field management and land use are usually simplified. For example, farmers may chose different crop varieties or sowing dates, which are subject to a number of weather-induced conditions (e.g., soil wetness, soil temperature) and the timely availability of labour and machinery (Mueller et al., 2017).

Although adaptation strategies involving tillage and residues management, planting dates, irrigation, crop cultivars and fertiliser are considered in some crop model studies, studies often fail to represent the broad scope of adaptation strategies present in the real world and are often limited to practices that can be simulated with confidence (Challinor et al., 2017). Moreover, most global impact studies assume optimal agronomic management (Rosenzweig et al., 2013), which may only reflect the most intensive cultivation systems. On the contrary, low input systems, managed by smallholders, are usually not well represented as common farming techniques such as crop diversification and intercropping are often not considered (Claessens et al., 2012; Hertel and Lobell, 2014).

Drivers and constraints of field management are highly diverse and complex, which can be influenced by gender, access to capital, access to extension services, access to climate information and water, tenure security and farming experience (Gbetibouo et al., 2010; Bryan et al., 2009). Therefore, model assumptions about field management require detailed information on environmental and socio-economic circumstance at a study location, which are often not available. The complexity of farm management is an important source of uncertainty in large-scale and global studies, in which farm management and adaptation can only be simulated in a simplified way and cannot achieve the level of spatial detail necessary to identify the costs and benefits of adaptation measures (Patt et al., 2010). Moreover, crop model studies often do not consider that farmers react to changing environmental and socio-economic conditions by adapting field management (Below et al., 2012; Quinn et al., 2011). In addition, crop modelling studies usually fail to account for technological development, which may lead to systematic underestimation of future crop yields (Challinor et al., 2017; Lobell, 2014). The increase of yields in past decades is largely driven by technological development such as improved crop varieties and increased nutrient application, particularly in Asia; thus, studies on future climate impact should ideally take such developments into account (Ewert et al., 2005). To approximate future management, integrated assessment studies are necessary, in which biophysical models are linked to economic models to account for interactions between socio-economic and biophysical factors (Ewert et al., 2015).

To test the adaptive potential of different farming techniques against climate change impacts, field management scenarios building on agronomic measures applied today can be created. However, the effectiveness of climate change adaptation strategies are uncertain without considering soil quality and soil health factors, which can determine the adaptive potential of farming techniques against climate change impacts (Webb et al., 2017). Moreover, climate change impacts vary across degraded and nondegraded lands with different levels of nutrient availability, infiltration rates, and soil moisture retention (Herrick et al., 2013). Although the gap in the investigation of land degradation factors, such as changes in soil organic carbon, erosion, runoff, and soil humidity, and their interaction with yields has been raised partly two decades ago (Feddema and Freire, 2001), uncertainties due to the lack of the representation of land degradation in crop models are still mostly neglected in recent climate change impact assessments.

One notable exception is Balkovič et al. (2018), who used an EPIC-based GGCM to bracket the uncertainty of climate change impacts on major crops in Europe driven by varying model assumptions and parameters for simulating land degradation through nutrient depletion. They concluded that in parts of Eastern Europe where fertiliser use is comparatively low, soil degradation may negatively outweigh positive impacts of +2 °C warming for certain crops when not prevented by adaptation measures. Since fertilisation is often the most important factor controlling the vulnerability of yields to degrading soil nutrients, unclear future fertiliser use due to unknown economic and political constraints contributes strongly to the uncertainty of crop yield projections. As increasing fertiliser rates is often presented as an effective measure to reduce climate change impacts (Rosenzweig et al., 2014; Balkovič et al., 2014; Folberth et al., 2013), neglecting the impact of land degradation processes on soil fertility leads to uncertainty in climate change impact assessments. Next to fertiliser management, most farming techniques used to mitigate climate change impacts interact with soil process such as nutrient cycles and degradation processes. For example, Folberth et al. (2014) projected potential climate change impacts on maize yields in Africa under different intensification options involving crop rotation options to replenish soil nutrients, while recognizing the need to account for land degradation processes in their study. Although Folberth et al. (2014) simulated land degradation processes such as soil erosion, they were not able to assess the robustness of their model outputs.

3.7 Discussion

Among the different crop model types, process-based models provide the most possibilities to explore the links between climate change and crop production. Process-based crop models coupled with the outputs of climate models can simulate the daily interactions between the growth and development of crops and changing climate parameters to explore the impact of climate change on crop yields. In addition, process-based crop models can be used to explore the effectiveness of climate change adaptation strategies, involving the timing and amount of irrigation and fertilisation or planting dates, for maintaining crop yields. Complex process-based crop models also simulate carbon, nutrient and water dynamics in the plant-soil-system and thus are useful tools to assess the sustainability of crop production systems. The main disadvantages of process-based crop models are the substantial data requirement and the complicated parameterisation of the various crop development processes. However, these have been reduced recently due to increasing availability of global datasets on soil and field management and the growing efforts to improve the application of models through model intercomparison projects such as AgMIP. The robustness and thoroughness of processes considered in crop production systems differs between models. Whilst numerous crop models exist, only a few models are commonly used in climate change adaptation or impact assessment studies. The most frequently used models to explore different links between climate change and crop production are EPIC, APSIM and DSSAT models (White et al., 2011). These models are also used in recent efforts to assess the impact of climate change on crops on a global scale (Mueller et al., 2017).

Climate change impact assessments based on GGCMs have provided insights on the effects of changes of the most important climate parameters on yield changes in different world regions. Crop models have been used most widely to study the geographic variations of long-term trends of major staple crops under different projected levels of warming. In addition, crop models have been used to assess crop yield changes due to increases in atmospheric CO_2 , changes in precipitation, as well as extreme rainfall and heat events. However, simulating the impact of varying climate change parameters on crop yields includes many uncertainties driven by input data or model structures. Whilst some of these uncertainties are addressed frequently and quantified using uncertainty ranges, others are hardly addressed.

Often, uncertainty analysis is focused on the response of crops to climate parameters, whereas many soil processes affected by changes in climate and biomass are neglected. Interactions between soil and biomass in response to climate influence the availability of nutrients for crops and thus play a crucial role in projecting crop yields. Moreover, these interactions play an important role in the impact of field management on crop yields, which is often a major focus in climate change adaptation studies. Without accounting for relevant soil processes influencing nutrient availability for crops, studies on the potential of adaptation strategies to maintain or increase projected crop yields are highly uncertain. Therefore, the integration of processes influencing soil quality in crop modelling studies will be crucial for climate change adaptation studies to better assess the long-term sustainability of agriculture under different field management scenarios.

Moreover, accounting for the degradation of soil quality will support the identification of cropland prone to declining productivity, which is a major objective in climate change impact assessments. As declining yields are often the consequence of several processes, trying to understand how agricultural systems are affected by several stressors simultaneously would be most realistic (Mittler, 2006). Further, interactions between climate change and land degradation may intensify negative impacts on agriculture (UNCCD, 2017). Whilst land degradation and climate change have been assessed extensively as separate phenomena, less is known about their interactions in different agroecological systems, and how land users must simultaneously adapt to their impacts (Reed and Stringer, 2016). Thus, climate change impact studies may underestimate a potential decline in crop yields if changes in soil quality in response to a changing climate are not considered. According to Webb et al. (2017), incorporating land degradation processes into systems analyses at all scales is needed to assess agroecosystem resilience, the agroecological and socioeconomic impacts of climate change, and scenarios by which agriculture may adapt to climate change.

Water erosion is among the land degradation processes most closely linked to climate change as it is directly linked to precipitation. Although water erosion has been addressed in some GGCM studies (Balkovič et al., 2018; Folberth et al., 2014), it is generally neglected in climate change impact assessment. Models used to simulate crops and water erosion are often similar in structure, and some models, such as EPIC, are frequently used for both crop and water erosion modelling. Due to the strong focus on soil processes in EPIC, it is one of the most suitable models among those often used in climate change impact assessments to simulate crop-water erosion interactions. Furthermore, EPIC has proven to be a reliable model that can be combined with a global data infrastructure to simulate changes in crop production on a global scale. Although these features of EPIC make it a suitable model for this study, there are many uncertainties and knowledge gaps in water erosion modelling that need to be addressed first (Chapter 2.3.4).

3.8 Conclusion

This literature review on crop modelling studies identified several knowledge gaps about the impact of climate change on crop yields. Among the many uncertainties, the lack of consideration of land degradation and its effects on the robustness of climate change impact and adaptation studies is rarely explored. This leads to important stress factors for crop yields being ignored. Moreover, the links between land degradation and climate change may significantly impact crop yields in the future, especially in the case of water erosion, which is closely linked to the climate system.

A major reason for this neglect is the difficulty of simulating land degradation processes such as water erosion and their impact on crops at large scale. As such, increasing the robustness of simulated crop-water erosion interactions to improve large-scale water erosion indicators, as discussed in Chapter 2, will also improve crop yield projections. By improving the representation of water erosion in GGCMs, the model results can be used to i) identify locations where crop productivity may decline due to the combined effects of water erosion and climate change, and ii) investigate the links between climate change and water erosion and their effects on crops. This analysis will be part of the third research objective of this thesis.

As many process-based crop models can simulate the daily interactions in the plant-soil system, including the dynamics of carbon, nutrients and water, they are useful tools for studying the effects of water erosion on crop yields. EPIC is a process-based crop model that has proven useful in both GGCM and water erosion studies. Thus, the GGCM EPIC-IIASA is suitable to fulfil the research objectives of this study.

Chapter 4

Overall methodology

4.1 Study framework

The framework in Figure 4.1 illustrates the modelling approach used to generate the results for this study, as well as the different steps to analyse the uncertainties, sensitivities and robustness of this approach. The input data are used to i) simulate daily maize and wheat growth and water erosion with EPIC and ii) analyse the sensitivity of the relevant model parameters to simulate global water erosion with all equations in EPIC. The output data is used to i) generate a baseline scenario for estimating global water erosion and maize and wheat yields and ii) address the uncertainty of simulated water erosion rates and impacts. The robustness of the model outputs is analysed using erosion values measured in the field and crop yields reported by FAOSTAT.



Figure 4.1: Scheme of the modelling framework for generating the model outputs and the different steps for analysing the uncertainties, sensitivities and robustness of the modelling approach

4.2 The EPIC crop model

This study uses the Environmental Policy Integrated Climate (EPIC), v. 0810 model. EPIC simulates the growth of a wide range of crops as a response to crop growing conditions (a combination of climate, soil and field management) and has a sophisticated representation of carbon, nutrient and water dynamics as well as a wide variety of possible field management options, including tillage operations and crop rotations (Izaurralde et al., 2006; Sharpley and Williams, 1990). EPIC operates with a daily time step and requires detailed input data on weather, topography, soil, crop, and crop management. Originally EPIC was named Erosion-Productivity Impact Calculator and was developed to determine the relationship between erosion and soil productivity. Due to its origin, EPIC has several options to simulate water erosion caused by precipitation, runoff and irrigation (Williams, 1990). Compared to other commonly used crop models in climate change impact assessments of the agricultural sector, EPIC has the most detailed representation of soil processes and the interactions between farming techniques and soil resources (Folberth et al., 2019). In addition to estimates on crop yields, EPIC generates outputs on nutrient consumption (N, P), components of water and soil organic carbon cycle, sediment loss, and loss of nutrients and soil organic carbon with sediments. EPIC is regularly used in both assessments of climate change impacts on global crop production and in assessments of farming techniques on water erosion rates.

4.2.1 Crop growth

The following routines are used in the EPIC model and are relevant for interpreting the results presented in this study. Their description is based on the original documentation of the EPIC model (Sharpley and Williams, 1990) and on the summary provided by Folberth et al. (2016, 2019).

In EPIC, phenological development of a crop is based on the heat unit (HU) approach. Daily HU are calculated as

$$HU_k = \frac{T_{max,k} + T_{min,k}}{2} - T_b \tag{4.1}$$

where HU_k are the accumulated HUs, maximum temperature and minimum temperature (°C) on day k, and T_b is the base temperature (°C) of a specific crop. The base temperature provides a crop-specific threshold under which no growth occurs. A crop reaches maturity when the sum of daily HU equals potential HUs (°C), which are based on long-term climate data and reported growing seasons provided by Sacks et al. (2010).

Daily potential biomass increase is estimated with the equation

$$\Delta B_p = 0.001 * BE * PAR \tag{4.2}$$

where ΔB_p (t ha⁻¹) is biomass gain, BE is a crop-specific coefficient for converting energy to biomass (kg MJ⁻¹) and PAR is intercepted photosynthetic active radiation based on leaf area index (LAI), solar radiation (MJ m⁻²) and plant row width. The daily potential biomass is subsequently adjusted to daily actual biomass by accounting for the maximum stress factor out of nutrient and water deficiency, temperature and aeration stress (further details below). The LAI of wheat and maize is initially zero or close to zero, increases exponentially during early vegetative growth and decreases after reaching a maximum. LAI is calculated as a function of heat units, crop stress, and crop development stages.

Total biomass is split between above and below-ground biomass. The fraction of total biomass partitioned to the root system usually decreases from emergence to maturity, which is simulated as a linear reduction in EPIC. Rooting depth usually increases rapidly from the seeding depth to a crop-specific maximum, which is based on heat units, crop-specific parameters and the soil profile depth. At maturity, crop yield is calculated by multiplying total aboveground biomass with a water stress-adjusted harvest index (HI_a). HI_a is estimated from simulated potential HI (HI_{max} ; depending on HU accumulation) and a defined minimum HI (HI_{min}) according to

$$HI_{a} = (HI_{max} - HI_{min}) * \frac{WUR}{WUR + exp(6.13 - 0.0883 * WUR)} + HI_{min}$$
(4.3)

where WUR is the water use ratio. WUR is estimated at harvest as

$$WUR = 100 * \frac{\sum_{i=1}^{K} U_i}{\sum_{i=1}^{K} E_{Pi}}$$
(4.4)

where $U_i (mm d^{-1})$ is the actual and $E_{Pi} (mm d^{-1})$ the potential plant water use rate for day *i*. *K* is the total number of days of the growing season.

4.2.2 Crop growth constraints

Potential crop growth and crop yields are constrained mainly by water, nutrients (N and P), temperature and aeration stress. The most severe stress factor on a given day limits biomass accumulation, root growth and yield by a fraction ranging from 0 to 1. The stress factors are calculated as follows.

Water stress is based on water deficit for the crop controlled by water supply and demand, which is consistent with the concept that drought stress limits biomass accumulation proportional to the transpiration reduction. It is calculated according to

$$WS_k = rac{\sum\limits_{i=1}^{M} u_{i,k}}{E_{Pk}}$$
 (4.5)

where WS_k is the amount of water stress on day k, l is a given soil layer, M is the total number of soil layers, $u_{l,k}$ is the plant available water in layer l on day k (mm), and E_{Pk} is the potential ET on day k (mm).

Temperature stress occurs if the air temperature is above the optimum temperature or below the base temperature of a specific crop according to

$$TS_k = \sin\left(\frac{\pi}{2} * \frac{TG_k - T_b}{T_o - T_b}\right) \tag{4.6}$$

where TS_k is the plant temperature stress factor on day k, TG_k is the soil surface temperature (°C) on day k, T_b is the base temperature, and T_o is the optimal temperature for a specific crop.

The nutrient stresses, Nitrogen (N) and Phosphorus (P) stress, are based on the ratio of accumulated plant N and P to the optimal values. N and P stress vary nonlinearly from 1 at optimal or excess N and P supply to 0, when N or P is half the optimal level. As a first step in determining nutrient stress, a scaling factor is calculated.

$$SNS_k = 2 * \left(\frac{UN_k}{c_{NBk} * B_k}\right) \tag{4.7}$$

where SNS_k is the scaling factor for calculating N stress on day k, UN_k is the N uptake on day k (kg ha⁻¹), c_{NBk} is the optimum N concentration of the crop on day k (kg ha⁻¹) and B_k is the total plant biomass on

day k (kg ha⁻¹). The scaling factor is used to calculate actual N stress (NS_k) using the equation

$$NS_{k} = \frac{SNS_{k}}{SNS_{k} + exp(3.39 - 10.93 * SNS_{k})}$$
(4.8)

The same equations used for the N stress factor apply for the P stress factor.

Aeration stress occurs when soil water content approaches saturation. To estimate the degree of stress, the water content and pore volume of the top 1 m of soil is considered with the following equation

$$SAT_k = \frac{SW1_k}{PO1_k} - CAF \tag{4.9}$$

where SAT_k is the saturation factor at day k, $SW1_k$ is the water content of the top 1 m of soil (mm) at day k. $PO1_k$ is the porosity of the top 1 m of soil (mm) at day k and CAF is the critical aeration factor for a specific crop, which can vary between 0 and 1. If $SAT_k > 0$, aeration stress (AS) is calculated as

$$AS_k = 1 - \frac{SAT_k}{SAT_k + exp(-1.291 - 56.1 * SAT_k)}$$
(4.10)

Crop growth can be additionally constrained through stresses affecting root development due to soil strength, aluminium toxicity and temperature stress.

Soil strength depends on bulk density and soil sand content according to

$$SS_{l,k} = 0.1 + \frac{0.9 * BD_{l,k}}{BD_{l,k} + exp(bt_1 + bt_2 * BD_{l,k})}$$
(4.11)

where $SS_{l,k}$ is the soil strength of layer *l* on day *k*, $BD_{l,k}$ is the bulk density (g cm⁻³) of layer *l* on day *k* and bt_1 and bt_2 are coefficients based on the soils sand content.

Aluminium toxicity stress is estimated if the crop-specific aluminium tolerance is lower than the amount of dissolved aluminium controlled by the soils pH according to

$$ATS_{k} = \frac{100 - ALS_{l,k}}{100 - (10 + 20 * (ALT - 1))}$$
(4.12)

where ATS_k is aluminium toxicity stress on day k, ALT is a crop-specific aluminium tolerance index that can vary between 1 (=highly sensitive) and 5 (=very tolerant) and $ALS_{l,k}$ is the amount of dissolved aluminium in soil layer l on day k.

4.2.3 Soil

The soil organic matter (SOM) model in EPIC is based on the Century model (Izaurralde et al., 2006). SOM is split into different compartments, which differ in size, function and turnover rates: standing dead residue and roots, structural and metabolic litter, slow humus and passive humus and microbial biomass. The coupled organic C and N module in EPIC distributes organic C and N between the different soil organic matter compartments. The system interacts directly with soil moisture, temperature, erosion, tillage, soil density, soil texture, leaching, and translocation functions. Erosion, surface runoff, leaching and volatilization can lead to the loss of C, N and P from the system.

In EPIC, the main impact of water erosion on crops is driven by nutrient stress through the export of organic C, organic N and P from the topsoil layer through sediment runoff. The loss of all three nutrients with sediment runoff is estimated in relation to the eroded soil after a common approach used in SOM models (Polyakov and Lal, 2004; Sharpley and Williams, 1990).

$$M_{loss} = E * M_{soil} * ER \tag{4.13}$$

where M_{loss} is the runoff loss of either organic C, organic N or P (kg ha⁻¹); *E* is soil loss (Mg ha⁻¹); M_{soil} is the concentration of either organic C, organic N or P in the top soil layer (gt^{-1}) ; *ER* is the enrichment ratio, which is the concentration of either organic C, organic N or P in the sediment relative to the original soil (dimensionless). The enrichment ratio in sediment is logarithmically related to sediment loss.

4.2.4 Hydrology

Daily precipitation is split at the surface into runoff and infiltration. Daily runoff Q (mm) is estimated according to the USDA SCS curve number equation

$$Q = \frac{(R - 0.2s)^2}{R + 0.8s}; if R > 0.2s$$
(4.14)

where *R* is precipitation (mm) and *s* is the retention parameter (mm). If $R \le 0.2s$, no runoff occurs. Parameter *s* depends on the curve number (*CN*) according to

$$s = 254 * \frac{100}{CN} - 1 \tag{4.15}$$

The CN reflects different types of soils, landuse and management, and are adjusted for slope, surface roughness, and soil humidity in the model. The curve number can be obtained by using the SCS hydrology handbook (USDA, 1972).

4.2.5 Water erosion

EPIC includes seven empirical equations to calculate water erosion (Wischmeier and Smith, 1978). The basic equation is:

$$E = P * K * LS * C * P \tag{4.16}$$

where *E* is soil erosion in Mg ha⁻¹ (mass/area), *R* is the erosivity factor (erosivity unit/area), *K* is the soil erodibility factor in Mg MJ^{-1} (mass/erosivity unit), *LS* is the slope length and steepness factor (dimensionless), *C* is the soil cover and management factor (dimensionless) and *P* is the conservation practices factor (dimensionless).

The main difference between the water erosion equations available in EPIC is their energy components used to calculate the erosivity factor. The USLE, RUSLE and RUSLE2 equations use precipitation intensity as an erosive energy to calculate the detachment of soil particles. The MUSLE equation and its variations MUST and MUSS use runoff variables to simulate water erosion and sediment yield. The Onstad-Foster equation combines energy through rainfall and runoff (Table 4.1).

Table 4.1: Equations to calculate the erosivity factor in each water erosion equation available in EPIC.

Erosivity factor	Equation
R = EI	USLE, RUSLE, RUSLE2 (Renard et al., 1997; Wischmeier and Smith, 1978: USDA-ARC, 2013)
$R = 0.646 * EI + 0.45 * (Q * Q_p)^{0.33}$	Onstad-Foster (Onstad and Foster, 1975)
$R = 1.586 * (Q * Q_p)^{0.56} * WSA^{0.12}$	MUSLE (Williams, 1975)
$R = 2.5 * Q * q_p^{0.5}$	MUST (Williams, 1995)
$R = 0.79 * (Q * q_p)^{0.65} * WSA_{0.009}$	MUSS (Williams, 1995)

The erosion energy component is calculated as a function of either runoff volume Q (mm), peak runoff rate q_p (mm h⁻¹) and watershed area WSA (ha), or via the rainfall erosivity index EI (MJ ha⁻¹). The latter determines the detachment of soil particles through the energy of daily precipitation and a statistical estimate of the daily maximum intensity of precipitation falling within 30 minutes. RUSLE2 is the only equation calculating soil deposition. If the sediment load exceeds the transport capacity, determined by a function of flow rate and slope steepness, soil is deposited, which is calculated by a function of flow rate and particle size (USDA-ARC, 2013).

The soil cover and management factor is updated for every day where runoff occurs using a function of crop residues, biomass cover and surface roughness.

$$C = FRSD * FBIO * FRUF \tag{4.17}$$

where *FRSD* is the crop residue factor, *FBIO* is the growing biomass factor and *FRUF* is the soil random roughness factor, which are calculated with the following equations (Wang et al., 2011; Williams et al., 2012):

CTI

$$FRSD = exp(-P23 * CVRS) \tag{4.18}$$

$$FBIO = 1 - \frac{STL}{(STL + exp(SCRP1(23) - SCRP2(23) * STL))} * exp(-P26 * CPHT)$$
(4.19)

$$FRUF = exp(-0.026 * (RR - 6.1))$$
(4.20)

where *P*23 is an exponential coefficient ranging from 0.01-0.5, *CVRS* is the amount of above ground crop residue (t ha⁻¹), *STL* is the amount of standing live biomass of the crop (t ha⁻¹), *SCRP*1(23) and *SCRP*2(23) are coefficients defining an S-shaped growth curve used to estimate the fraction of the ground covered by the plant as a function of the Leaf Area Index, *P*26 is an exponential coefficient ranging from 0.01-0.2, *CPHT* is the crop height (m) and *RR* is the soil surface random roughness (mm).

Although the root system of plants is important for erosion through its regulation of soil infiltration and its stabilising effect on soil structure, it is not directly considered in the erosion calculation in EPIC. The root system is however included in the calculation of crop growth (Chapter 4.2.1) and soil organic matter (Chapter 4.2.3) that directly affect erosion.

The impact of soil erodibility on simulated water erosion is calculated for the top-soil layer at the start of each simulation year. The soil erodibility factor is calculated the same way for the USLE, Onstad-Foster, MUSLE, MUST and MUSS equation using a function of sand, silt, clay and organic carbon contents in the soil:

$$K = X1 * X2 * X3 * X4 \tag{4.21}$$

$$X1 = 0.2 + 0.3 * exp(-0.0256 * SAND * (1 - 0.01 * SILT))$$
(4.22)

$$X2 = \left(\frac{SILT}{CLAY + SILT}\right)^{0.3} \tag{4.23}$$

$$X3 = \frac{1 - 0.25 * OC}{OC + exp(3.718 - 2.947 * OC)}, IF OC \le 5$$
(4.24)

$$X3 = 0.75, IF \ OC > 5 \tag{4.25}$$

$$X4 = \frac{1 - 0.7 * SN1}{SN1 + exp(-5.509 + 22.899 * SN1)}$$
(4.26)

$$SN1 = 1 - 0.01 * SAND$$
 (4.27)

Where *SAND*, *SILT*, *CLAY*, and *OC* are the sand, silt, clay, and organic carbon contents of the soil (%). For the RUSLE and RUSLE2 method soil erodibility is calculated without the organic carbon contents of the soil using the following equation:

$$KR = 9.811 * \left(0.0034 + 0.0405 * exp\left(-0.5 * \left(\frac{Log10(DG) + 1.659}{0.7101} \right)^2 \right) \right)$$
(4.28)

$$DG = exp(SUM)$$
(4.29)
SAND * 0.0247 - SILT * 3.65 - CLAY * 6.908

$$SUM = \frac{SAND * 0.0247 - SILI * 3.65 - CLAY * 6.908}{100}$$
(4.30)

The topographic factor is calculated the same way for the USLE, Onstad-Foster, MUSLE, MUST and MUSS equation using a function of slope length and slope steepness:

$$LS = \left(\frac{SLPL}{22.127}\right)^{XM} * (SLP * (65.41 * SLP + 4.56) + 0.065)$$
(4.31)

$$XM = 0.3 * \frac{SLP}{SLP + exp(-1.47 - 61.09 * SLP)} + 0.2$$
(4.32)

Where *SLPL* is the slope length (m), *SLP* is the land surface slope (m/m) and *XM* is an exponent dependent upon slope. The topographic factor for the RUSLE method is calculated using a function of slope length and slope steepness as well:

$$LSR = RSF * RLF \tag{4.33}$$

$$RSF = 10.8 * SLP + 0.03, IF SLPL > 4.57 \& SLP < 0.09$$
(4.34)

$$RSF = 16.8 * SLP - 0.5, IF SLPL > 4.57 \& SLP > 0.09$$
(4.35)

$$RSF = X1, IF SLPL < 4.57$$
 (4.36)

$$X1 = 3 * SLP^{0.8} + 0.56 \tag{4.37}$$

$$RLF = \frac{SLPL^{RXM}}{22.127} \tag{4.38}$$

$$RXM = \frac{B}{1+B} \tag{4.39}$$

$$B = \frac{SLP}{0.0894 * X1} \tag{4.40}$$

Where *SLPL* is slope length (m) and *SLP* is land surface slope (m/m). The slope steepness factor *RSF* is adjusted for different slope steepness and slope length thresholds based on experimental data (Renard et al., 1997). The slope length factor *RLF* includes an exponent *RXM*, which is a function of the ratio *B* of rill erosion caused by flow and interill erosion caused by raindrop impact (USDA-ARC, 2013). *B* reflects how steepness affects rill erosion differently than it does interrill erosion. Rill erosion is assumed to vary linearly with steepness. The topographic factor for the RUSLE2 method is calculated the same way as for the RUSLE equation if the transport capacity determined by a function of flow rate and slope steepness exceeds sediment load. When sediment load exceeds transport capacity, RUSLE2 computes deposition. Interrill erosion is assumed to occur even when RUSLE2 computes deposition, which can be calculated without a distance term as detachment is solely caused by impacting raindrops (USDA-ARC, 2013). Therefore, the slope length factor is not considered in the RUSLE2 equation when deposition occurs.

The conservation practice factor is included in all equations as a static coefficient ranging between 0 and 1, where 0 represents conservation practices that prevent any erosion and 1 represents no conservation practices. Typical conservation practice factors can be derived from tables, which include values ranging from 0.01 to 0.35 for terracing strategies and from 0.25 to 0.9 for different contouring practices (Morgan, 2005; Wischmeier and Smith, 1978). Alternatively, values can be derived from local field studies and remote sensing (Karydas et al., 2009; Panagos et al., 2015), from equations using topographical data (Fu et al., 2005; Terranova et al., 2009), or from economic indicators (Scherer and Pfister, 2015).

4.3 EPIC-IIASA

GGCMs are integrated modelling frameworks for process-based crop modelling at large scale. Global datasets on static (topography, soil texture) and dynamic (climate, crop management, soil nutrient and

water pools) factors are organised within geographical grids to provide inputs for process-based crop models to generate spatially-explicit estimates of crop yields over the globe. EPIC-IIASA is a GGCM based on EPIC, which has been used as a tool for climate change impact studies on food security and the agricultural sector (Mueller et al., 2017; Balkovič et al., 2018).

4.3.1 Global grids and input data

EPIC-IIASA requires global soil and topography data and daily weather data. The basic spatial resolution of the model is 5' x 5' at which soil and topographic data are provided. These are aggregated to homogenous units (based on the same topography and soil classes) and further intersected with a 30' x 30' climate grid, the resolution at which global gridded climate data are available. This results in a total of 131,326 grid cells with a spatial resolution ranging between 5' to 30' (about 9 km to 56 km near the equator) depending on landscape heterogeneity (Skalský et al., 2008). Weather parameters for the simulation period 1980 - 2010 are taken from historic bias-corrected daily weather data combining data from the Modern-Era Retrospective Analysis for Research and Applications (MERRA), station data, and remotely-sensed datasets, covering the years 1980 - 2010 (Ruane et al., 2015). Weather parameters for representative concentration pathways (RCP) are taken from the HadGEM2 RCP 2.6 and 8.5 datasets covering the years 2005 - 2099 (Hempel et al., 2013). These datasets represent two contrasting climate scenarios and are used to bracket varying projections of weather parameters. Soil information is taken from the Harmonized World Soil Database (FAO/IIASA/ISRIC/ISSCAS/JRC, 2012), and topography from USGS GTOPO30 (USGS, 1997). Each grid cell is represented by a single field characterized by the combination of topography and soil conditions prevailing in this landscape unit. Each representative field has a defined slope length (20 - 200 m) and field size (1 - 10 ha) based on a set of rules for different slope classes (Table 4.2). The slope of each representative field is determined by the slope class covering the largest area in each grid cell (Table 4.2). Slope classes are taken from a global terrain slope database (IIASA/FAO, 2012) and are based on a high-resolution 90 m SRTM digital elevation model (Figure A.1).

Table 4.2: Set of rules for field size and slope length estimation for each dominant slope class. The area/dominant slope class was assigned to each grid cell from a global slope and terrain dataset (IIASA/FAO, 2012) providing 3 arc-sec spatial resolution distributions of nine slope gradient classes: 0-0.5%, 0.5-2%, 2-5%, 5-8%, 8-16%, 16-30%, 30-45%, and > 45% interpreted from SRTM elevation data (CGIAR-CSI, 2006). The mid-interval value of the dominant slope class was used as an input for each grid cell in EPIC-IIASA.

Dominant slope class	Lower value (%)	Upper value (%)	Mid value (%)	Slope length (m)	Field size (ha)
1	0	0.5	0.25	200	10
2	0.5	2	1.25	200	10
3	2	5	3.5	200	10
4	5	8	6.5	200	10
5	8	16	12	100	5
6	16	30	18	75	5
7	30	45	35.5	50	1
8	45	100	60	20	1

4.3.2 Climate change scenarios

The two contrasting RCPs used for the climate inputs for EPIC-IIASA represent different socioeconomic and emission scenarios and were produced in collaboration between integrated assessment modelers, climate modelers, terrestrial ecosystem modelers and emission inventory experts (van Vuuren et al., 2011). The scenarios are created to improve the understanding of complex interactions of the climate system, ecosystems, and human activities and conditions to provide plausible descriptions of how each of these components might change in the future. However, the goal of scenarios is not to predict the future, but to better understand uncertainties to reach decisions that are robust under a wide range of possible futures. As such, these scenarios support the analysis of complex issues such as costs, benefits and risks of different policy choices and climate and socioeconomic futures (Moss et al., 2010). In total, a set of four pathways were produced that lead to radiative forcing levels of 8.5, 6, 4.5 and 2.6 W m⁻², by the end of the 21st century (Moss et al., 2010). The radiative forcing levels are used as inputs for climate models to project daily climate parameters covering the years 2005 - 2099. The RCPs selected for this study are based on climate data from the HadGEM2-ES model (Collins et al., 2008). The data are available via the CMIP5 website (https://pcmdi.llnl.gov/mips/cmip5/) and were further bias-corrected with respect to historical observations by Hempel et al. (2013) in line with the ISI-MIP project.

RCP 2.6 was developed by the IMAGE modelling team of the PBL Netherlands Environmental Assessment Agency to represent the scenarios that lead to very low greenhouse gas concentration levels (van Vuuren et al., 2011). The radiative forcing level in RCP 2.6 peaks at a value of around 3.1 W m⁻² and a concentration of 490 ppm CO2- equiv. by mid-century, and returns to 2.6 W m⁻² by the year 2100; thus, it is referred to as a "peak-and-decline" scenario. To reach such radiative forcing levels, greenhouse gas emissions (and indirectly emissions of air pollutants) are reduced substantially over time (van Vuuren et al., 2011).

RCP 8.5 was developed using the MESSAGE model and the IIASA Integrated Assessment Framework (Riahi et al., 2007). The radiative forcing level in the RCP 8.5 reaches more than 8.5 W m⁻² and a concentration of more than 1370 ppm CO2-equiv. in 2100. It represents scenarios for increasing greenhouse gas emissions over time leading to the highest greenhouse gas concentration levels (van Vuuren et al., 2011).

4.3.3 Field management scenarios

Maize and wheat are chosen as two contrasting crop types for analysing water erosion in different cultivation systems. Maize is a row crop with relatively large areas of bare and unprotected soil between the crop rows. The plant density in wheat fields is much higher, which improves the protection of soils against water erosion. Growing seasons for maize and wheat in each grid cell are taken from Sacks et al. (2010), and spatially explicit nitrogen and phosphorus fertiliser application rates are taken from Mueller et al. (2012).

In the absence of spatial data for crop-specific water application rates at the global level, irrigated crops are provided with sufficient amounts of water to avoid water stress. Irrigation water is supplied to

refill soil water content to its field capacity each time the plants are under stress, with an annual limit of 1000 mm per crop (Balkovič et al., 2018).

Field management techniques influencing soil properties and soil cover have a significant impact on the amount of water erosion. However, these methods vary greatly around the world and data on different field management techniques is sparse. Therefore, each crop is simulated for six field management scenarios (three tillage x two cover crop scenarios), each influencing soil properties, water erosion and plant growth differently. The tillage management scenarios represent conventional, reduced and notillage, which differ by the amount of cultivation operations, tillage depth, mixing efficiency of tillage and sowing mechanisations, surface roughness and the amount of plant residues left on the field after crop harvest. Further, runoff curve numbers are altered for each tillage scenario to account for different runoff intensities for the cover treatment classes presented in Table 4.3. Runoff curve numbers indicate the runoff potential of a hydrological soil group, land use and treatment class and allow to take the impact of different tillage practices on the hydrologic balance into account (Chung et al., 1999). The different tillage intensities account for the impact of gradually changing surface cover and roughness on water erosion rates. Each tillage scenario is simulated with and without green fallow (grass-type green fallow) cover in between growing seasons, leading to a total of six field management scenarios.

	Conventional tillage	Reduced tillage	No- tillage
total cultivation operations	6-7	4-5	3
max. tillage depth	150 mm	150 mm	40-60 mm
mixing efficiency	99 %	75 %	2 %
max. surface roughness	30-50 mm	20 mm	10 mm
plant residues left	25 %	50 %	75 %
cover treatment class	straight	contoured	contoured & terraced

Table 4.3: Tillage management scenarios for maize and wheat cultivation.

Irrigation and additional conservation practices in all field management scenarios are based on the underlying slope class of each grid cell. On slopes steeper than 5%, only rainfed agriculture is considered, as hilly cropland is irrigated predominantly on terraces that prevent water runoff.

P-factors are used to simulate additional, or more efficient conservation practices. These are static coefficients ranging between 0 and 1, where 0 represents conservation practices that prevent any erosion and 1 represents no conservation practices. Whilst conservation practices are introduced implicitly through various crop growth assumptions as presented in Table 4.3, additional erosion control measures are simulated on slopes > 16% assuming that conservation practices are most likely implemented on steep slopes. On slopes steeper than 16%, a P-factor of 0.5 is used, and on slopes steeper than 30%, a P-factor of 0.15 is used to simulate contouring and terracing based on the range of P-values presented in Morgan (2005).

To determine the impact of water erosion on maize and wheat yields, all field management scenarios are additionally simulated with no erosion (P=0). The comparison of crop yields simulated with a P-factor value of zero with crop yields simulated under higher P-factor values can be used to identify grid

cells where crop yields are reduced by water erosion. The simulation outputs at those grid cells are used to quantify the reduction of maize and wheat production and the relative reduction of crop yields due to water erosion.

4.3.4 Baseline field management scenario

The six field management scenarios reflect a range of potential impacts occurring due to different farming techniques on water erosion rates and erosion–crop yield relationships. To account for geographic variations in field management while addressing each thesis objective, the model outputs are aggregated using a baseline scenario determined by climatic and country-specific indicators.

As the only global statistical data on the type of tillage systems are provided for the extent of Conservation Agriculture area at the national scale (FAO, 2016), only the lowest tillage intensity scenario is assigned to specific countries in the baseline scenario. Therefore, conventional and reduced tillage are simulated in each grid cell globally, whereas the additional no-tillage scenario is simulated only for countries in which at least 5% of cropland is cultivated under conservation agriculture according to AQUASTAT (2007–2014) (FAO, 2016).

The main Koeppen-Geiger regions (Peel et al., 2007) were used to avoid unsuitable cover crop options. The off-season green fallow cultivation is not simulated in dry regions due to high competition for water. In contrast, the off-season green fallow is always considered in tropical areas with high rainfall, where a year-round growing season is common. In temperate and snow regions, average simulation results from both cover crop scenarios are used (Figure 4.2).

The MUSS equation is chosen for the baseline scenario as it generates the lowest deviation between simulated and measured water erosion, which will be discussed in Chapter 5. Table 4.4 summarises the field management assumptions.



Figure 4.2: Distribution of selected (a) tillage and (b) cover crop scenarios determined by countryspecific and environmental indicators used for compiling the baseline scenario. In each grid cell, crop yields are averaged over all selected field management scenarios.

TT 1 1 4 4 C	C	1 .		1 1' '
Table 4.4. Sum	nmary of management as	ssumptions and erosion	equation selected for the	baseline scenario
Iuoie I. I. Dull	innury of munugement a	soumptions and croston	. equation beleeted for the	ousernie seenurio.

Option	Baseline
Tillage	• Mix of conventional, reduced and no-tillage in regions where the national share of conservation agriculture is > 5% according to the latest reported data in FAO AQUASTAT (2007-2014) (FAO, 2016): Argentina, Australia, Bolivia, Brazil, Canada, Chile, China, Colombia, Finland, Italy, Kazakhstan, New Zealand, Paraguay, Spain, USA, Uruguay, Venezuela, Zambia and Zimbabwe.
	• Mix of conventional and reduced tillage in the rest of the world.
Off-season	• Cultivation only with cover crops in tropics according to Koeppen-Geiger regions (Figure A.4) (Peel et al., 2007).
cover	• No cover crops in arid regions.
	• Mix of off-season cover with and without cover crops in temperate and snow zones.
Conservation	• $P = 1, IF SLOPE < 16\%$
Practice	• $P = 0.5$, IF SLOPE $\geq 16\%$ & SLOPE $< 30\%$
Factor (P)	• $P = 0.15$, IF SLOPE $\ge 30\%$
Crop	• Model outputs are generated for wheat and maize fields based on the global crop distribution by MIRCA2000 (Figure A.3) (Portmann et al., 2010).
Irrigation	• irrigation if $SLP \leq 5\%$.
Ingation	• Weighted average of irrigated and rainfed cropland based on Portmann et al. (2010).
Water erosion equation	• MUSS

4.4 Uncertainty analysis of model outputs

The large spatial resolution of GGCMs causes uncertainty from various input sources including climate, soil, field management, distribution of crop cultivars and cropland, irrigation area, growing seasons, as well as model structure and model parametrisation, most of which have been addressed by prior studies (Folberth et al., 2016, 2019; Mueller et al., 2017; Porwollik et al., 2017). Since not all uncertainties can be considered, the most important ones for the simulation of water erosion in EPIC are analysed. This includes the uncertainties due to different field management scenarios, different water erosion equations, and slope steepness data.

Field management practices have a major influence on the extent of water erosion. However, their application is not well known worldwide, which could lead to errors in the model results. Further, the different water erosion equations in EPIC differ in their parameters and sensitivities, which causes varying water erosion estimates. To account for both sources of uncertainty, uncertainty ranges are calculated globally in Chapter 5 based on the varying water erosion estimates with different water erosion equations and field management scenarios. In addition, uncertainty ranges are calculated globally for the estimated impacts of water erosion on crops in Chapter 6 based on different field management scenarios.

While the range of slopes and the fraction of cropland in each grid cell is known, it is not possible to determine how much land in each slope class is cultivated. Therefore, it is assumed that the cropland in each grid cell is on the slope class that is most common in the grid cell, as this represents the prevailing topographical conditions. This assumption is likely to introduce spatially-varying uncertainty as the fraction of each grid cell containing the dominant slope category varies from 20% to 100%, with an average share of 48%. The share of land covered by cropland in each grid cell also varies greatly, from 1% to 100%, with an average share of 14% (Figure A.5). Therefore, the extrapolation of the simulation outputs to the entire cultivated area in a grid cell can provide only a rough estimate. The implications of this assumption are explored in Chapter 5 and Chapter 6 by comparing the simulation results to a second set of simulation outputs based on an ideal cropland distribution scenario, in which the flattest terrain available rather than the most common slope in each grid cell is cultivated. This assumes that farmers would prefer to cultivate flatter land where possible.

The uncertainty from slope steepness data will be addressed in an example region, as this requires a large number of additional model runs for various combinations of slope assumptions and field management scenarios per grid cell. Italy is chosen as an example region, as it is susceptible to water erosion and includes large and heterogeneous maize and wheat cultivation areas on flat terrain in the north and mountainous regions in the south.

4.5 Sensitivity analysis of water erosion simulation

The most essential input parameters to the factors in the seven water erosion equations are identified with a sensitivity analysis in Chapter 5. The Sobol method (Sobol, 1990), a variance-based sensitivity analysis, is popular in environmental modelling (Nossent et al., 2011). With this method, it is possible to

quantify the amount of variance that each parameter contributes to the total variance of the model output. These amounts are expressed as sensitivity indices, which rank the importance of each input parameter for simulated water erosion. In addition, the sensitivity indices can be used to determine the impact of parameter interactions on the model output.

The sensitivity analysis is run for 30 parameters directly connected to the water erosion equations in EPIC. In total, 126,976 random values are assigned to all input parameters along a pre-defined triangular distribution or a range of discrete values (Table A.1). Water erosion is simulated with EPIC using the seven available equations for each random input combination at 40 locations where wheat and maize are cultivated. To represent a heterogeneous distribution of global precipitation regimes, the natural break optimisation method is used to choose locations based on average annual precipitation amounts from 1980 to 2010 (Jenks, 1967). For each location and equation, the most sensitive parameters are ranked. To analyse the impact of precipitation regimes on the sensitivity of each parameter, Spearman coefficients (ρ) are used to determine if positive or negative relationships exist between each parameter's sensitivity and annual precipitation.

4.6 Evaluation of model outputs

4.6.1 Evaluation of simulated water erosion against field measurements

Simulated water erosion rates are compared with 606 soil erosion measurements on arable land from 36 countries representing plot and field scale in Chapter 5. Most of the selected erosion rates are based on the ¹³⁷Cs method. In addition, data from erosion plots and volumetric measurements of rills collected by Auerswald et al. (2009), Benaud et al. (2020) and García-Ruiz et al. (2015) are used. In total, 315 records are derived by the ¹³⁷Cs method, 188 records from runoff plots, and 103 records from volumetric measurements of rills. An overview of the field data is presented in Figure A.6 - A.9, and the full dataset is available in Carr et al. (2020).

The overwhelming effect of the experimental methodology on measured erosion rates (Chapter 2.3.5), the lack of sufficient metadata accompanying erosion measurements and the granular spatial resolution of the simulation setup hinders a direct comparison between simulated and observed water erosion rates. Instead aggregated simulated and observed erosion values are compared for different slope and precipitation classes to analyse the robustness of simulated water erosion rates under different environmental conditions. Therefore, only measurements with recorded slope steepness and annual precipitation are used. Where annual precipitation is not recorded, it is taken from the WorldClim2 dataset (Fick and Hijmans, 2017).

4.6.2 Evaluation of simulated crop yields against reported yields

Maize and wheat yields simulated under the baseline field management scenario are evaluated against FAOSTAT reported yields per country in Chapter 6. As simulated crop yields are calculated in t ha^{-1} dry matter in EPIC, they are converted to fresh matter assuming a net water content of 12% following

Wirsenius (2000), so that they can be compared with yields reported by FAOSTAT (FAO, 2020). In addition, calculated total crop production per country are compared against FAOSTAT statistics in Chapter 6. Simulated crop yields and total production per country are compared against FAOSTAT statistics for the years 1995–2005. The years are chosen based on the years of reported fertiliser application rates that are used to simulate maize and wheat yields. The agreement between simulated and reported data is determined by the coefficient of determination (R^2) and the relative error (%) between both datasets.

4.7 The computation process

The IIASA computer platform was used to run EPIC-IIASA to simulate water erosion and maize and wheat for each grid cell and for each combination of climate and field management scenarios. A total of 288 scenarios were created.

Model outputs were aggregated using the programming language R (Version 4.0.5) and RStudio (Version 1.4.1106). The aggregation method differs for the different results and is explained in the Methodology section of Chapters 5, 6, and 7.

Chapter 5

Uncertainties, sensitivities and robustness of simulated water erosion with EPIC-IIASA

5.1 Introduction

Water erosion models such as the USLE and the RUSLE have been widely used to estimate global water erosion (Borrelli et al., 2017; Doetterl et al., 2012; Van Oost et al., 2007). However, a comprehensive evaluation of erosion estimates is lacking in the previous studies. Because the USLE and its modifications were developed using field data from the Midwestern United States, their outputs should ideally be evaluated against soil erosion measurements from other agro-environmental zones when applied in those environments (Evans and Boardman, 2016b). However, the uneven distribution of field data around the world, the lack of long-term soil measurements in most global regions, and the great variability of the designs of erosion rate measurements hamper the evaluation of global soil loss estimates derived from models (Auerswald et al., 2004; Borrelli et al., 2017; García-Ruiz et al., 2015). In addition, model input data on topography, soil properties and land use are often aggregated over large areas and thus simulation results cannot be directly compared to single field measurements at specific locations. Thus, evaluating global erosion models is difficult on a global scale, but improvements are needed due to their widespread application (Chapter 2.3.1).

Beyond improving the robustness of global erosion models, there is a need to consider water erosion in GGCMs (Chapter 3.6). Throughout the past decade, GGCMs have become essential tools for climate change impact assessments, evaluations of agricultural externalities, and as input data providers for agroeconomic models (Mueller et al., 2017). However, few assessments have considered land degradation processes and found their inclusion and understanding crucial for evaluating climate change mitigation and adaptation strategies (Balkovič et al., 2018; Chappell et al., 2016).

Moreover, crop models can be used to simulate water erosion in agricultural fields more realistically

by considering the daily interactions between crops and water erosion. Most global soil removal estimates using water erosion models are based on static observation approaches or on very coarse timescales that do not fall below annual time steps (Borrelli et al., 2017). Therefore, seasonal patterns of soil cover and precipitation intensities are neglected even though they are crucial factors for water erosion. The state of soil and its cover is influenced by land management, such as the choice of crops, planting and harvest dates, tillage and plant residue management. Accordingly, neglecting the impact of seasonal changes in vegetation cover and field management practices constitutes large uncertainty in global water erosion estimates. Crop models usually simulate crop growth on a daily timescale, which allows attached water erosion models to account for daily changes in weather, soil properties and vegetation cover. However, uncertainty remains due to the increasing requirement of input data for daily simulations, which is especially challenging at a global scale (Chapter 2.3.4).

Given the difficulties of simulating water erosion on large scales, this chapter will focus on the uncertainties, sensitivities and robustness of simulated water erosion in EPIC-IIASA. First, the global water erosion estimates and their uncertainties due to different estimates under different field management scenarios and water erosion equations are presented. The sensitivities of the water erosion equations are then analysed to discuss the main causes of variability in the simulation results. Finally, the simulated erosion values will be compared with field data to investigate the robustness of the water erosion estimates in all global agro-environmental regions simulated with EPIC-IIASA.

5.2 Methodology

5.2.1 Summary of methodology

Water erosion in maize and wheat fields is simulated with seven different empirical erosion equations in EPIC under different field management scenarios. Global water erosion is estimated from the baseline field management scenario based on the environmental and country-specific assumptions and indicators presented in Chapter 4.3.4.

The uncertainties of the estimated water erosion rates due to the different field management scenarios are calculated for each grid using the range between the minimum and maximum water erosion rates simulated with all field management scenarios. Similarly, the range of uncertainty due to different water erosion equations per grid is calculated using the minimum and maximum water erosion rates simulated with all available equations.

The sensitivity of the model to all inputs used in the simulation of water erosion is calculated using the method presented in Chapter 4.5. Sensitivity indices are used to interpret the variability and uncertainties of the simulation results and to discuss the differences between the water erosion equations.

To evaluate the robustness of the simulated water erosion rates, they are compared to field measurements for different slope and precipitation classes (Chapter 4.6.1). Due to the non-normal distribution of the simulated and measured data, the median deviation (MD) is used as a measure to compare the agreement between aggregated simulated and measured water erosion values.

5.2.2 Aggregation of model outputs

Simulated water erosion rates are averaged over the historic simulation period (1982 - 2010). Subsequently, water erosion rates simulated for irrigated and rainfed maize and wheat fields are averaged per grid cell and weighted by the area of the respective crop and irrigation scenario (Equation 5.1). Finally, water erosion rates are presented as median values of all baseline field management scenarios per grid cell or of all grid cells per country, region or globally.

$$Ew_{cgm} = Eav(r)_{cgm} * Af(r)_{cg} + Eav(i)_{cgm} * Af(i)_{cg}$$
(5.1)

 Ew_{cgm} is area-weighted mean water erosion (tha^{-1}) for crop c, grid cell g and management scenario m; Eav is yield averaged over the selected years simulated under rainfed (r) and irrigated (i) conditions; Af(r) is the rainfed area fraction; and Af(i) is the irrigated area fraction.

5.3 Results

Global median water erosion rates are 7 t ha⁻¹ and 5 t ha⁻¹ in maize and wheat fields, respectively. The total removal of soil in global maize and wheat fields is estimated to be 5.3 Gt yr⁻¹ and 1.9 Gt yr⁻¹, respectively. The map in Figure 5.1 illustrates the global distribution of simulated water erosion rates. Highest water erosion is simulated in mountainous regions and regions with strong precipitation, especially in tropical climate zones. In Asia, those regions are widespread in the east, south-east and the Himalaya region. In Africa, similar areas with high water erosion values are spread around the continent and are most common at the west coast and in East Africa including broad areas in Guinea, Sierra Leone, Liberia, Ethiopia and Madagascar. In South America, highest water erosion is simulated in the south of Brazil and regions around the Andes mountain range and the Amazon river basin. The highest water erosion values on the American continent are simulated in tropical Central America and the Caribbean. In North America, highest water erosion occurs along the west coast and in the east. Water erosion in Europe is highest in Mediterranean areas and around the Alps.

Median annual water erosion values for the five largest maize and wheat producing countries demonstrate the strong impact of climate and topography on simulated water erosion. In Brazil, China and India, where a large proportion of cropland is in tropical areas, simulated water erosion is relatively high with annual median values of 10 t ha⁻¹, 6 t ha⁻¹, and 37 t ha⁻¹, respectively. In Russia and the United States annual median values are much lower with 1 t ha⁻¹, and 2 t ha⁻¹, respectively. Overall, Figure 5.1 illustrates the large variation in simulated water erosion between tropical climate regions and regions with a large proportion of flat and dry land.



Figure 5.1: (a) Soil loss due to water erosion in maize and wheat fields simulated with the baseline scenario. (b) The bars in the bottom plot illustrate the median, 25^{th} and 75^{th} percentile of simulated soil loss for major world regions. The classification of world regions is illustrated in Figure B.1. Due to the large gap between aggregated values, all values in the bottom plot have been log-transformed to improve the visual presentation.

5.3.1 Sources of model uncertainty related to management assumptions and method selection

Uncertainty in simulation results due to management scenarios and choice of water erosion equations is highest in the regions most at risk of water erosion (Figure 5.2). The annual median uncertainty range at each grid cell due to management is 30 t ha⁻¹. For 97% of grid cells, the lowest erosion rates are simulated with management scenarios including no-tillage and cover crops. For 86% of grid cells, maximum erosion rates are simulated under conventional tillage without cover crops. The annual median uncertainty range at each grid cell due to the choice of erosion equation is 23 t ha⁻¹. In 74% of grid

cells, the lowest erosion rates are simulated with the MUSS equation. The highest erosion values are simulated with the RUSLE equation (46%), followed by the USLE equation (25%).

In most locations, uncertainty due to field management exceeds the uncertainty caused by the choice of erosion equation. For 46% of grid cells, management scenarios cause the prevailing uncertainty, which is defined as the higher uncertainty range by at least 5 t ha⁻¹. The selected erosion equation causes higher uncertainty by at least 5 t ha⁻¹ in 14% of grid cells. The map in Figure 5.3 illustrates the global distribution of prevailing uncertainty sources.



Figure 5.2: Water erosion uncertainty due to (a) field management assumptions and (b) water erosion equations.

5.3.2 Main drivers of the global erosion model

The sensitivity study is designed to explain the large variability of simulated water erosion rates in different regions and to discuss the main differences between water erosion equations. Water erosion is highly sensitive to slope steepness (SLP) for all equations. The first-order sensitivity index of the



Figure 5.3: Prevailing uncertainty due to field management assumptions and water erosion equations, defined as the higher uncertainty range by at least 5 t ha^{-1} .

slope parameter indicates that 46–54% of the variance in the model output is attributable to the slope, without considering interactions between the input parameters (Table 5.1). Daily precipitation (PRCP) is the second most important parameter for calculating water erosion, with an individual contribution of around 9–20% to the variance of the output. The remaining parameters contribute together 4–13% to the output variance.

Table 5.1: First-order sensitivity indices (SI) ranking for the five most sensitive input parameters (PARM) for each water erosion equation including slope steepness (SLP), daily precipitation (PRCP), soil hydrologic group (HSG), land use number (LUN), soil silt content (SILT), soil sand content (SAND), curve number parameter (S301), maximum air temperature (TMX) and crop residues left after harvest (ORHI). The sensitivity indices of the remaining parameters are available in Carr et al. (2020).

	AOF		AOF MU		MUS	MUSL		MUSS		MUST		RUSLE2		RUSLE		E
	PARM	SI	PARM	SI	PARM	SI	PARM	SI	PARM	SI	PARM	SI	PARM	SI		
1	SLP	0.47	SLP	0.46	SLP	0.46	SLP	0.48	SLP	0.46	SLP	0.5	SLP	0.54		
2	PRCP	0.13	PRCP	0.1	PRCP	0.12	PRCP	0.09	PRCP	0.16	PRCP	0.2	PRCP	0.18		
3	HSG	0.03	HSG	0.04	HSG	0.05	HSG	0.04	HSG	0.02	SAND	0.05	SILT	0.02		
4	SILT	0.02	LUN	0.02	LUN	0.02	LUN	0.02	SAND	0.01	TMX	0.01	TMX	0.01		
5	LUN	0.01	SILT	0.02	S301	0.01	SILT	0.02	LUN	0.01	ORHI	0.01	ORHI	0		
	sum	0.69	sum	0.68	sum	0.71	sum	0.68	sum	0.71	sum	0.78	sum	0.76		

The first-order sensitivity indices do not include interactions between input parameters, which leads to the sum of all first-order sensitivity indices being lower than 1. The total-order sensitivity indices sum all first-order effects and interactions between parameters, which leads to overlaps in case of interactions and a sum greater than 1. The differences between the first-order and the total-order indices can be used as a measure to determine the impact of the interactions between a specific parameter with other parameters. The total-order sensitivity indices show that slope steepness, including interactions to other parameters, contributes 63–75% of the output variance from which 18–21% are due to interactive effects with other parameters (Table 5.2). The total-order sensitivity indices from precipitation range from 21–36%, from which 10–18% is due to interactions with other parameters.

Table 5.2: Total-order sensitivity indices (SI) ranking for the five most sensitive input parameters (PARM) for each water erosion equation including slope steepness (SLP), daily precipitation (PRCP), soil hydrologic group (HSG), land use number (LUN), soil silt content (SILT), soil sand content (SAND), maximum air temperature (TMX) and crop residues left after harvest (ORHI). The sensitivity indices of the remaining parameters are available in Carr et al. (2020).

	AOF		AOF		MUSL		MUSS		MUST		RUSLE2		RUSLE		USLE	
	PARM	SI	PARM	SI	PARM	SI										
1	SLP	0.68	SLP	0.68	SLP	0.63	SLP	0.68	SLP	0.66	SLP	0.69	SLP	0.75		
2	PRCP	0.28	PRCP	0.23	PRCP	0.22	PRCP	0.21	PRCP	0.32	PRCP	0.36	PRCP	0.36		
3	HSG	0.09	HSG	0.12	HSG	0.13	HSG	0.12	HSG	0.08	SAND	0.12	SILT	0.05		
4	SILT	0.07	LUN	0.07	LUN	0.07	LUN	0.07	LUN	0.05	TMX	0.02	TMX	0.02		
5	LUN	0.05	SILT	0.07	SILT	0.05	SILT	0.07	SAND	0.04	ORHI	0.01	SAND	0.01		
	sum	1.29	sum	1.3	sum	1.25	sum	1.27	sum	1.34	sum	1.27	sum	1.27		

The high sensitivity to slope and precipitation is similar for all equations, but the importance of the other parameters varies between equations. Equations estimating erosion energy by surface runoff and the RUSLE2 equation are very sensitive to the hydrological soil group (HSG), which determines the soils infiltration ability. This parameter is used in the calculation of the curve number, which defines the partition of precipitation into runoff and infiltration. Also, the land use number (LUN), which is ranked among the most sensitive input parameters, is used for the calculation of the curve number. The most sensitive parameters of the USLE and RUSLE equation, following slope inclination and daily precipitation, are soil texture classes (SAND and SILT) followed by daily temperature changes (TMX). Crop residues (ORHI) are relatively important for all equations but especially important for equations based on rainfall-energy. Other parameters relevant for field management, such as surface roughness and mixing efficiency of the topsoil, have little influence on water erosion.

The sensitivity of slope steepness has a strong positive correlation with the amount of annual precipitation at each location ($\rho = 0.69$, p<0.01). The increase in the sensitivity of slope steepness with increasing annual precipitation is demonstrated in Figure 5.4, which illustrates substantially lower sensitivity indices at dry locations compared to wet locations. In contrast, the sensitivity indices of daily precipitation are negatively correlated to annual precipitation with moderate strength ($\rho = 0.45$, p<0.05). Depending on the equation, strong positive or negative correlations between SIs and annual precipitation also exist for other parameters such as slope length, soil texture, soil organic carbon, channel length, channel slope and watershed area.

5.3.3 Evaluation of simulation results against field data

The most recent estimated global water erosion rates on cropland of 11 - 13 t ha⁻¹ derived from a comparable method (Borrelli et al., 2017; Doetterl et al., 2012; Van Oost et al., 2007) lie above the simulated median water erosion rates of 7 t ha⁻¹ and 5 t ha⁻¹ for maize and wheat fields, respectively. Similarly, the global water erosion estimates in maize and wheat fields are lower than the median value of 9 t ha⁻¹ from 606 water erosion measurements from cropland around the world.

To evaluate the agreement between simulated and observed data, median values between simulated and measured erosion rates are compared while grouped by precipitation and slope classes, which are



Figure 5.4: First-order and total-order Sensitivity indices for (a) slope steepness (%) and (b) precipitation (mm). The dashed vertical line illustrates median annual precipitation at all tested locations (1248 mm).

defined along the whole range of recorded slope inclinations and annual precipitation amounts of the field data (Figure 5.5a). Although slope and precipitation classes from the field are spread unevenly, they cover most climatic and topographic characteristics relevant to global agriculture. The comparison illustrates that the deviation between simulated data and field data is highest for locations with steep slopes and high annual precipitation. Where slopes are steeper than 8% and annual precipitation is higher than 1000 mm, the median of simulated water erosion exceeds the median of measured water erosion in most cases by at least 50 t ha⁻¹. With decreasing slope steepness and annual precipitation, the median deviation between simulated and measured data is decreasing. Where both slope steepness is below 8% and annual precipitation is below 1000 mm, the median deviation is lower than 5 t ha⁻¹ in most cases. A comparison of measured and simulated water erosion using other equations with the baseline scenario can be found in Figure B.2.

The boxplots in Figure 5.5b illustrate the range of water erosion values measured in the field and simulated with the baseline scenario. The high deviation between observed and simulated values for grouped locations with slopes steeper than 8% and annual precipitation higher than 1000 mm can also be observed between the range of simulated and measured water erosion values. Outside locations combining steep slopes and strong precipitation, median deviation between simulated and measured data is lower than the variability within the field data. The range of values at locations with lower precipitation and slope steepness demonstrates that simulated values are mostly below measured values in those environments.

The uncertainty in the choice of management scenarios and water erosion equations included in the baseline scenario leads to an uncertainty of the deviation between simulated and measured erosion values. This uncertainty is demonstrated in Figure 5.5b by additional three bars illustrating the range of simulated medians due to contrasting tillage management scenarios, cover crop scenarios and different water erosion equations. At locations with low to moderate slope steepness and annual precipitation, the measured water erosion values agree best with the simulation values generated under scenarios implying larger water erosion, such as high intensity tillage and low soil cover. On the other hand, at locations with steep slopes and intensive precipitation, the measured values are closer to the simulated values under scenarios with less intensive tillage and more soil cover. In addition, the varying sensitivities of each water erosion equation lead to a different magnitude of water erosion values in different environments. On low to moderate slopes, water erosion simulated with the MUSS equation is lowest, whereas RUSLE generates the highest values. On steep slopes, the RUSLE equation generates the lowest water erosion values, which agree best with the measured values. The options to increase and decrease simulated water erosion with different field management scenarios and water erosion equations creates both uncertainty in the model results, but also the possibility to closely match field data.

At locations combining steep slopes and intense precipitation, most management scenarios and equations generate water erosion values that are higher than the measured values. However, those environmental conditions cover only a small share of global cropland. Cultivation areas with slopes steeper than 8% and annual precipitation higher than 1000 mm represent only 7% of global maize and wheat cropland in the grid cells. The map in Figure 5.6 illustrates that the highest concentration of these areas is in East and Southeast Asia, followed by Central and South America, and sub-Saharan Africa.



Figure 5.5: Comparison of simulated erosion with measured erosion for different slope and precipitation classes. a) Median deviation (MD) in t ha⁻¹ between simulated erosion using the baseline scenario and measured erosion. Simulated and measured data is grouped into precipitation classes and slope classes used for the simulation setup. b) Distributions of measured erosion rates, erosion rates simulated with the baseline scenario and uncertainty ranges for management assumptions and erosion equations. The boxplots are defined by the median, the 25^{th} percentile and the 75^{th} percentile of simulated and measured erosion rates. Whiskers illustrate the 10^{th} and 90^{th} percentiles. The three bars next to the boxplots illustrate minimum and maximum median erosion rates calculated with all tillage and cover crop scenarios and with all water erosion equations. The values have been log-transformed for better visualisation.



Figure 5.6: Distribution of low to high slope steepness (SLP) and annual precipitation (PRCP) in maize and wheat fields. Dark areas illustrate grids where slopes are steeper than 8% and annual precipitation is above 1000 mm. Correspondingly, blue, red, and grey pixels are below one or both thresholds.

5.4 Discussion

5.4.1 Varying robustness of simulated water erosion in different global regions

Global water erosion estimates generated with EPIC-IIASA and the baseline scenario overlap with observed water erosion values under most of the climatic and topographic environments where maize and wheat are grown. However, global maize and wheat land include locations where environmental characteristics differ significantly from the Midwestern United States, where the data was collected to develop the water erosion equations embedded in EPIC. The USLE model and its modification were developed with data for slopes of up to 20%, which makes model application for steeper slopes uncertain (McCool et al., 1989; Meyer, 1984). Furthermore, the relations between kinetic energy and rainfall energy in the American Great Plains differ from other regions in the world (Roose, 1996). Similarly, the runoff curve number method, which is the key methodology for the calculation of surface runoff, is based on an empirical analysis in watersheds located in the United States and might be less reliable in different regions of the world (Rallison, 1980). Due to the high sensitivity of slope steepness and daily precipitation for the calculation of water erosion, the reliability of the tested equations decreases in regions where typical slope and precipitation patterns differ from the Midwestern USA. Although some studies have successfully used USLE and its modification under a different environmental context (e.g., Alewell et al., 2019; Almas and Jamal, 2009; Fischer et al., 2018; Sadeghi and Mizuyama, 2007), many studies have concluded that the accuracy of these models may be reduced outside the environments they were created without calibration and model adaptation (Cohen et al., 2005; Labrière et al., 2015).

The skewed distribution of simulated water erosion values influenced by extreme soil loss rates in few fields highly sensitive to water erosion results in a large difference between the global median value of 6 t ha⁻¹ a⁻¹ and the global average value of 19 t ha⁻¹ a⁻¹ (Figure B.3). Due to the strong influence of outliers on average values, median values better represent global values and the differences in regional

water erosion rates in wheat and maize fields. The high sensitivity of the simulation results to slope inclinations and precipitation suggests that a significant share of the estimated soil removal of 7.2 Gt a^{-1} originates from small wheat and maize cultivation areas on steep slopes with strong annual precipitation.

5.4.2 Sources of uncertainties in global water erosion estimates

Uncertain land use in mountainous regions

Changing climatic conditions with increasing elevation and the variable soils in mountainous regions can favour crop cultivation in higher elevations over lower elevations (Romeo et al., 2015). However, upland farming without soil conservation measures can lead to exhaustive soil erosion and can become a critical problem for agriculture (Montgomery, 2007). Large areas of land have been abandoned due to high erosion rates as soils were no longer able to support crops (Figure 5.7) (Romeo et al., 2015). As mountain agriculture is determined by various environmental and socio-economic factors, the cultivation of steep slopes can be very variable between regions. Regional erosion assessments in mountainous cropland suggested that areas with extreme water erosion rates are mainly limited to marginal steep land cultivated by smallholders (Haile and Fetene, 2012; Long et al., 2006a; Nyssen et al., 2019). In some mountainous regions, efforts to remove marginal farmlands from agricultural production, and programs to improve land management on steep slopes have reduced high water erosion rates (Deng et al., 2012; Nyssen et al., 2015). On the contrary, recent pressure through increasing population and crop production demands has resulted in re-cultivation of hillslopes and a reduction of fallow periods, which limits the recovery of eroded soil (Turkelboom et al., 2008; Valentin et al., 2008).

To analyse the sustainability of simulated maize and wheat cultivation systems that are exposed to high erosion rates, the simulated annual eroded soil depth is compared with a global dataset on modelled sediment deposition thickness (Pelletier et al., 2016) in Figure 5.8. The comparison shows that at 4% of grid cells permanent maize and wheat cultivation would not be sustainable as the entire soil profile would be eroded at the end of the simulation period. Most of the unsustainable agriculture is simulated on steep slopes. Although conservation techniques and cover crops are considered in the field management scenarios, the highly complex cultivation methods, including intercropping and fallow periods, common on hills that are normally farmed by smallholders, could not be modelled (Turkelboom et al., 2008). Moreover, the modelling approach assumes that the slope class representing the largest area in each grid cell most likely represents the largest share of arable land. This builds on the idea that a spatially extensive and diverse landscape can be represented by a single "representative field" characterised by the prevailing topography and soil conditions found in the landscape. On hilly terrain this setup simulates maize and wheat cultivation on steep slopes and thus mainly represents unsustainable agriculture. Although unsustainable maize and wheat cultivation can be observed in several mountain regions, cropland is very heterogeneously distributed in mountains and thus erosion rates from one representative field are highly uncertain.

The uncertainty in cropland distribution could partly be reduced by developing a higher resolution global gridded data infrastructure, which is currently not available for EPIC-IIASA. However, due to



Figure 5.7: (a) Sugar cane cultivation on steep slopes in southern China (Nanning, Guangxi Zhuang Autonomous Region). The steepest slopes are already abandoned and reforested by eucalyptus trees. (b) Maize cultivation on strongly eroded slopes (30%–60%) in south-west Uganda (Kigwa, Kabale District). (c) Abandoned fields and maize cultivation on a steep slope (30%–60%) in south-west Uganda (Kigwa, Kabale District). (d) Degraded and abandoned maize fields on steep slopes (20%–60%) in northern El Salvador (San Ignacio, Chalatenango Department) (Carr et al., 2020).



Figure 5.8: Simulated years left until the whole soil profile is eroded under permanent maize and wheat cultivation. Calculated as a ratio of the sedimentary deposit thickness (m) (Pelletier et al., 2016) and the eroded soil depth per year (water erosion (t $ha^{-1} a^{-1}$) x bulk density (g m^{-3})).

the large uncertainty in global land cover maps (Fritz et al., 2015; Lesiv et al., 2019), an explicit spatial link between cropland distribution and the corresponding slope category cannot be established without on-site observations. The effect of this uncertainty on erosion estimates is analysed for Italy, where large
maize and wheat growing areas are spread across both flat terrain in the north and mountainous areas in the south. In an ideal scenario where cropland is limited to flattest land available per grid cell, median simulated water erosion in Italy would be reduced to tolerable levels below 1 t ha⁻¹. However, in a scenario, where the most common slopes per grid cell are cultivated, median simulated water erosion increases to 14 t ha⁻¹ due to high water erosion simulated in Italy's mountainous regions (Figure B.4). This suggests a high uncertainty in global water erosion estimates due to uncertain spatial links between maize and wheat cultivation areas and different slope categories.

Uncertain field management

Simulated water erosion values are highly variable depending on the field management scenario. Simulating cover crop and no-tillage worldwide results in the lowest global soil removal of 2 Gt a^{-1} with median water erosion rates of 1 t $ha^{-1} a^{-1}$ and simulating no cover crops and conventional tillage worldwide results in the highest global soil removal of 13 Gt a^{-1} with median water erosion rates of 17 t $ha^{-1} a^{-1}$. These variations cause further uncertainties in the simulation results.

Indeed, a proper reconstruction of a business-as-usual field management is important to further narrow down the uncertainty in global crop modelling (Folberth et al., 2019). In this study, the predominant field management was assigned based on a series of environmental and country-specific indicators, similar to Porwollik et al. (2019). For example, conservation agriculture was only considered in countries where this management strategy is likely according to AQUASTAT (FAO, 2016). In addition, by assuming cover crops between the wheat and maize seasons, more complex cultivation systems were simulated in the tropics, in which long and year-round vegetation periods and frequent multi- cropping practices hardly leave the soil uncovered. Bare fallow land was therefore not simulated in the tropics, as high water erosion values would have been falsely simulated at locations where heavy precipitation fell on bare ground. In addition, conservation practices such as contouring and terracing are crucial to reduce the simulation of high water erosion values on steep slopes. These practices were simulated for specific slope classes under the assumption that farmers around the world uniformly use conservation practices when cultivating on steep slopes. The most relevant parameters used for tillage scenarios are related to crop residues left in the field. In addition, equations directly connected to surface runoff are strongly influenced by the land use number used to determine the impact of cover type and treatment on soil permeability. While both crop residues and green fallow decrease water erosion significantly, especially in the tropics, their use varies widely between regions and even farms, based on a complex web of factors such as institutional factors, farm sizes, risk attitudes, interest rates, access to markets, farming systems, resource endowments, and farm management skills (Pannell et al., 2014). Also, soil conservation measures such as terraces or contour farming significantly influence water erosion but are very heterogeneously used between regions, farming systems and farmers. The baseline scenario is a very crude representation of the complex patterns of field management around the world, but attempts to depict these highly influential practices with the limited data available.

Variable estimates from different water erosion equations

The water erosion equation chosen for the baseline scenario generates the lowest global soil removal estimate. Different water erosion equations embedded in EPIC estimate a higher global soil removal of up to 11 Gt a^{-1} as well as higher median water erosion rates up to 19 t $ha^{-1} a^{-1}$. The MUSS water erosion equation chosen for the baseline scenario generates water erosion rates closest to the field data. The focus of equations on either rainfall energy or runoff energy is relevant for the different simulation results under specific environmental conditions. Equations based on rainfall-energy such as RUSLE and USLE simulate higher water erosion values than the other equations at most locations. However, on steep slopes they generate the lowest water erosion values as runoff becomes a greater source of energy than rain with increasing slope steepness (Roose, 1996). Also, the varying sensitivities of other parameters to the equations such as soil properties and management parameters lead to a varying agreement between simulated data and field data depending on the equation selection. Detailed field data would facilitate the choice of an appropriate equation to simulate water erosion worldwide or for a specific region.

5.4.3 The difficulty of evaluating large-scale erosion estimates with field data

The selection of field data for evaluating simulated water erosion was limited by the low availability of suitable water erosion observations covering the entire globe. The lack of reliable data on water erosion rates is a severe obstacle for understanding erosion, developing and validating models and implementing soil conservation (Boardman, 2006; Nearing et al., 2000; Poesen et al., 2003; Trimble and Crosson, 2000). The main reasons for the low availability of suitable data to evaluate simulated water erosion rates are twofold: i) erosion monitoring is expensive, time consuming and labour demanding; and, ii) primary data and metadata of measurement sites accompanying final results are often not available and many older measurements are poorly accessible as they are not available online (Benaud et al., 2020).

A variety of factors influencing water erosion such as climate, field topography, soil properties and field management need to be considered when modelling water erosion but are often not reported in available field measurements (García-Ruiz et al., 2015). This hampers a direct comparison between simulated and observed water erosion values. The varying match between measured and simulated water erosion on the basis of different tillage and cover crop scenarios was demonstrated in this chapter. Metadata on field management often only provides the crop cultivated and therefore the conditions under which erosion was measured in the field are not known sufficiently to evaluate erosion values simulated under different field management scenarios. Similarly, information on field topography and soil properties is often not provided with recorded field measurements and thus their use is limited in an evaluation of water erosion estimates simulated in different global environments. Moreover, most data are concentrated in the United States, West Europe and the West Mediterranean (García-Ruiz et al., 2015). In summary, there is a lack of field data representing all needed regions, situations and scenarios (Alewell et al., 2019).

The appropriate selection of field data to evaluate model outputs needs to be considered as well. At different spatial scales different erosion processes are dominant and consequently different erosion measurement methods are suitable (Boix-Fayos et al., 2006; Stroosnijder, 2005). Most authors use very heterogeneous data sets to evaluate their models, involving data generated by different methods at variable time and spatial scales and variable quality. For example, Doetterl et al. (2012) used plot data, suspended sediments from rivers, and data from RUSLE modelling. Borrelli et al. (2017) used soil erosion rates (measurement methods are not specified), remote sensing, vegetation index (NDVI) and results of RUSLE modelling. In his review on erosion rates under different land use, Montgomery (2007) used field data derived from erosion plots, field-scale measurements, catchment-scale measurements using hydrological methods, ¹³⁷Cs-method, soil profile truncation and elevated cemetery plots.

While all erosion measurement methods can be criticised, only data obtained through field measurements from runoff plots, using the ¹³⁷Cs-method, and volumetric surveys were used in this study, as these methods are best suited on the plot, slope and field scales. Geodetic methods such as erosion pins and laser scanner are also used at plot to field scales, but their accuracy is much lower than the accuracy of plot measurements and ¹³⁷Cs-method. Furthermore, erosion pins are mainly suitable for areas with extreme erosion rates (Hsieh et al., 2009; Hudson, 1993), and laser scanners have difficulties to recognize vegetation (Hsieh et al., 2009). Other commonly used methods such as hydrological methods (measurements of discharge and suspended sediment load) and bathymetric methods are more suitable for larger scales and involve a significant portion of channel erosion, which is not related with agricultural land (García-Ruiz et al., 2015). Plot experiments using rainfall simulators have also been neglected as they are usually performed on small plots with artificially generated rainfalls, which mostly have very low energies and therefore generate low erosion rates (Boix-Fayos et al., 2006; García-Ruiz et al., 2015).

The ¹³⁷Cs method was criticised by Parsons and Foster (2011), who questioned assumptions about the ¹³⁷Cs behaviour in the environment (variability of the ¹³⁷Cs input by wet fallout, its microspatial variability at reference sites, its possible mobility in certain soils, the ¹³⁷Cs uptake by plants and other aspects of ¹³⁷Cs behaviour in soil). To confront the criticism against the ¹³⁷Cs method, Mabit et al. (2013) discussed all objections raised by Parsons and Foster (2011) and confirmed its accuracy by listing several studies, in which ¹³⁷Cs based erosion rates are compared with erosion rates derived from direct measurements. The ¹³⁷Cs method is based on a set of presumptions which should be met to produce useful results and thus careful interpretation of the obtained results is needed (Fulajtar et al., 2017; Mabit et al., 2014a; Zapata, 2002).

Similarly, erosion rates obtained by volumetric measurements require careful interpretation as they are exposed to various potential sources of errors and do not account for interill erosion. Although the latter can be neglected under certain circumstances, studies from Europe and semiarid areas of the USA have reported that interill erosion contributed significantly to the amount of soil eroded in fields (Boardman and Evans, 2020; Parsons, 2019). Further, measuring the lengths and cross-sections of rills during field surveys or on terrestrial and aerial photos can be very subjective (Panagos et al., 2016). Different approaches used to detect and measure rills in fields can cause variability in calculated erosion volumes up to a factor of two (Boardman and Evans, 2020; Casali et al., 2006; Watson and Evans, 1991). In order to obtain soil erosion rates in weight units, soil volumes need to be converted using the soil bulk

density, which is often based on estimates (Evans and Brazier, 2005).

The shortcomings of erosion plot measurements were discussed by several authors (Auerswald et al., 2009; Brazier, 2004; Evans, 2002; Loughran et al., 1988). Erosion plots have various sizes and shapes (few meters to few hundreds of meters) and various approaches of sediment recording are used (total collection, multislot divisors, tipping buckets, Coshocton wheels), which all involve significant uncertainties. Although some long-term plot experiments exist, many plot measurements fail to cover the whole year erosion cycle (Auerswald et al., 2009). Often, they have to be removed during land management operations such as seeding, ploughing, or they are too expensive and labour demanding.

Despite all the shortcomings of available soil erosion data, most data provide valuable information (Benaud et al., 2020). The evaluation against field measurements in this study provided a first indication of the robustness of results under specific topographic and climatic conditions. In most environments relevant for maize and wheat cultivation the deviation between simulated and measured water erosion values is lower than the variability within the field data. The mentioned uncertainties cannot be narrowed down further on the basis of the reported data. Although the metadata accompanying the field measurements includes information on slope steepness and annual precipitation (or geographic coordinates allowing for overlay with climatic data), information on soil types or texture classes, crop type and tillage system implemented over time are provided only for few points. Also, the various methods used to measure erosion rates, their complex implementation and the bias of field studies towards locations sensitive to erosion lead to an uncertain representation of large-scale erosion rates based on field measurements. To facilitate in-depth evaluation of erosion models across different scales, it is crucial to provide detailed information on site characteristics and to harmonise approaches to measure erosion in the field. Moreover, the accessibility of field data should be improved as raw data is often not published or needs to be collected from numerous publications, grey literature and conference proceedings to obtain the large amount of data necessary for regional or global erosion studies. Recent efforts to compile erosion measurements and metadata from existing studies on a single platform (Benaud et al., 2020) will likely increase the availability of field data, which will greatly benefit future modelling studies and understanding of soil erosion at all levels.

5.5 Conclusion

The simulation of water erosion with GGCMs is largely influenced by the resolution of global datasets providing topographic, soil, climate, land use and field management data, which is currently not available at the field scale. Yet, considering water erosion in global crop yield projections can provide useful outputs to inform assessments of the potential impacts of erosion on global food production and to identify soil erosion hotspots on cropland for management and policy interventions. To improve the quality of the estimates and to further develop these models, it is crucial to identify, communicate and address the existing uncertainties. Increasing the resolution of global soil, topographic and precipitation data is central for improving global water erosion estimates. In addition, the model results provide

an insight into the importance of considering field management. The numerous options to simulate the cultivation of fields result in a large range of possible water erosion values, which can only partly be narrowed down at a global scale. Further improvement of global water erosion estimates requires detailed and harmonised field measurements across all environmental conditions to validate and calibrate simulation outputs. On the basis of existing field data, it was possible to identify specific environments for which the modelled erosion rates show poor robustness. These are mainly found in the tropics and mountainous regions due to the high sensitivity of simulated water erosion to slope steepness and precipitation strength, and the complexity of mountain agriculture. However, these areas represent only a small fraction of global cropland for maize and wheat. The overlap of simulated and measured water erosion values in most environments used to produce maize and wheat underlines the robustness of EPIC-IIASA to simulate the differences in water erosion rates of major global crop production regions.

Chapter 6

The impact of water erosion on global maize and wheat yields

6.1 Introduction

The negative effects of land degradation on social and economic wellbeing has been widely recognised. Yet its present and future impacts are not adequately quantified globally in physical and economic terms to inform major environmental and agricultural policies (Montanarella, 2007; Nkonya et al., 2011).

Soil loss due to water erosion has been estimated at many sites worldwide and modelled globally (Borrelli et al., 2017; Doetterl et al., 2012; García-Ruiz et al., 2015; Montgomery, 2007). However, from a food security standpoint, it is more relevant to quantify the impact of water erosion on crop productivity. There are substantial variations in the estimates of productivity losses from the few studies in the literature (Bakker et al., 2004, 2007; Den Biggelaar et al., 2004a; van den Born et al., 2000; De la Rosa et al., 2000; Lal, 1995; Larney et al., 2009; Oyedele and Aina, 1998). This variability is not surprising as erosion-productivity relationships are difficult to generalise due to the location-specific nature of soil erosion determined by soil properties, climate and management (Den Biggelaar et al., 2004a). Moreover, the choice of method to measure water erosion impacts on crops is one of the most important factors explaining variations between studies (Bakker et al., 2004). Hence, different methodological approaches in field studies can mask the impact of regional differences on water erosion impacts on crops.

Previous global erosion impact assessments (Pimentel et al., 1995; Sartori et al., 2019) relied on simple linear assumptions about the impact of water erosion on crop yields, and neglected differences between crops and regional characteristics (Chapter 2.4). Crop models can facilitate the extrapolation of experimental and small-scale data across a range of environments and management strategies (Nelson et al., 1996). Moreover, models are essential to determine long-term effects of degradation processes, which are challenging to observe in short-term field experiments (Enters, 1998). However, due to the lack of evaluated global water erosion estimates, GGCM studies have so far neglected soil erosion and its impact on crop yields and production (Chapter 3.6).

After confirming in Chapter 5 that EPIC-IIASA can represent the differences in water erosion rates between major global cropping regions, the simulated water erosion rates are used to quantify their impact on global maize and wheat yields and production. In addition, regions where crops are most exposed to the risks of water erosion are identified based on the environmental conditions and the use of fertilisers. Finally, the uncertainties of the simulated water erosion impacts on crops are addressed.

6.2 Methodology

6.2.1 Summary of methodology

To simulate the impact of water erosion on maize and wheat yields, the same modelling approach as in the previous chapter is used. As such, the model outputs are based on the baseline field management scenario presented in Chapter 4.3.4. The range between minimum and maximum water erosion impacts on crops simulated with all field management scenarios is used to bracket the field management uncertainties for each grid cell and country.

6.2.2 Aggregation of model outputs

Water erosion impact

Water erosion impacts on maize and wheat yields are calculated for the end of the simulation period for the years 2001–2010. Crop yields generated with all field management scenarios selected under the baseline scenario assumptions are averaged over the respective years and weighted by the irrigated and rainfed cultivation area (Portmann et al., 2010) of the respective crop per grid cell (Equation 6.1). The difference between average maize and wheat yields, simulated with and without the impact of water erosion, are used to filter grid cells where water erosion reduces crop yields (i.e. the area where crop yields are vulnerable to water erosion). Subsequently, the relative reduction of maize and wheat yield due to water erosion is calculated on grid cell level (Equation 6.2).

$$Yw_{cpg} = Yav(r)_{cpg} * Af(r)_{cg} + Yav(i)_{cpg} * Af(i)_{cg}$$

$$(6.1)$$

$$dYrel_{cg} = \frac{Yw(e0)_{cg} - Yw(e1)_{cg}}{Yw(e0)_{cg}}; if Yw(e0)_{cg} > Yw(e1)_{cg}$$
(6.2)

 Yw_{cpg} is area-weighted mean crop fresh matter yield (tha^{-1}) for crop c, P-factor value p and grid cell g; Yav is yield averaged across the tillage and cover crop scenarios selected in each grid following the baseline scenario assumptions and for the selected decade simulated under rainfed (r) and irrigated (i) conditions; Af(r) is the rainfed area fraction; and Af(i) is the irrigated area fraction. dYrel is the relative loss of the yield of crop c, at grid cell g; Yw is weighted average yield simulated with a P-factor value of 0 (e0) and a P-factor value greater than 0 (e1).

To calculate the loss of crop production in each country, first, the absolute reduction of crop yields

is estimated as the difference in the mean yield for the selected decade simulated without and with water erosion (*e*0 and *e*1, respectively) (Equation 6.3). Subsequently, this yield reduction is multiplied by the total area of rainfed and irrigated cropland of each grid cell in the country (Equation 6.4).

$$dYabs_{cwg} = Yav(e0)_{cwg} - Yav(e1)_{cwg}; if Yw(e0)_{cpg} > Yw(e1)_{cpg}$$

$$(6.3)$$

$$dP_{lc} = \sum_{g=1}^{n} dYabs(i)_{cg} * A(i)_{cg} + dYabs(r)_{cg} * A(r)_{cg}$$
(6.4)

 $dYabs_{cwg}$ is the absolute yield loss for crop *c*, irrigation scenario *w* and grid cell *g*; *Yav* is yield averaged across the tillage and cover crop scenarios selected in each grid cell following the baseline scenario assumptions and for the selected decades with *P*=0 (*e*0) and *P*>0 (*e*1); dP_{lc} is the loss of production (in tonnes) of crop *c* in country *l*; *n* is the number of grid cells in country *l*; dYabs(i) is the absolute decline in irrigated yields and dYabs(r) is the absolute decline in rainfed yields; A(r) is the rainfed area (in *ha*); and A(i) is the irrigated area (in *ha*).

The national market prices of crops taken from the FAOSTAT producer price (average 2013–2018, or the last five annual records available) is used to calculate the economic maize and wheat production losses (in USD) due to water erosion per country and globally (FAO, 2020). Two-tailed T-tests are used to filter countries with significant differences between average yields simulated with and without water erosion.

Evaluation of simulated crop yields

Simulated maize and wheat yields, which are calculated in t ha^{-1} dry matter in EPIC, are converted to fresh matter assuming a net water content of 12% following Wirsenius (2000), so that they can be compared with yields reported by FAOSTAT (FAO, 2020). Fresh matter crop yields are aggregated for each country using the same approach as for grid cell-level aggregation in Equation 6.1. Irrigated and rainfed crop yields (generated with all P-factor values, tillage and cover crop scenarios selected for the baseline scenario and the years 2001 and 2010) are averaged for each country and weighted by the cultivated area of the respective irrigated or rainfed crop per country (Portmann et al., 2010) (Equation 6.5). In addition, average maize and wheat yields per grid cell are used to summarise the total maize and wheat production for each country (Equation 6.6).

$$Yw_{cl} = Yav(r)_{cl} * Af(r)_{cl} + Yav(i)_{cl} * Af(i)_{cl}$$
(6.5)

$$Pcl = \sum_{g=1}^{n} Yav(r)_{cg} * A(r)_{cg} + Yav(i)_{cg} * A(i)_{cg}$$
(6.6)

 Yw_{cl} is weighted yield for crop c in country l; Yav is yield averaged for the years 2001–2010, with all P-factor values and all tillage and cover crop scenarios selected under the baseline scenario assumptions simulated under rainfed (r) and irrigated (i) conditions; Af(r) is the rainfed area fraction

and Af(i) is the irrigated area fraction; P_{cl} is the total production (in tonnes) of crop *c* in country *l*; *g* is any grid cell in country *l*; *n* is the number of grid cells in country *l*; A(r) is the rainfed area and A(i) is the irrigated area in hectares.

6.3 Results

6.3.1 The impact of water erosion on global maize and wheat yields

In the last decade of the 30-year simulation period, the average annual maize and wheat yields are reduced due to water erosion at 58% and 62% of grids cells, respectively, by a global median of 3% for each crop. The affected grid cells represent 51% and 46% of global maize and wheat cultivation areas, respectively. Median annual soil loss at grid cells where crop yields are reduced is 11 t ha^{-1} and 6 t ha^{-1} on maize and wheat fields, respectively. The simulated relative reduction of average annual maize and wheat yields per grid cell at the end of the simulation period is illustrated in Figure 6.1. Most grid cells where high yield reduction is simulated represent fields with low fertiliser input on steep slopes exposed to intensive precipitation.

The distribution of annual average crop yield losses for the 40 most vulnerable maize- and wheatproducing countries is plotted in Figure 6.2. Countries in which the median annual reduction of maize yields due to water erosion is higher than 5% by the end of the simulation period are most abundant in sub-Saharan Africa and across Asia. There are similarly high median maize yield losses for countries in Central America and the Caribbean, but only Chile and Uruguay are badly affected in South America, and only Albania, Croatia and Greece in Europe. Median wheat yield losses per country are generally lower than for maize. Countries with median wheat losses higher than 5% are mostly in Asia and Europe. In Africa, annual median wheat yield losses higher than 5% are simulated in Ethiopia, Uganda and Tanzania, and in South America in Uruguay, Bolivia and Chile. These crop yield losses are modelled using the prevailing environment and management conditions in each country. Actual crop yield losses could only be determined based on an explicit spatial link between the extent of crop cultivation areas and areas vulnerable to water erosion, which would only be possible with on-site observations.

The distribution of the magnitude of crop yield losses and the share of grid cells affected by water erosion needs to be considered to assess each countries vulnerability to water erosion. In some large countries, the majority of cropland is exposed to low water erosion despite extensive vulnerable areas within the country. For example, large areas in the United States, Brazil, India and China are affected by water erosion. However, as these areas are only a small part of the entire cropland area, overall median crop losses are low. On the other hand, in some countries a small number of grid cells with high water erosion cause high median crop productivity losses. Afghanistan, Pakistan and Iran are ranked among the most vulnerable countries even though fewer than half of the grid cells are affected by water erosion under all scenarios.

In several countries, field management scenarios have a significant impact on the area affected by water erosion and on the magnitude of crop yield losses, as demonstrated by the uncertainty ranges in

Figure 6.2. In most countries, the median maize and wheat yield losses are lowest with no tillage and cover crops and highest with conventional or reduced tillage and bare soil fallow. On a global scale, annual maize and wheat yield losses simulated under all field management scenarios range from 2-5% and 3-4%, respectively.



Figure 6.1: Maize (a) and wheat (b) yield loss due to water erosion (% yr^{-1}) simulated with the baseline scenario and averaged for the years 2001 – 2010. Each grid cell is represented by one representative field capturing the most common site characteristics. Cropland areas are not considered in grid cell size.



Figure 6.2: Maize (a) and wheat (b) yield losses due to water erosion (% yr-1) for the 40 most vulnerable countries estimated with the baseline scenario. Countries contributing less than 0.01% to global maize and wheat production are excluded. The countries are ranked by median crop yield losses. Boxes include values from the 25th to the 7^{5th} percentiles and whiskers bracket values between the 10th and the 90th percentiles. The points illustrate minimum and maximum median crop yield losses generated from all field management scenarios. Medians and percentiles are converted to logarithmic scale. Grey barplots on the right illustrate the share of grid cells affected by water erosion impacts in each country, and errorbars indicate the variability of affected grid cells due to all management scenarios. The distributions of all relevant maize and wheat producing countries are provided in Figure C.2 and Figure C.3.

6.3.2 Fertiliser use and environmental drivers affect the impacts of water erosion

The simulated impact of water erosion on crop yields is strongly influenced by fertiliser input and environmental drivers in each country such as slope inclination and precipitation amount. Figure 6.3a shows that median maize and wheat yield losses per country tend to be higher in countries with higher levels of water erosion. Losses are relatively lower in countries with high rates of fertiliser application, which replace nutrients lost through soil runoff (Figure C.4). Global median rates of organic nitrogen runoff from maize and wheat fields are 7 kg ha⁻¹ yr⁻¹ and 5 kg ha⁻¹ yr⁻¹, and the global median rates of soil organic carbon runoff from maize and wheat fields are 107 kg ha⁻¹ yr⁻¹ and 72 kg ha⁻¹ yr⁻¹ during the whole simulation period (global maps of nitrogen and carbon runoff are provided in the supplementary information in Figures C.5 and Figure C.6).

Slope steepness and precipitation strength are the most important environmental drivers influencing the impact of water erosion on crop yields. Figure 6.3b and 6.3c show how yield losses increase as a

function of slope classes and rainfall erosivity classes. The distribution of maize and wheat cropland in the grid cells per slope and rainfall erosivity classes is illustrated by the grey bars in the same plots. Around 73% of maize and wheat cropland is on slopes whose steepness does not exceed 5%. On those slopes, median global maize and wheat yield losses range from 0% to 1%. On steeper slopes, median yield losses range from 3% to 9%. Similarly, 69% of maize and wheat land is exposed to rainfall erosivity below 3000 MJ mm ha⁻¹ h⁻¹ yr⁻¹, which is the average rainfall erosivity on global cropland. For those areas, median crop yield losses range from 1% to 2%. Median crop yield losses on fields exposed to higher rainfall erosivity range from 2% to 4%.



Figure 6.3: (a) Modelled median maize and wheat yield loss plotted against median soil loss through water erosion for each country. The linear relationship between national soil loss and crop yield loss is illustrated by the dashed regression line. Colours indicate the rate of fertiliser application per country. (b,c) Maize and wheat yield losses, respectively, per grid cells classified by slope steepness and rainfall erosivity. Grey bars illustrate the share of cropland in grid cells summarised for the different slope and rainfall erosivity classes.

The highest yield losses tend to occur in regions with low fertiliser input and high rates of water erosion. Figure 6.4 identifies agricultural regions susceptible to water erosion as indicated by overlapping areas of slope steepness (IIASA/FAO, 2012) and rainfall erosivity (Panagos et al., 2017), and shows the average fertiliser application rates for maize- and wheat-producing countries (Mueller et al., 2012). Dark areas highlight most vulnerable locations characterised by high abundance of steep slopes in regions of high rainfall erosivity. These are most common in South, East and Southeast Asia, sub-Saharan Africa, and Latin America. The cultivation on steep slopes is a common factor of vulnerability outside the tropics as well, but rainfall erosivity decreases there, reducing the energy of rainfall to erode soil. fertiliser application per country varies significantly. In most African countries and in several countries in Asia and Latin America, the fertiliser use is substantially lower than in the rest of the world.



Figure 6.4: water erosion vulnerability on global cropland indicated through the most important environmental drivers, rainfall erosivity (MJ mm $ha^{-1} h^{-1} yr^{-1}$) and slope steepness (%), and the average sum of Nitrogen, Phosphorous and Potassium fertiliser application rates (kg ha^{-1}) per country represented by the red bars. To improve the overview of the map, fertiliser application from countries contributing less than 0.1% to global maize and wheat production have been excluded, and fertiliser application from all relevant EU27 countries has been averaged.

6.3.3 The impact of water erosion on total maize and wheat production

The total annual production losses per country are based on the average absolute yield losses of maize and wheat, which were extrapolated over the entire irrigated and rainfed cultivation area of each crop in a grid cell (Figure 6.5). Water erosion reduces the global production of maize and wheat by 9 million tonnes and 6 million tonnes annually. This accounts for less than 1% of the global average maize and wheat production of 1,091 million tonnes and 739 million tonnes, respectively, from 2013–2018 reported by FAOSTAT (FAO, 2020). Market values of the national maize and wheat production losses, derived by multiplying production losses with the average market prices (USD t⁻¹) in each country, add up to an annual global loss of approximately USD 2bn in maize production, and USD 1.3bn in wheat production. Highest production losses in absolute terms are in countries with the largest maize and wheat cultivation areas rather than in the most vulnerable countries. Tables 6.1 and 6.2 list the 20 countries with the highest annual reduction in maize and wheat production due to water erosion. These countries account for 84% and 77% of the global maize and wheat production.

The largest maize production declines are estimated for the most important producers such as Mexico, Brazil, United States, India, China and Indonesia. Nevertheless, losses in the United States and China are only 0.2% of their national production, but reach 5% of Mexico's production. Few countries with the highest absolute losses have low shares of global production (e.g. Guatemala; Nicaragua; Nepal; Myanmar).

Similarly, the modelled loss of wheat production due to water erosion in absolute terms is highest for India and China as they produce nearly a third of global wheat, but is less than 1% of their total production. High production losses in absolute terms for small producers are rarer than for maize. Countries with lowest production losses in absolute terms are most abundant in Africa, Southeast Asia and Latin America.



Figure 6.5: The impact of water erosion on national maize (a) and wheat (b) production based on the sum of estimated production losses in all grid cells in each country. NA marks countries without maize or wheat production area. Estimates of production losses in each grid cell assume uniform site characteristics for the entire cropland in each grid cell.

country	prod. (million t)	prod. loss (million t)	prod. loss (%)	prod. loss (million \$)
Mexico	25.6	1.3	5.0	264.8
Brazil	81.6	0.8	1.0	157.7
USA	376.7	0.7	0.2	104.9
India	25.6	0.6	2.5	92.0
China	246.7	0.5	0.2	199.8
Indonesia	23.3	0.5	2.1	151.8
Philippines	7.6	0.4	5.2	111.3
Nepal	2.2	0.3	12.5	74.2
Guatemala	1.9	0.2	12.8	37.2
Russia	12.7	0.2	1.5	24.6
Argentina	38.6	0.2	0.5	31.1
Tanzania	6.0	0.2	2.7	29.8
Nigeria	10.2	0.1	1.3	41.2
Myanmar	1.8	0.1	6.5	27.1
Nicaragua	0.4	0.1	27.8	31.9
Romania	12.7	0.1	0.9	20.6
Ukraine	28.6	0.1	0.4	14.8
France	14.4	0.1	0.7	17.9
Ethiopia	7.5	0.1	1.3	20.8
Viet Nam	5.2	0.1	1.7	26.4
World	1091.1	8.9	0.8	1960.7

Table 6.1: National maize production (FAO, 2020) and absolute and relative maize production losses for the countries with the highest absolute losses.

country	prod. (million t)	prod. loss (million t)	prod. loss (%)	prod. loss (million \$)
India	94.4	0.7	0.7	137.4
China	130.0	0.6	0.5	213.7
Turkey	21.0	0.5	2.5	139.4
USA	55.1	0.5	0.8	89.4
Russia	67.5	0.4	0.6	60.2
France	37.4	0.3	0.8	56.9
Argentina	13.2	0.2	1.8	56.5
Iran	12.4	0.2	1.6	77.4
United Kingdom	14.6	0.1	1.0	30.1
Italy	7.3	0.1	1.9	32.5
Germany	24.8	0.1	0.5	22.9
Ukraine	25.0	0.1	0.5	17.7
Australia	24.5	0.1	0.4	22.0
Kazakhstan	14.1	0.1	0.6	11.3
Spain	6.9	0.1	1.2	17.9
Syria	2.0	0.1	3.6	9.7
Morocco	6.2	0.1	1.1	19.4
Romania	8.6	0.1	0.8	11.9
Greece	1.5	0.1	4.4	15.8
Ethiopia	4.4	0.1	1.4	23.4
World	739.5	5.6	0.8	1292.5

Table 6.2: National wheat production (FAO, 2020) and absolute and relative wheat production losses for the countries with the highest absolute losses.

6.3.4 The relevance of slope and field management assumptions

The impact of the assumption that the most common slope represents the whole grid cell is examined for Italy in Figure 6.6. The plots compare the distribution of modelled maize and wheat yield losses due to water erosion for cases in which all cropland is either on the most common slope class or on the flattest terrain in each grid cell. Median annual maize and wheat yield losses for the flattest terrain assumption are 0.2% and 1.2%, respectively, leading to annual maize and wheat production losses of 0.01 million tonnes and 0.04 million tonnes, respectively. For the most common slope scenario, median annual maize and wheat yield losses are 2.1% and 4.1%, with substantially higher annual maize and wheat production losses of 0.05 million tonnes and 0.1 million tonnes, respectively.



Figure 6.6: Range of simulated maize and wheat yield losses (% yr⁻¹) in Italy simulated with different cropland distribution scenarios for maize (a) and wheat (b). Boxes illustrate medians and 25^{th} and 75^{th} percentiles, whiskers illustrate values between the 10^{th} and the 90^{th} percentiles. Grey bars mark the baseline scenario used for the main results of this study.

The uncertainty due to lacking field management information varies around the globe and is most pronounced in erosion-sensitive areas, where soil conservation techniques can reduce extreme water erosion rates considerably. In those areas, contrasting field management scenarios generate a large range of values with varying degrees of water erosion impacts on crop yields (Figure 6.7). This large uncertainty range is reduced in the baseline scenario by identifying and removing field management practices that are unlikely to be used in specific regions. However, due to the multitude of field management practices around the world, this uncertainty can only be limited in part.

6.3.5 Evaluation of simulated crop yields

To evaluate simulated wheat and maize yields (Figure C.1), national yields reported by FAOSTAT are compared to national averages of the simulation outputs (Figure 6.8). National maize yields are evaluated for the most important maize producers, which account for 99.9% of global maize production. Due to the low agreement between simulated and reported national average maize yields for countries contributing only a small share to the global production, the smallest maize producing countries have been excluded to facilitate the evaluation in respect to the global focus of this study. The evaluation of the simulation of global wheat production is conducted with all countries as small wheat producing countries do not significantly influence the evaluation results.

With an R^2 of 0.58 simulated national maize yields are significantly related to reported values. The regression intercept of 1.71 and slope of 0.49 indicates overestimated maize yields at low yield levels and underestimated yields at higher yield levels. Simulated wheat correlates stronger to observed values than simulated maize yields. The comparison of simulated and observed national wheat yields results in



Figure 6.7: Uncertainty range of the impact of water erosion on maize (a) and wheat (b) yields due to contrasting management scenarios.

an R^2 of 0.65. The regression intercept of 0.45 and slope of 0.76 indicates overestimated wheat yields at low yield levels and underestimated yields at higher yield levels, as observed for maize yields but by a smaller magnitude. National production estimates for both maize and wheat are sufficiently simulated for most countries with R^2 values of 0.85 and 0.95, respectively.

Relative errors between simulated and reported national average maize and wheat yields and production are presented in Figure 6.9. Relative errors between \pm 30% are defined as acceptable and errors above a threshold of \pm 50% as high. The majority of average simulated maize yields of the 20 most important maize producing countries lay within an error range of \pm 30%. High producing countries with a larger error include Argentina, Ethiopia, Hungary, Mexico and the Philippines. The latter two exceed the \pm 50% threshold. The highest occurrence of countries with high errors exceeding \pm 50% is in Africa, followed by Central and South American countries, which is caused mostly due to an overestimation of simulated maize yields. In the majority of countries in North America and Europe, simulated maize



Figure 6.8: Comparison of average national maize (a) and wheat (b) yields simulated with EPIC and reported by the FAO. Comparison of national maize (c) and wheat (d) production simulated with EPIC and reported by the FAO (FAO, 2020).

yields are below reported yields.

For the majority of the 20 most important wheat producing countries average simulated wheat yields have an error within the magnitude of \pm 30%. Exceptions are average wheat yields in Argentina and Spain, which have a relative error of 32% and 36%. Generally, most countries have a relative error below the \pm 50% threshold on all continents. The highest proportion of countries with simulated wheat yields above the \pm 50% threshold is in Africa, mostly due to lower simulated values than reported values.

Relative errors between estimated national maize production and reported production for most of the top-20 maize producers lay within the \pm 30% margin defined as the acceptable error range. Relative errors for the major producers Ethiopia, Mexico, Russia, Philippines and Ukraine are high, exceeding the \pm 50% threshold. Countries with high relative errors are most abundant in Africa on a continental level and on a regional level in Western Asia followed by Central America. The majority of relative errors in Africa, America and Europe is positive, whereas in Asia the majority of countries has a negative relative error.

Relative errors of national wheat production estimates are lower than for national maize production estimates. Relative errors for most of the top-20 wheat producing countries lay within the acceptable \pm

30% margin. Argentina, France, Italy, Spain and United Kingdom have a slightly larger relative error, without exceeding the \pm 50% threshold. Countries exceeding the \pm 50% threshold are most abundant in Asia and Africa, mostly due to underestimations. Generally, wheat production on all continents is underestimated rather than overestimated.



Figure 6.9: relative difference between national maize (a) and wheat (b) yields simulated with EPIC and reported by the FAO. Relative difference between national maize (c) and wheat (d) production simulated with EPIC and reported by the FAO (FAO, 2020).

6.4 Discussion

6.4.1 Erosion-induced crop yield losses and fertiliser requirements for compensation

Previous studies suggest that soil loss rates up to 11 t ha⁻¹ are tolerable to maintain crop productivity for soils in the United States (Schertz and Nearing, 2006) and in Europe (Panagos et al., 2018) based on the assumption that fertiliser will compensate for nutrient runoff. On fields with higher water erosion rates, Panagos et al. (2018) assumed that crop productivity would reduce by 8%, based on a review of relevant studies on erosion-crop productivity relationships. Similarly, the estimates of this study suggest a median global reduction of maize and wheat yields of 6% for grid cells with water erosion of at least 11 t ha⁻¹. In fields with water erosion below 11 t ha⁻¹ a considerably lower median crop yield reduction of 1% is simulated. However, large variations in fertiliser input between countries affect the impact of water erosion on crop yields. If fertiliser were not sufficiently supplied to compensate for nutrient losses in certain countries, their crop yield losses would likely be higher than in countries with both higher water erosion and fertiliser application rates (Balkovič et al., 2018). Although synthetic fertilisers can quickly compensate for nutrient loss, the recovery of lost organic matter and the consequent damage to soil structure can take decades (Poulton et al., 2018). Therefore, acceptable soil loss rates should not consider only the extent to which fertiliser application can replenish soil fertility. An assessment should also consider soil formation rates and off-site concerns such as the proximity to sensitive areas (Montgomery, 2007; Schertz and Nearing, 2006).

The additional fertiliser costs to compensate for water erosion can be higher than the loss of income due to production losses (Graves et al., 2015). Global median nitrogen runoff of 7 kg ha⁻¹ yr⁻¹ in maize fields and 5 kg ha⁻¹ yr⁻¹ in wheat fields, which are estimated in this study, would cost USD 1.7 ha⁻¹ yr⁻¹ and USD 1.2 ha⁻¹ yr⁻¹. The global annual nitrogen fertiliser replacement costs for maize and wheat fields would be USD 642m and USD 255m, respectively. Although this is lower than the estimated annual maize and wheat production losses (USD 2.0bn and USD 1.3bn), replacement costs for lost nutrients would be considerably higher if phosphorus and potassium runoff would be considered. In addition, carbon runoff of median 107 kg ha⁻¹ yr⁻¹ and 72 kg ha⁻¹ yr⁻¹ in maize and wheat fields might add additional costs through nutrient replacement efforts such as manure application. On a global scale, the relative fertiliser replacement costs might be too low to incentivise farmers to introduce soil conservation measures, but they can be considerably higher for vulnerable areas (Hein, 2007).

For a comprehensive assessment of water erosion impacts, off-site impacts on surrounding environments such as the pollution of surface water and emission of greenhouse gases also need to be considered (Chappell et al., 2016; Tilman et al., 2001). Several studies estimate higher costs of off-site impacts due to erosion than on-site costs through production losses and fertiliser replacement (Görlach et al., 2004; Graves et al., 2015). In addition, sediment redistribution is not considered in this study as global assessments are currently based on simple water erosion models (Chapter 2.3.4). Topsoil accumulation in deposition areas may improve nutrient availability and soil properties and could offset the negative effects on crops in eroded areas (Bakker et al., 2007; Duan et al., 2016).

Due to high fertiliser use in major maize and wheat production areas, which are mostly located on flat terrain and in regions with lower rainfall erosivity than the global average, water erosion has had a low impact on annual global production losses in absolute terms. Vulnerable regions with potentially high crop yield losses are mostly outside major production regions and therefore they hardly affect changes in global maize and wheat production. Den Biggelaar et al. (2004a) also estimated a low impact of water erosion on a global scale, and concluded that the small losses would likely be masked over the short term by market fluctuations, weather, and other environmental perturbations. Furthermore, market mechanisms such as trade flows can considerably reduce production losses. Sartori et al. (2019) used a global market simulation model that accounted for market impacts of soil erosion, which reduced direct production losses by three times. Nevertheless, as erosion impacts are cumulative, they may cause more serious losses if erosion continues unabated over a long period of time (Den Biggelaar et al., 2004a), and could ultimately lead to total topsoil loss and the land being abandoned. Moreover, water erosion could be self-reinforcing, by decreasing the protective cover through reduced crop cover and residues on the soil surface (Ponzi, 1993).

¹based on global urea price for the period 2015–2019 taken from World Bank (2020a).

Slope inclination and precipitation intensity are the dominant environmental characteristics affecting water erosion (Chapter 5.3.2). Soil types are generally relevant in GGCM crop yield simulations (Folberth et al., 2016) and for erosion-productivity relationships (den Biggelaar et al., 2001; Lal, 1995), but on a global scale their impact on water erosion is small compared to slope steepness and precipitation. This means water erosion impacts are highest in hilly areas, in the tropics and in other regions with heavy precipitation. In countries with diverse environmental conditions, the variation in water erosion impacts is usually wide ranging and therefore a comparison of the extent of cropland vulnerable to water erosion should be further analysed on a sub-national scale.

6.4.2 Potential impacts of water erosion on livelihoods

High production losses from water erosion on a national or regional scale can severely impact livelihoods of farmers (Wynants et al., 2019). The agricultural sector of both sub-Saharan Africa and South Asia contributes roughly 16% to their GDP, compared to a worldwide share of approximately 4% (World Bank, 2020b). Moreover, food security is a pressing issue in those regions (von Grebmer et al., 2012). Whilst in some of these regions water erosion was recently reduced through programs improving land management (Nyssen et al., 2015), increasing crop demand through population growth and market effects led to re-cultivation of tropical steep slopes (Turkelboom et al., 2008) or soils prone to degradation (Wildemeersch et al., 2015). Pressures are likely to increase through climate change impacts on agriculture, which are projected to decrease agricultural productivity highest in low latitudes (lizumi et al., 2017; Rosenzweig et al., 2014), which will likely enhance food security issues (Knox et al., 2012; Wheeler and Von Braun, 2013). The impact of climate change on water erosion impacts is still unclear but projected increases in rainfall intensity (Olsson et al., 2019; Wang et al., 2014) and diminishing vegetation cover through increasing temperature (Zhao et al., 2017) may accelerate water erosion and its impacts on crop yields (Li and Fang, 2016). The simulation results presented here show that several countries in regions that are most affected by food security problems today and that are likely to be under high pressure in the future from population growth and climate change are most affected by high relative production losses due to water erosion.

6.4.3 Uncertainties in water erosion estimates

Uncertain slopes of modelled fields

Slope data is the most critical parameter for estimating water erosion. However, due to the uncertainty of the global land use datasets (Fritz et al., 2015; Lesiv et al., 2019), no explicit spatial links between maize and wheat growing areas and slopes can be established without on-site observations. Instead, the slope that covers the largest area in a grid cell is used to capture the slope that is most likely covered by most of the cropland. This approach represents the prevailing topographic differences of global crop production regions but cannot capture the heterogeneity of fields in certain areas. In an ideal situation where all cultivated areas are concentrated on the flattest terrain available, simulated water erosion impacts on

crop yields are reduced substantially. However, the distribution of cropland is based on more factors than the topography of land, such as the suitability of soil, climate and socio- economic circumstances or limitations such as land tenure and competing land use (Hazell and Wood, 2008; Nyssen et al., 2019).

Uncertainties in field management

Field management can vary substantially between regions, farming systems and farmers, and is based on a complex web of factors (Pannell et al., 2014). While the management scenarios bracket the range of field management intensities and soil surface coverage, the baseline scenario narrows down prevailing field management by selecting or excluding scenarios based on environmental- and country-specific indicators. Apart from similar approaches (Porwollik et al., 2019), no detailed representation of the diversity in global field management currently exists. Moreover, the field management scenarios are constant for every season and do not account for the farmer's actions to mitigate soil erosion, which might significantly reduce water erosion impacts (Tiffen et al., 1994).

However, one advantage of simulating constant field management is that the effects of water erosion on soil resources can be recorded over the long term, which otherwise might have been masked by technological advances such as higher yielding crop varieties, herbicides, insecticides, new planting technologies, and increased fertiliser input to compensate for sediment runoff (Littleboy et al., 1996). Moreover, the potential differences in water erosion impacts with different intensities of field management can be addressed, as the model outputs reflect the ability of cover crops, crop residues and low tillage intensity to decrease water erosion rates and to maintain and replenish soil nutrients. Although this reduces crop yield losses due to water erosion, it does not necessarily translate into higher crop yields due to other growth constraints being influenced by the choice of farming techniques. Since field management practices greatly influence crop yields in general, and water erosion in particular, improving their representation and understanding the decision processes of farmers responding to changing physical conditions in their fields would help to improve our understanding of water erosion impacts on crop yields.

6.4.4 Uncertainties in crop yield estimates

The agreement between simulated and reported yields can be negatively influenced by the lack of representation in field management, fertiliser input, irrigation strategies, crop cultivars, and factors influencing crops, such as windstorms, pests, diseases, weeds or freezing of seeds in winter (Balkovič et al., 2013). Typically, GGCMs underestimate yields in high-yielding countries and overestimate yields in low-yielding countries (Balkovič et al., 2014; Mueller et al., 2017). This is also indicated by Figure 6.8.

Non-weather-related events, which are not simulated, can be especially important in low-input regions where unsuitable management or pest outbreaks, may contribute substantially to yield variability (Mueller et al., 2017). In addition, the automatic application of irrigation to effectively minimise daily water stress, may lead to an overestimation of irrigation efficiency in less intensively managed fields. Other reasons can be that crop technologies are overestimated in low-yielding countries, whereas in highly developed countries, they may be underestimated (Balkovič et al., 2013).

Another important consideration for the robustness of crop yield estimates is that the choice of model used to simulate crops affects the results due to differences in model sensitivities. As a result, the simulated impact of soil, weather and management parameters on crop yields varies depending on the model. For example, variations in crop yields in response to temperature rise could be attributed in large part to differences in the simulation of this relationship between models (Asseng et al., 2013). A further discussion of uncertainties in crop models simulating the effects of climate parameters on crops can be found in Chapter 7.4.6.

Compared to other commonly used GGCMs, EPIC-IIASA repeatedly performs well. In Mueller et al. (2017) the performance of EPIC-IIASA for simulating maize and wheat was compared against 13 different GGCMs. EPIC-IIASA showed relatively high correlation coefficients with national reported maize and wheat yields compared to the other GGCMs. Similarly, in Yin and Leng (2021) the correlation of reported and simulated national maize yields using EPIC-IIASA was among the highest of the eight analysed GGCMs. Folberth et al. (2019) compared the agreement between simulated and reported maize yields among five different EPIC-based GGCMs. EPIC-IIASA agreed best with national reported yields among the five tested EPIC-based GGCMs. The robustness of yield estimates in the EPIC-based GGCMs was highly sensitive to simulated crop management, soil parameters and soil process reparametrisation, including hydrology and nutrient turnover. The necessary improvements in model mechanisms, parametrisation and agricultural data for model input and model evaluation to reduce errors have been addressed in detail in numerous studies (e.g., Folberth et al., 2019; Mueller et al., 2017).

6.5 Conclusion

In this chapter, EPIC-IIASA was used to analyse the vulnerability of maize and wheat producing regions to water erosion. Locations that are highly vulnerable to water erosion are concentrated in regions combining hilly terrain, strong precipitation and low fertiliser inputs. But water erosion has only a small impact on global maize and wheat production, because the major maize and wheat production areas are on relatively flat terrain and nutrient losses through water erosion are offset by high fertiliser applications. However, this compensation of soil loss with fertilisers to maintain crop yields hides the negative impacts of water erosion on soil resources and surrounding environments.

The simulation results provide the first globally-consistent and transparent analysis of the effects of water erosion on maize and wheat production. The most crucial data requirements to improve the robustness of simulated water erosion impacts on global crops include well-defined field data covering all global regions to evaluate water erosion estimates, higher-resolution global land use datasets and detailed information on field management patterns (Chapter 5.4). A better understanding of the soil protection and erosion control measures that are used in the cultivation of slopes in each region would make it possible to improve the representation of regions at risk. As these datasets are currently not

available in higher detail at the global scale, further research on water erosion impacts could focus on the most vulnerable regions by analysing land use patterns and all environmental circumstances on-site at a finer resolution. The high vulnerability to water erosion in sub-Saharan Africa, and parts of South Asia and Latin America, where future changes in population growth and climate could amplify land degradation processes, should be priorities for further research.

Chapter 7

The combined effects of climate change and water erosion on global maize and wheat yields

7.1 Introduction

Climate change impacts on agriculture include irregular rainfall and prolonged droughts, which are among the biggest threats for farmers. It has been suggested that interactions between climate change and land degradation are likely to exacerbate the negative impact on agriculture (Olsson et al., 2019). Although climate change and land degradation are interacting processes, which may affect farmers simultaneously, they have been studied separately in the past (Webb et al., 2017). As these interactions are highly complex, the impact of climate change on soil functions is seen as the largest source of uncertainty in any projections of the trends in key ecosystem services provided by the soil (Montanarella et al., 2016).

Water erosion is influenced directly through changing precipitation amount and intensity and indirectly through the impact of precipitation, temperature and atmospheric CO_2 on vegetation. In addition, interactions between the climate system, soil hydrology and microbial activity, as well as land use changes due to climate change may influence future water erosion rates (Nearing et al., 2004). Thus, the impact of climate change on water erosion is very diverse and can be both positive and negative. Due to the complex interactions between water erosion and climate change, future changes in water erosion rates are unclear. In a comprehensive review of studies on the effects of climate change on water erosion by Li and Fang (2016), the authors concluded that in most studies, future water erosion rates are increasing due to climate change. However, due to different geographic locations, climate scenarios, precipitation patterns, topographic conditions, and land management scenarios the result of prior studies on climate change impacts on water erosion are highly variable (Chapter 2.5). One study concluded that water erosion can be expected to increase where precipitation increases are significant, but if precipitation decreases, water erosion may both increase or decrease depending on the complex interactions between plant biomass, runoff, and erosion (Pruski and Nearing, 2002a). Many studies concluded that field management practices and planting dates are decisive in climate change impacts on water erosion and may sometimes affect changes in water erosion more than changes in precipitation (O'Neal et al., 2005; Pruski and Nearing, 2002a; Routschek et al., 2014). Positive climate change impacts on vegetation cover through warmer temperatures and CO_2 fertilisation may even mitigate impacts of increased precipitation on water runoff (Nunes et al., 2013).

The complex interactions between climate change and water erosion and the high variability of study results lead to difficulties extrapolating study results to larger scales. As most studies are focused on the United States, Europe and China (Li and Fang, 2016), direct insights on climate change impacts on water erosion do not exist in many world regions. Only three studies were found that include estimates of the impact of climate change on global soil erosion (Ito, 2007; Yang et al., 2003; Borrelli et al., 2020). These studies focused on estimating future changes in water erosion rates based on current land use and climate change. Ito (2007) included estimates of changing water erosion impacts on the soil carbon cycle. However, these studies do not consider interactions between water erosion and crops. As previous field scale studies have shown, the interactions between soil and biomass are critical to the effects of changing climate on water erosion.

Field scale studies on climate change impacts on water erosion have often used process-based crop models such as EPIC (Chapter 2.5). EPIC is also a commonly used model for global climate change impact assessments based on GGCMs (Chapter 3.5). Climate change impact assessments based on GGCMs are among the most important studies that provide information on the varying resilience of agriculture to climate change around the world. Although the models used to conduct these studies simulate interactions between soil resources and crop yields, soil degradation processes are not considered as they are not included in most models or have been neglected due to their complexity and lack of evaluation (Chapter 2.3.4).

In this chapter, model outputs from EPIC-IIASA generated under different climate change scenarios are used to link the analysis of climate change and water erosion impacts on maize and wheat. The modelling approach is similar to that in Chapter 5 and Chapter 6, except that different climate data are used. The model outputs are first used to identify hotspots where maize and wheat production is vulnerable to climate change and water erosion due to environmental conditions and geographical differences in fertiliser use and irrigated cropland. Subsequently, the model outputs are used to show the interactions between crops and water erosion in response to climate change and to discuss in which regions water erosion may increase in the future. The final results chapter provides estimates of global maize and wheat production losses due to water erosion under different climate change scenarios, which are compared to changes in maize and wheat production due to climate change.

7.2 Methodology

7.2.1 Summary of methodology

The daily growth of maize and wheat is simulated for the years 2005-2099 with and without considering water erosion under the two contrasting Representative Concentration Pathways RCP 2.6 and RCP 8.5, as well as different field management scenarios. Water erosion impacts are calculated using the baseline field management scenario presented in Chapter 4.3.4. The impact of climate change on crops is calculated without considering water erosion and with a static soil profile, which reinitialises all soil properties except plant-available nutrients and water at the beginning of each year. This disables the impact of soil degradation processes on crop yields to focus on the impact of changing climate parameters on crop yields. To further increase the focus on the impact of climate parameters on crops, only the field management scenario simulating conventional tillage without cover crops is considered to minimise the impact of different farming techniques on crop yield changes.

7.2.2 Aggregation of model outputs

Water erosion impact

Water erosion impacts on maize and wheat yields are calculated using the model outputs for the years 2050-2059 to estimate medium-term erosion impacts and for the years 2090-2099 to estimate long-term erosion impacts. The aggregation of the respective model outputs to estimate water erosion impacts on crops for grid cells, countries and regions follows the same approach as in Chapter 6.2.2.

Climate change impact

Crop yield changes due to climate change are calculated as the relative difference of average crop yields between the years 1980 - 2010 simulated with the AGMERRA climate data and the years 2050-2059 and 2090-2099 simulated under the RCP 2.6 and the RCP 8.5 scenario (Equation 7.2). Relative crop yield changes for each decade are based on average crop yields weighted by the irrigated and rainfed cultivation area (Portmann et al., 2010) of the respective crop per grid cell (Equation 7.1).

$$Yw_{cgd} = Yav(r)_{cg} * Af(r)_{cg} + Yav(i)_{cg} * Af(i)_{cg}$$

$$\tag{7.1}$$

$$dYrel_{cgd} = \frac{Yw_{cgd} - Yw(d0)_{cg}}{Yw(d0)_{cg}} * 100$$
(7.2)

 Yw_{cgd} is area-weighted mean crop yield (tha^{-1}) for crop c, grid cell g and decade d; Yav is yield averaged for the selected decade simulated under rainfed (r) and irrigated (i) conditions; Af(r) is the rainfed area fraction; and Af(i) is the irrigated area fraction; $dYrel_{cgd}$ is the relative change in area-weighted mean yields of crop c and grid cell g between the decade d and area-weighted mean crop yields of the years 1980 - 2010 (d0).

National crop production losses and gains due to climate change are calculated by first calculating the difference in crop yields between the selected decades and the years 1980 - 2010 (Equation 7.3), and subsequently multiplying the difference with the rainfed and irrigated area per crop of each grid cell and summarising all positive and negative values of all grid cells per country (Equation 7.4).

$$dYabs_{cwgd} = Yav_{cwgd} - Yav(d0)_{cwg}$$
(7.3)

$$dP_{lcd} = \sum_{g=1}^{n} dYabs(r)_{cgd} * A(r)_{cg} + dYabs(i)_{cgd} * A(i)_{cg}$$
(7.4)

 $dYabs_{cwgd}$ is the absolute yield change for crop *c*, irrigation scenario *w* and grid cell *g* and decade *d*; *Yav* is yield averaged across the selected decade *d* or the years 1980 - 2010 (*d*0); dP_{lcd} is the change of production (in tonnes) of crop *c* in country *l* and decade *d*; *n* is the number of grid cells in country *l*; dYabs(r) is the absolute change in rainfed yields and dYabs(i) is the absolute change in irrigated yields; A(r) is the rainfed area (in *ha*); and A(i) is the irrigated area (in *ha*).

7.3 Results

7.3.1 Regions and countries vulnerable to both climate change and water erosion

The impacts of water erosion on maize and wheat yields are shown in Figure 7.1 and Figure 7.2 as medium-term (2050s) and long-term (2090s) relative declines in crop yields due to water erosion under the RCP 2.6 and RCP 8.5 scenario, respectively. In addition, regions with negative and positive climate change impacts on maize and wheat are coloured in the maps. The plots next to the maps illustrate the median changes in crop yields due to climate change per latitude.

A region's vulnerability to climate change and water erosion is strongly determined by geographic differences in environmental characteristics. While topography, precipitation amount and intensity largely determine the extent of water erosion in a region, the effects of climate change are strongly dependent on latitude. At low latitudes, defined here as latitudes between \pm 30 degrees, negative climate change impacts dominate for both maize and wheat in both RCP scenarios and both observed time periods. At mid- to high-latitudes, the distribution between positive and negative climate change impacts is more variable and depends strongly on the crops and the rate of climate change.

Under the RCP 8.5 scenario, maize yields decline at most latitudes. The decline is most severe at low latitudes and worsens over the long term. Under the RCP 2.6 scenario, climate change impacts on maize are less severe and can be both positive and negative at mid- to high-latitudes, but are mostly negative at lower latitudes. Wheat yields in temperate latitudes in both hemispheres are largely positively affected by climate change under both RCP scenarios. The negative impacts of climate change on wheat yields are most severe and widespread at low latitudes, especially in the long term under the RCP 8.5 scenario.

The global patterns of water erosion impacts on maize and wheat yields are similar under both RCP

scenarios and largely follow the patterns described in Chapter 6. The strongest impacts of water erosion on maize and wheat are observed in hilly terrain and in regions with heavy precipitation, most commonly in tropical climates. While the regions most vulnerable to water erosion remain the same over time for both RCP scenarios, the magnitude of water erosion impacts increases globally over time under both RCP scenarios.

To illustrate the differences among global regions in their vulnerability to climate change and water erosion, the medium- and long-term relative changes in maize and wheat yields due to climate change and water erosion per region are summarised in Figure 7.3 and Figure 7.4. Trends in yield changes due to climate change and water erosion, as well as areas affected by water erosion over the entire simulation period, are shown in Figures D.3,D.4,D.5.

The regions with the greatest impact of water erosion on affected maize fields at the end of the simulation period include Western, Southern and Southeast Asia (16%, 14%, 10% decline in median maize yields simulated with both RCP scenarios, respectively), Eastern Africa (14%) and Southern Europe (11%) (Figure 7.3). Unlike water erosion impacts, climate change impacts on maize vary greatly among RCP scenarios. In all regions, the most negative climate change impacts on maize are projected for the 2090s under the RCP 8.5 scenario. The most affected regions are Western Africa (61% decline in median maize yields compared to 1980-2010), Western Europe (44%), Southeast Asia (41%), Eastern Europe (39%) and Central Asia (38%). The positive effects of climate change on maize at the end of the simulation period are greatest in the RCP 2.6 scenario. The most positively affected regions are Southern Africa (14%), Western Asia (3%), Australia and New Zealand (3%), Eastern Asia (3%), and Central Asia (2%). In some regions, the positive effects of climate change on maize are greatest in the medium term and decrease in the long term (Northern Europe, Northern Africa, Australia and New Zealand), while in other regions the positive effects increase over time (Southern Africa, Western Asia, Eastern Asia).

The decline in wheat yields due to climate change in the most affected regions is greatest at the end of the simulation period and under the RCP 8.5 scenario. Wheat yields are projected to decline the most in Western Africa (59%), Southeast Asia (58%), Central America (53%), Southern Asia (31%) and Eastern Africa (27%) (Figure 7.4). The largest positive impacts of climate change on wheat are also simulated at the end of the simulation and under the RCP 8.5 scenario. The regions with the highest increase in wheat yields by the 2090s include Western and Central Asia (20% and 17%, respectively), Northern America (13%), Northern Europe (13%), and Eastern Africa, South and Southeast Asia, and Central America. Similarly, the largest yield losses in wheat fields due to water erosion are projected for the same regions as for maize, but to a slightly lesser extent: Western, Southern and Southeast Asia (9%,7%,14%, respectively), Eastern Africa (8%) and Southern Europe (10%). Water erosion impacts on maize and wheat and cropland affected by erosion are greatly influenced by the management scenarios (Figure D.6, D.7).

To narrow down where crop yields are affected by both water erosion and adverse climate change impacts, Figure 7.5 shows aggregated medium- and long-term relative crop yield changes at the country



Figure 7.1: Relative reduction of maize yields due to water erosion per grid cell in the medium-term (2050-2059) and the long-term (2090-2099) simulated with the RCP 2.6 (a-b) and the RCP 8.5 scenario (c-d). Blue and red grid cells indicate positive and negative climate change impacts (CCI) on maize yields. The plots on the right to each map indicate median maize yield changes per latitude due to climate change (e-h). Climate change impacts are calculated by comparing mean maize yields from 2050-2059 and 2090-2099 simulated with the RCP scenarios to mean maize yields from 1980-2010 simulated with the AGMERRA dataset.



Figure 7.2: Relative reduction of wheat yields due to water erosion per grid cell in the medium-term (2050-2059) and the long-term (2090-2099) simulated with the RCP 2.6 (a-b) and the RCP 8.5 scenario (c-d). Blue and red grid cells indicate positive and negative climate change impacts (CCI) on wheat yields. The plots on the right to each map indicate median wheat yield changes per latitude due to climate change (e-h). Climate change impacts are calculated by comparing mean wheat yields from 2050-2059 and 2090-2099 simulated with the RCP scenarios to mean wheat yields from 1980-2010 simulated with the AGMERRA dataset.

level.

In most countries, median maize yield losses in areas affected by water erosion are higher than the negative medium- and long-term impacts of climate change under the RCP 2.6 scenario. Under the RCP 8.5 scenario, the negative impacts of climate change on maize increase substantially and thus outweigh the impacts of water erosion in the long term in most countries. The countries most affected by both the negative impacts of climate change and water erosion are mostly located in Southern Europe, sub-Saharan Africa, Central and South America and South and Southeast Asia.

In several countries where water erosion impacts on wheat yields are most severe, climate change impacts are positive in the RCP 2.6 scenario. Under the RCP 8.5 scenario, both the positive and negative impacts of climate change on median wheat yields increase in most countries. Countries with greatest negative impacts of climate change include several countries from Central and South America, South and Southeast Asia and sub-Saharan Africa that are vulnerable to water erosion. Wheat yields in several countries with high water erosion risk areas are positively affected by climate change, especially in Northern America, Europe, and Eastern Asia.

7.3.2 Interactions between crops and water erosion in response to climate change

The impacts of climate change on water erosion and crop yields are shown in Figure 7.6, which plots relative changes in water erosion rates and crop yields against relative changes in precipitation and absolute temperature changes. The values shown in Figure 7.6 reflect the relationship between long-term changes in climate parameters based on the two RCP scenarios and changes in crop yields and water erosion in the low and mid to high latitudes. In wheat fields, latitude has a notable effect on the relationships between changing climate parameters, crop yields, and water erosion, while for maize fields these relationships vary less between latitudes.

At all latitudes, water erosion in maize fields mostly increases with increasing precipitation, while a strong decline in precipitation results in less water erosion. When precipitation decreases only slightly, water erosion mostly increases. Decreasing precipitation is causing maize yields to decline at all latitudes. An increase in precipitation reduces maize yields in low latitudes, while maize yields in mid- to high-latitudes change little. Small temperature increases up to 2 °C hardly affect maize yields, whereas greater warming reduces maize yields at all latitudes. Water erosion mostly increases with rising temperatures at all latitudes. Despite minor differences at low levels of warming, the relationships between temperature, maize yields, and water erosion are similar at all latitudes.

Water erosion in wheat fields increases with increasing precipitation and decreases at all latitudes when precipitation decreases by more than 30%. In most cases, a smaller decrease in precipitation increases water erosion at low latitudes. Wheat yields at low latitudes decrease considerably with both increasing and decreasing precipitation, while wheat yields at higher latitudes increase with increasing precipitation. The effects of temperature on wheat also vary greatly depending on latitude. While mid-and high-latitude wheat yields increase with increasing temperature, they decrease significantly at low latitudes. Water erosion in wheat fields increases significantly with increasing temperature at low latitudes.



Figure 7.3: Relative medium and long-term changes in maize yields due to climate change and water erosion in major world regions. Boxes illustrate median changes and the range of values between the 25th and 75th percentile. Yield changes are averaged over the RCP 2.6 scenario and the RCP 8.5 scenario. Black squares indicate the variability of the median changes between both RCP scenarios. Values on the top right in each plot indicate the share of grid cells per region, where water erosion reduces crop yields.

tudes, while changes in water erosion are smaller at mid- and high-latitudes and vary between positive and negative changes.

The above relationships illustrate that an increase in precipitation is the most obvious impact of climate change on water erosion (Figure D.9, D.10). The effects of increasing precipitation on water erosion vary by crop, field management, RCPs and latitude (Figure 7.7). Water erosion increases most with increasing precipitation at low latitudes in both RCP scenarios. The most significant difference is simulated for wheat fields under the RCP 8.5 scenario, where water erosion increases by median 3.5% per 1% increase in precipitation at mid to high latitudes and by median 11.4% per 1% increase in precipitation at low latitudes. In most cases, water erosion increases more under the RCP 8.5 scenario than under the RCP 2.6 scenario. Only in wheat fields located in mid to high latitudes, median increase in



Figure 7.4: Relative medium and long-term changes in wheat yields due to climate change and water erosion in major world regions. Boxes illustrate median changes and the range of values between the 25th and 75th percentile. Yield changes are averaged over the RCP 2.6 scenario and the RCP 8.5 scenario. Black squares indicate the variability of the median changes between both RCP scenarios. Values on the top right in each plot indicate the share of grid cells per region, where water erosion reduces crop yields.

water erosion is slightly lower under the RCP 8.5 scenario than under the RCP 2.6 scenario. The impact of field management scenarios on precipitation-erosion relationships is indicated by the variability of median values in Figure 7.7. In most cases, the range between median changes in water erosion simulated under different field management scenarios is larger than the range between median changes in water erosion due to different RCP scenarios and latitudes.

7.3.3 Global production changes due to climate change and water erosion

At the global level, annual production losses of maize and wheat due to climate change are strongly influenced by the RCP scenario and time scale (Figure 7.8). Average annual production losses for maize due to climate change in the 2090s compared to 1980-2010 are 120 and 290 million tonnes for RCP


Figure 7.5: Relative medium and long-term changes in median maize and wheat yields due to climate change and water erosion per country and RCP. Country names mark top five countries per panel with highest erosion impact and negative climate change impacts. The size of the dots shows the relative extent of the national cropland affected by water erosion. Smallest maize and wheat producers have been excluded (<0.001% share of global production).

2.6 and RCP 8.5, respectively, while production gains are 20 and 7 million tonnes, respectively. Wheat production losses due to climate change in the 2090s are 40 to 80 million tonnes for RCP 2.6 and RCP 8.5, respectively, while production gains are 80 to 100 million tonnes for each. Changes in global maize and wheat production are mainly driven by the main growing regions in Europe, Asia and North America (Figure D.11).

Compared to the climate change impacts on maize and wheat production between contrasting RCP scenarios, the variations in production losses due to water erosion are small. Average annual production losses for maize due to water erosion in the 2090s are 14 and 11 million tonnes for RCP 2.6 and RCP 8.5, respectively, and production losses for wheat reach 13 million tonnes for both RCPs. Relative to the global production of maize and wheat reported by FAOSTAT (2013-2018) of 1091 million tonnes and 739 million tonnes, respectively, the production losses of maize and wheat due to water erosion are 1% and 2%, respectively. While production losses are similar across the different RCP scenarios, production



Figure 7.6: Relative change in water erosion and crop yields in maize fields (a-d) and wheat fields (e-h) due to changing precipitation and increasing temperature for the years 2090-2099 compared to the years 2010-2019 and aggregated for both RCP scenarios. Changes in water erosion and crop yields are log-transformed. Changes in precipitation are relative, while changes in temperature are absolute. Dots illustrate median changes and shadows illustrate the range of values between the 25th and 75th percentile. Blue represents values from mid to high latitudes, red represents values from low latitudes.



Figure 7.7: Increase in water erosion (%) per 1% increase in precipitation for contrasting crops, RCPs and latitudes for the years 2090-2099 compared to the years 2010-2019. Boxes illustrate medians, 25^{th} and 75^{th} percentiles. Whiskers mark 10^{th} and 90^{th} percentiles. Points illustrate the variance of the medians due to different field management scenarios.

losses vary significantly between different field management scenarios (Figure D.12).



Figure 7.8: Changes in global annual average maize and wheat production due to climate change and water erosion in the 2050s and the 2090s for RCP 2.6 and RCP 8.5.

7.4 Discussion

The simultaneous effects of climate change and water erosion on maize and wheat yields vary widely across regions. The factors influencing these differences are diverse and complex, and require detailed discussion, which will form the first part of this discussion chapter. Potential regional changes in future water erosion due to climate change are then discussed based on the simulated interactions between climate parameters, crops, and water erosion. Because the impacts of water erosion and climate change on crops are strongly influenced by land use and field management, their role in mitigating the negative impacts is discussed subsequently. The final part of the discussion revolves around model uncertainties in simulating changes in future crop growth and water erosion.

7.4.1 Factors driving regional differences in climate change and water erosion impacts

The model outputs suggest that maize and wheat production in sub-Saharan Africa, South and Southeastern Asia, Central and South America include most regions with highest water erosion impacts and is overwhelmingly negatively affected by climate change. Although maize and wheat production in other regions may be also heavily affected by climate change and water erosion, wide areas where both crops are severely affected by both impacts are less common. Hotspots where crop production is susceptible to both climate change and water erosion impacts are determined by many factors such as regional characteristics of climate and topography, as well as farmer's management of crops, soil, fertiliser and irrigation.

Environmental factors

The environmental factors that most significantly influence the simulated differences of water erosion impacts in global regions are slope steepness and precipitation. Therefore, hilly agricultural land exposed to strong precipitation is most vulnerable to water erosion (in accordance with the results of Chapter 5 and Chapter 6). The impact of other environmental factors on water erosion, most specifically soil types and soil texture, are not further discussed as the strong sensitivity of slope and precipitation in the water erosion model used for this study strongly determine regional differences at this scale of analysis

(Chapter 5.3.2). In addition, global soil datasets are highly uncertain (Folberth et al., 2016).

The energy of precipitation to initiate water erosion can be expressed by the rainfall erosivity factor, which is strongest in tropical regions (Panagos et al., 2017). In the same climate regions, maize and wheat are most negatively affected by climate change. Generally, the regional differences in climate change impacts are strongly influenced by changing climatic conditions along latitudes. The model outputs of this study confirm previous studies, which suggested that negative climate change impacts on maize and wheat increase from high to low latitudes (Rosenzweig et al., 2014; Iizumi et al., 2018; Zhao et al., 2017; Mora et al., 2015).

The spatial differences in climate change impacts can vary significantly between crops. Each crop has a defined temperature range within which growth occurs and an optimum temperature at which plant growth progresses fastest (Hatfield et al., 2011). Therefore, warming can both increase and decrease crop yields by shifting temperature inside or outside the optimum temperature range for the respective crop. Maize has an optimum temperature range of 18 - 30 °C, which varies depending on the maize cultivar and differs for individual crop growth stages (Ramirez-Cabral et al., 2017; Sánchez et al., 2014). The optimum temperature of wheat is more narrow and is generally considered to be between the range 17 - 23 °C over the course of an entire growing season (Porter and Megan, 1999). The hill-shaped yield-temperature function determines that decreases in yields due to increasing temperatures are larger the warmer a country is (Lobell and Burke, 2008). In this study, the long-term impacts of increasing temperatures above the optimum for wheat. Thus, wheat in warm regions is highly susceptible to warming. For maize, the contrast in the reduction of crop yields with increasing temperatures between higher and lower latitudes is visible for small levels of warming, whereas under higher levels of warming, maize yields are reduced widely in all latitudes.

Next to changes in mean temperature, crop yield changes due to climate change are also determined by extremes of temperature and precipitation at key stages of crop development (Slingo et al., 2005). Tolerance to temperature stress and water stress during different crop cycle periods can be very variable between crops and cultivars and thus the effects of changing weather extremes on crops are highly complex. For example, strong precipitation events have been observed to reduce crop yields due to flooding (Rosenzweig et al., 2002). In addition, anomalously low precipitation often reduces crop yields due to stress effects from droughts, especially during critical crop development stages (Grassini et al., 2009; Krishna Kumar et al., 2004; Sivakumar et al., 2005; Verdin et al., 2005). The model results show that wheat at low latitudes is more susceptible to decreasing precipitation than at higher latitudes. An increased susceptibility of crops to reduced rainfall may be caused by increased evapotranspiration through warming, which can increase drought stress (Carter et al., 2016). Generally, precipitation changes are an important driver of interannual yield variability, but are less relevant for long-term crop productivity trends (Lobell and Burke, 2008; Lobell et al., 2011).

Management factors

Next to environmental conditions, differences in fertilisation and irrigation management determine global differences of both water erosion and climate change impacts significantly. Mueller et al. (2012) estimated that 60 - 80 % of global yield variability for most major crops are related to climate, fertiliser and irrigation. Irrigation and fertiliser use varies extensively around the globe. High fertiliser application rates are concentrated in high-income and some rapidly developing countries (Mueller et al., 2012) (Figure A.2). Irrigation is heavily concentrated in South, West and East Asia, as well as parts of the United States, where water availability is an important limitation for crop growth (Mueller et al., 2012; Portmann et al., 2010) (Figure A.3). However, in many developing regions where crops are limited by water availability, irrigation is lacking, which increases the vulnerability of crops to weather variability (Rosegrant et al., 2009; Van Ittersum et al., 2013). Sub-Saharan Africa is often named as a large region where crops suffer nutrient and water deficiencies leading to currently low levels of agricultural productivity (GYGA, 2015; Sánchez, 2010; Cassman and Grassini, 2013; Folberth et al., 2013). In addition, outside of the major maize and wheat growing areas, there are large yield gaps due to lack of irrigation and fertilisation, indicating potential to improve agricultural resilience in many regions vulnerable to climate change and water erosion (Licker et al., 2010; Mueller et al., 2012; Jägermeyr et al., 2016).

In summary, geographic differences in environmental characteristics and field management practices result in varying crop vulnerability to climate change and water erosion around the world. The climatic conditions at low latitudes combine high levels of rainfall erosivity and high susceptibility of maize and wheat to rising temperatures, two important factors for the adverse effects of climate change and high rates of water erosion. In addition, low application of fertilisers and lack of irrigation infrastructure reduce the resilience of crop production systems to both soil degradation and climate change in several developing regions. On the contrary, the largest maize and wheat producing regions in Europe, Asia and North America are predominantly on flat terrain with high fertiliser inputs and often have an abundant irrigation infrastructure. Hence, resilience to the effects of climate change and water erosion is higher in these regions.

7.4.2 The effects of precipitation and temperature changes on water erosion

An increase in water erosion with increasing precipitation is the most consistent relationship between changing climate parameters and water erosion in all regions and for both crops. However, the rate of increase in water erosion varies widely between regions, RCP scenarios, field management and crops. Similarly, the impact of changing precipitation amounts on simulated water erosion varies widely in previous studies, which are based on different land use and climate change assumptions (Chapter 2.5).

According to the model results, the effects of increasing precipitation on water erosion in maize and wheat fields is strongest under the RCP 8.5 scenario at low latitudes. In addition, water erosion increases sharply in regions where rising temperatures reduce crop yields. At locations where warming does not reduce crop yields, water erosion does not change significantly. This suggests that the adverse effects of climate change on crops, and thus on the protective vegetation layer against rainfall and runoff energy,

largely contribute to the spatially varying effects of precipitation changes on water erosion. Decreasing plant cover due to lack of precipitation could also explain the positive correlation between water erosion and slightly decreasing precipitation at several sites.

As such, the model outputs demonstrate the decisive role of land use in the relationships between climate change and water erosion. The different relationships between warming and water erosion in high-latitude maize and wheat fields indicate that the effects of climate change on water erosion can be mitigated by growing crops that are not negatively affected by warming. In addition, the large variability in the effects of increasing precipitation on water erosion between different field management scenarios suggests that the relationship between precipitation and erosion may be predominantly influenced by field management rather than the magnitude of climate change. This is in line with previous suggestions that changes in water erosion are primarily due to field management rather than climate change (Olsson et al., 2019).

Nevertheless, the model outputs from this study demonstrate that changes in precipitation contribute substantially to changes in water erosion. For example, increasing precipitation largely increases water erosion in wheat fields despite positive climate change impacts on wheat in mid and high latitudes. Despite some clear relationships between climate change and water erosion, there is large complexity in the impacts of climate change on water erosion as some of the interactions between climate, crops and water erosion are difficult to disentangle. For example, the relationship between decreasing precipitation amounts and water erosion is very variable as it controls both crop cover and rainfall runoff and can be both positive and negative. Also, the reasons for the increasing water erosion impacts on crops over time are difficult to identify. This can be caused by impacts of climate change on water erosion.

7.4.3 Where is water erosion increasing due to climate change?

Spatial uncertainty in precipitation projections makes it difficult to identify regions where water erosion could increase

The model outputs of this study suggest that water erosion rates increase strongest in Eastern, Southern, and Southeast Asia throughout the simulation period (Figure D.8). The changes in water erosion are most directly driven by changes in precipitation, but these are difficult to project (Giorgi and Lionello, 2008).

It is well known that atmospheric moisture increases with warming. However, agreement about the spatial patterns of precipitation changes among different models is low. The largest regions where most models agree on an increase in precipitation at the end of the 21st century are in the highest latitudes including vast areas in Canada, Northern Europe, and Northern and Central Asia (Trenberth, 2011; Fischer et al., 2014; Pendergrass et al., 2017). In addition, several models predict an increase in precipitation in large areas of Eastern, Southern, and Southeast Asia, Eastern Africa and parts of South America, with moderate agreement between models in these regions. In other regions model agreement is low or models agree on a reduction of precipitation. The changes in water erosion in regions that are becoming drier

are difficult to determine, as the model results of this Chapter show.

The agreement in projected increases in heavy precipitation is significantly higher among models and affects the majority of agricultural regions on all continents (Fischer et al., 2014). Pendergrass (2018) finds that extreme precipitation events are expected to increase faster than moderate precipitation events with increasing atmospheric moistening. Several studies already observed the intensification of precipitation with warming (Guerreiro et al., 2018; Blenkinsop et al., 2018; Liu et al., 2009; Bezak et al., 2020). Some observations even demonstrated that precipitation intensity can increase at locations where precipitation amounts decrease (Trenberth, 2011). Based on numerous studies on the relationships between climate change and precipitation, the IPCC concluded that heavy precipitation events have increased in frequency, and/or amount since 1950 (likely) and that further changes in this direction are likely to very likely during the 21st century (Stocker et al., 2013).

However, uncertainty about the spatial patterns of projected extreme precipitation changes remains. Tabari et al. (2019) showed that the uncertainty of modelled extreme precipitation changes due to variability between models is highest in tropical regions where rainstorms cannot be adequately represented by coarse-scale climate models, and where observation networks are sparse on which climate models can be tuned and improved.

Previous studies on spatial changes in erosion due to precipitation changes used an ensemble of climate models to capture this uncertainty (Panagos et al., 2022; Borrelli et al., 2020). For example, using 19 GCMs and three RCPs, Panagos et al. (2022) showed changes in rainfall erosivity by 2050 and 2070 that increased in most parts of the global land surface and were greatest in arid and cold zones. Rainfall erosivity generally increased with the RCP scenario. Whilst there was a strong agreement on the future directional trend of rainfall erosivity among the ensemble, the authors concluded that predicting changes in magnitude are uncertain, especially in low latitudes where the variability between models are very high.

Negative effects of warming on crops can help identify regions where water erosion could increase due to climate change

In addition to the direct impact of precipitation on water erosion, the indirect effects of increasing average temperatures on water erosion due to the decreasing crop cover are important for the localisation of hotspots of future changes in water erosion. The model outputs of this study demonstrate that in regions with severe negative climate change impacts on crop yields, water erosion will likely increase. Consequently, water erosion impacts are more likely to increase at low latitudes where important staple crops are more negatively affected by climate change. Combined with the expected increase in extreme rainfall in many tropical regions, this will particularly exacerbate erosion.

While the model results from this study can be used to explore key mechanisms that may lead to changes in future water erosion in different global regions, it is difficult to predict actual changes due to the uncertainties in climate change projections. In addition, other links between climate change and water erosion that were not considered in the analysis of model results must also be taken into account.

For example, another important impact of climate change on water erosion are temperature impacts on soil moisture through evapotranspiration, which influences the partitioning of water into surface and subsurface runoff (Li and Fang, 2016; Nearing et al., 2004). However, soil moisture changes depend on both temperature and precipitation; thus, they can be very variable around the globe and difficult to project (Trenberth, 2011). Next to the uncertainties in climate change projections, future land use is highly uncertain. Future water erosion rates will be strongly influenced through shifts in crop cultivation areas (Zabel et al., 2014) or changes in field management, which may be influenced by strategies for adapting to climate change (Borrelli et al., 2020).

7.4.4 Field management and land use planning will largely influence the future impact of water erosion

The model results show that climate change impacts on maize and wheat yields are strongly affected by RCPs, whereas water erosion impacts are most affected by field management. Due to the static field management scenarios and the uncertainties in the cropland distribution within the spatial unit studied, the modelling approach neglects many opportunities for farmers to reduce water erosion impacts. Nevertheless, the contrasting field management scenarios show that high production losses due to water erosion can be substantially reduced with farming techniques, which reduce the erosivity of precipitation and surface runoff. However, such strategies can limit the overall production volume as there may be a yield 'penalty' from employing Conservation Agriculture practices in the short-term (Giller et al., 2009). The trade-offs between farming techniques, crops and soil in response to climate change are highly complex; thus, the effectiveness of field management strategies to maintain or increase productivity should be analysed in areas susceptible to water erosion based on detailed data on environmental conditions and land use on-site.

Key to directly address water erosion will be to monitor cropland expansion. Increasing crop demand has already led to cultivation of tropical steeplands (Turkelboom et al., 2008) or soils prone to degradation (Wildemeersch et al., 2015). In many regions of Africa, Asia and Latin America, which are predicted to be most vulnerable to both climate change and water erosion, population growth and a high dependence on agriculture may lead to unpredictable impacts of cropland expansion on soil erosion. Due to the uncertainties in the spatial distribution of cropland in each grid cell in the modelling approach of this study, the model outputs reflect potential production losses that occur if the susceptibility of land to water erosion is not considered in the choice of cropland. Given the uncertainties in the distribution of current and future cropland and its significance on the resilience of agriculture to water erosion, further research and monitoring programs in the most affected regions are necessary.

7.4.5 Measures to reduce the adverse effects of water erosion and climate change can be combined

Since land degradation affects the resilience of agricultural systems to climate change, adaptation planning must consider the risks associated with ongoing land degradation to increase the effectiveness of climate change adaptation strategies (Reed and Stringer, 2016). To some extent, anti-erosion measures are integrated indirectly in climate change adaptation strategies, which maintain or increase crop cover. The most effective measures to reduce climate change impacts in food insecure regions are likely to arise from new crop varieties and new technology such as improved irrigation infrastructure (Lobell et al., 2008). Other adaptations strategies are shifting planting dates, modifying crop rotation and the up-take of pre-existing crop varieties (Knox et al., 2012), and are already in use to cope with variable weather conditions and shifting climate (Osbahr et al., 2010). Moreover, certain climate change adaptation strategies focus on the protection of soil moisture and soil fertility through increasing amounts of organic matter in the field, artificial and natural fertiliser application and lowering soil disturbance; thus, they include similar measures necessary to reduce water erosion impacts (Giller et al., 2015).

Whilst the flexibility of 'Climate-Smart Agriculture' as an option for addressing both land degradation and climate change has been recognised, the focus on land degradation needs to be increased in adaptation planning (Webb et al., 2017). A reduction in soil health may limit the adaptive capacity of farmers by increasing their socio-economic vulnerability and in addition, influences the exposure and sensitivity of crops to climate change (Webb et al., 2017; Laban et al., 2018). Despite uncertain climate change impacts, previous authors are confident that negative climate change impacts will be severe for low resilient agricultural systems in parts of Africa, Asia and South and Central America (Müller et al., 2011; Knox et al., 2012; Lobell et al., 2008).

7.4.6 Uncertainties related to crop yield projections and climate change impacts on crops and water erosion

The most important model uncertainties for estimating global water erosion impacts include the previously discussed uncertain resolution of the model input data (Chapter 6.4.3), global patterns of field management (Chapter 6.4.3), and the varying performance of the water erosion model in different global regions (Chapter 5.4). The simulation of crop yield projections in this chapter adds further uncertainty to the model outputs.

Future changes in field management and technologies have not been considered but will significantly influence the resilience of future maize and wheat production to climate change. For example, negative climate change impacts from warming such as droughts can be reduced through the adaptation of farming techniques, planting dates or the use of newly developed maize and wheat cultivars (Carr et al., 2022). Moreover, expansion of irrigation infrastructure or changes in water availability may alter climate change impacts (Balkovič et al., 2014). In addition, this study did not consider shifts in future crop production areas due to geographically changing climatic suitability for maize and wheat (Wu et al., 2007). Obviously, these changes will have a significant impact on future maize and wheat yields and production but are highly uncertain.

Additional important uncertainties include the robustness of RCPs and the simulated impact of climate parameters on crops, as well as water erosion. As these uncertainties directly affect the model outputs of this study they are discussed in more detail in the following.

Uncertain climate change scenarios

The RCPs chosen for this study represent two contrasting scenarios of possible future trajectories of the climate systems and climate change impacts on agriculture. Thus, they capture a wide spectrum of climate change impact possibilities, which are difficult to narrow down as they depend on future policy and socio-economic changes.

Under the RCP 2.6 scenario, global warming could be limited to $< 2 \,^{\circ}$ C above pre-industrial levels, which is defined as the long-term temperature target by most countries in the Paris Agreement (Schleussner et al., 2016). However, the likelihood that the intended national contributions are sufficient to meet the goals of the Paris Agreement is uncertain (Rogelj et al., 2016; Clark et al., 2020). Gasser et al. (2015) concluded that negative emissions (through capturing human-produced CO_2 , direct removal of carbon dioxide from the atmosphere, or engineered enhancement of natural carbon sinks) are physically needed to keep global warming below 2 $^{\circ}$ C even in the case of very high conventional mitigation rates (e.g. through lower fossil fuel consumption and less CO_2 production).

RCP 8.5, which would increase global mean temperatures to approx. 5 °C above preindustrial levels, is seen as increasingly unlikely. Although the RCP 8.5 is often referred to as the business-as-usual scenario, it's assumptions about a fossil-fuel intensive future without considering any climate mitigation policies are contrary to recent climate policy efforts of most countries (Peters and Hausfather, 2020). Nevertheless, due to unpredictable political circumstances or the uncertainty of climate tipping points and rapidly increasing greenhouse gas levels through feedback loops (e.g. due to thawing Arctic permafrost), this is still a valuable worst-case scenario to consider (Tollefson, 2020; Lenton et al., 2019).

Uncertain simulation of climate change impacts on crops

Next to the uncertain trajectories of future greenhouse gas emissions, uncertainties in simulated climate change impacts on crops also arise from the model structures. Although crop models are usually reliable in simulating measured maize and wheat yields under a range of environments, the robustness of simulated crop response to climate change is often criticised. The simulated relationships between crops and changes in atmospheric CO_2 , temperature and precipitation are critical in climate change impact studies. However, differences in model structure and parametrisation can lead to large differences in the simulated crop response to changing climate parameters.

Prior studies observed an increasing variability between models with increasing temperatures due to the difficulty of simulating high-temperature stress impacts on crops (Asseng et al., 2013; Mearns et al., 1999). Stress for crops due to high temperatures is initiated by enhanced phenology and the

consequent reduction in time for light interception and photosynthesis, which can result in lower crop yields. This process involves complex interaction between leaf area expansion and biomass growth with other processes of the model, which are simulated differently between models (Asseng et al., 2013). In addition, prior studies concluded that the sharp decline of crop yields exposed to extreme temperatures (>30 °C) observed in field experiments is not reflected in crop models (Rötter et al., 2011). Thus, the impact of spatial and temporal changes in temperature extremes due to climate change on crops may not be captured adequately in the model results.

Although precipitation changes are difficult to project and significantly affect simulated crop yields, changes in precipitation have been reported to cause less uncertainty in climate change impact studies than changes in temperature, as the understanding of crop responses to changing precipitation is more robust than knowledge about crop responses to temperature (Lobell and Burke, 2008). Nevertheless, as rainfall is important for annual variability of crop yields, the large disagreement of global climate models on projected precipitation changes plays a crucial part in the uncertainty of climate change impacts. Obviously, robust estimates of water erosion impacts on crops under a changing climate requires improvements in the quality of precipitation projections.

The representation of the impact of increasing atmospheric CO_2 on crops is another major source of uncertainty. Negative climate change impacts may be offset to some extent by increased CO_2 levels, especially for C3 crops, although the magnitude of these effects are uncertain and the subject of much debate (Lobell and Field, 2007). In general, there are many uncertainties in our knowledge of the impact of atmospheric composition on crops (Slingo et al., 2005). In addition, changes in other key climatic parameters, including solar radiation, wind speed, and relative humidity, are difficult to project.

Uncertain simulation of climate change impacts on water erosion

As crop cover influences water erosion indirectly, the above discussed uncertainties of the impact of climate parameters on crops result in uncertainties in the impact of climate change on water erosion.

In addition, uncertainty in precipitation projections, their coarse resolution, and the robustness of the MUSS equation determines the uncertainty of simulated climate change impacts on water erosion directly. As described in Chapter 5 simulated water erosion tends to exceed water erosion measured in field experiments in environments with heavy and abundant precipitation. Thus, the decreasing robustness of simulated water erosion with increasing precipitation may lead to an overestimation of the increase in water erosion where an increase in precipitation amount and intensity due to climate change is projected.

Moreover, the coarse resolution of global climate models hampers the simulation of climate change impacts on erosion. As climate parameters are provided on a daily timescale, important variables to estimate erosion events such as 30-min rainfall intensity are not provided in climate models. Therefore, statistical relationships are used between monthly and annual precipitation and rainfall erosivity to explore erosion changes with climate models (Nearing et al., 2004). As single extreme rainfall events can be responsible for great soil loss, the lack of sub-daily resolution is an important uncertainty in studies analysing climate change impacts on water erosion (Li and Fang, 2016).

Assessing the uncertainties of simulated climate change impacts using model ensembles

To cover the range of model-derived uncertainties in climate change impact studies, the results of model ensembles can be useful. Model ensembles include a range of models that differ in their robustness in representing different variables and/or regions. The range of results from ensembles can be used to assess the certainty of the results by identifying consensus between models or large disagreements. While there are no global ensemble studies that have considered the links between crops and erosion in response to climate change, there are studies that used ensembles of crop or climate models to individually project changes in crop yields or water erosion (Borrelli et al., 2020; Jägermeyr et al., 2021; Panagos et al., 2022). These studies thus provide important insights into the robustness of projected spatially explicit changes in crop yields or erosion.

Studies based on model ensembles have shown that the projection of erosion changes is highly uncertain, as erosion changes strongly depend on precipitation changes (Chapter 5.3.2), which are projected very differently between individual climate models. The limitations of projecting erosion changes due to a low agreement in precipitation changes between models have been discussed in Chapter 7.4.3. The latest study on future changes in erosion driven by rainfall concluded that predicting the magnitude of changes in rainfall erosivity is uncertain, especially at low latitudes where inter-model variability is very high (Panagos et al., 2022).

The projected temperature changes agree better between climate models, as do the simulated impacts of temperature changes on crops. For example, models agree that the negative effects of climate change on maize and wheat yields are greater at low latitudes than at higher latitudes (Jägermeyr et al., 2021). This trend is also in agreement with the crop yield projections presented in this chapter. Building on this commonly found relationship, the results showed how future erosion changes can be attributed to the varying effects of temperature changes on plants at different latitudes.

Despite the difficulties in projecting future erosion changes, exploring the differences of climateplant-soil relationships between global regions can provide insight into how vulnerable different regions are to increasing erosion. However, the existing model uncertainties make it difficult to project spatially explicit erosion changes due to climate change. In future studies, a combination of multiple climate models and crop models that take into account crop-erosion linkages can help identify erosion-prone regions where the ensemble finds both declining yields and increasing rainfall.

7.5 Conclusion

In this chapter, water erosion impacts on global maize and wheat yields were explored in the context of climate change. The results give insight into the complex links between climate change and water erosion and highlight regions where maize and wheat production is most vulnerable to both water erosion and climate change impacts. The simulation results show that maize and wheat production at low latitudes is most negatively affected by climate change and includes the majority of the areas most vulnerable to water erosion under both analysed climate change scenarios. Moreover, water erosion in maize and

wheat fields is likely to increase due to diminishing vegetation cover at low latitudes, where substantial negative impacts of climate change on crops are expected.

The model outputs reflect the influence of geographic variations of environmental characteristics and field management on climate change and water erosion impacts. Low-latitude regions combine high precipitation erosivity and high temperatures; thus, they are most susceptible to both water erosion and climate change impacts on maize and wheat. In addition, the different impacts of climate change and water erosion across regions are determined by the varying levels of fertiliser application and irrigation. Thus, the simulation results illustrate the contrast between the effects of climate change and water erosion on crops between highly developed agricultural regions and regions with abundant smallholders with typically low volumes and efficiency of inputs. The main determinants of adverse effects from climate change and water erosion are found in sub-Saharan Africa, South and Southeast Asia, and Central and South America.

Whilst the relationships between climate change and water erosion are highly complex, there is a clear positive relationship between increasing precipitation and water erosion. The magnitude of increasing water erosion rates with increasing precipitation significantly depends on crops, warming levels and field management. The simulation results demonstrate that the impact of increasing precipitation on water erosion will be stronger with higher levels of climate change. Negative impacts of warming on crop yields play an important role in increasing water erosion rates with climate change. Thus, climate resilience of crops will be critical to the impact of climate change on water erosion on cropland. Moreover, the model outputs illustrate that field management can reduce water erosion impacts on crop production more than different levels of climate change.

Although identifying future hotspots of water erosion changes is difficult due to the uncertainty in precipitation projections and future land use, the model results show that climate change impacts on crops can indicate where crop production is likely to become more vulnerable to water erosion. In addition, the model results demonstrates the value of field management in controlling future impacts of water erosion. The adaptation to both climate change and water erosion is likely to become increasingly important in the most vulnerable regions.

Chapter 8

Overall summary and future research needs

8.1 Summary of insights

Water erosion poses a serious threat to soil resources as it removes topsoil, affecting soil fertility and other soil properties important for plant growth. Therefore, water erosion is an important process to consider when assessing the productivity and environmental impact of crop production. In addition, climate change influences water erosion through the various links between water erosion, vegetation and the climate system.

To understand the state of agriculture today and in the future, it is important to explore how water erosion affects crop yields and how climate change might influence this. To analyse this on policyrelevant scales, large-scale or global assessments are necessary. Models can account for the effects of geographical variations in land use and environmental characteristics on water erosion using a consistent methodology over large scales.

However, large-scale and global agricultural modelling studies need to be improved in terms of robustness and representation of plant-soil interactions. Water erosion is an often neglected process in agricultural assessments. Therefore, there is no consistent information on the impact of water erosion on agricultural productivity on a large and global scale. In addition, the links between water erosion, vegetation and the climate system in global agricultural assessments need to be explored in more detail to provide a more complete picture of the impacts of climate change on agriculture.

This study addressed the need for large-scale indicators on water erosion rates and impacts using the GGCM EPIC-IIASA. The robustness of the model with respect to simulated water erosion in different global regions was evaluated using field data. Subsequently, EPIC-IIASA was used to estimate the impact of water erosion on global maize and wheat yields. The combined impact of climate change and water erosion on maize and wheat yields, and the potential impact of climate change on water erosion, were also assessed. This section summarises the findings of the thesis.

8.1.1 Simulation and evaluation of water erosion with EPIC-IIASA

EPIC-IIASA was used to develop a modelling approach to estimate long-term average water erosion rates in global maize and wheat fields under different field management scenario (Chapter 5). Simulated water erosion rates differed greatly between temperate climate regions where crops are grown in lowlands and tropical regions with abundant hilly farmland. Water erosion rates in maize fields were higher than in wheat fields. In addition, model results showed that water erosion rates decrease with decreasing tillage intensity and increasing crop residue cover in many regions of the world.

The robustness of the simulated water erosion rates was confirmed for the major maize and wheat growing regions by using water erosion measurements from field studies as a comparison. In regions with steep slopes and heavy precipitation, simulated water erosion rates exceeded field values, reducing confidence in model results at these locations. However, these sites accounted for only 7% of the world's maize and wheat land.

One of the most important challenge to the evaluation of water erosion was the limited amount of suitable field data in different regions. In addition, field data were highly variable, as assessing water erosion based on field experiments can be difficult and subject to many uncertainties (Chapter 2.3.5).

To better interpret the robustness of simulated water erosion estimates, an analysis of the sensitivity and uncertainty of water erosion estimates was conducted in Chapter 5. The sensitivity analysis highlighted the high sensitivity of the model to precipitation and slope steepness. The uncertainty analysis showed a wide range of water erosion estimates with different water erosion equations and field management scenarios in regions vulnerable to water erosion. Improving the limited information on global field management and the spatial resolution of topography and land use data would improve the robustness of water erosion estimates.

The insights on the varying robustness and uncertainty of simulated water erosion estimates in Chapter 5 are useful to interpret the robustness and uncertainty of the results in Chapter 6 and Chapter 7.

8.1.2 The impact of water erosion on global maize and wheat yields

The modelling approach evaluated in Chapter 5 was used in Chapter 6 to explore the impact of water erosion on maize and wheat yields and production. In addition, the model results were used to identify the fields most at risk of water erosion based on the amount of rainfall, the slope of the fields and the use of fertilisers.

Model results showed that water erosion will reduce maize and wheat yields on about half of the world's cropland by an estimated 3% per year if farmers do not adapt to changing environmental conditions. The areas at risk are located on steep slopes exposed to heavy precipitation and where little fertiliser is available. Fields with these characteristics are found primarily in parts of sub-Saharan Africa, South Asia, and Latin America. Because the main areas of maize and wheat cultivation are in flat terrain outside the tropical heavy rainfall areas and are managed with large amounts of inputs such as fertilisers, they are less affected by water erosion. The model results showed that annual production losses of maize and wheat due to water erosion are less than 1% of global production volume. In addition, the calculated replacement cost of nitrogen fertiliser to offset nutrient runoff was low compared to crop production value. However, the analysis of water erosion impacts did not include the cost of replacing other plant nutrients (phosphorus, potassium, carbon), and external costs such as pollution from sediment runoff.

8.1.3 The impact of climate change and water erosion on maize and wheat yields

Chapter 7 extends the analysis of water erosion impacts on maize and wheat yields by using the modelling approach evaluated in Chapter 5 under two contrasting climate change scenarios. In addition to analysing regional differences in water erosion impacts as in Chapter 6, regional differences were also analysed for climate change impacts on maize and wheat yields.

The model results showed that low-latitude maize and wheat production are most affected by water erosion and climate change simultaneously. Climatic conditions at low latitudes include heavy precipitation, resulting in increased water erosion, and higher temperatures, which exacerbate the negative effects of warming on many crops. In addition, global differences in water erosion and climate change impacts are due to differences in agricultural intensity and associated differences in the quantity and efficiency of inputs such as fertilisers and irrigation. Regions most negatively affected by both climate change and water erosion are most abundant in sub-Saharan Africa, South and Southeast Asia, and Central and South America.

The model results were further analysed to explore the links between climate change and water erosion. There was a clear positive relationship between increasing precipitation and water erosion. The impact of increasing precipitation on water erosion was stronger with higher levels of climate change. The erosion-precipitation relationship was strongly affected by field management. In addition, the model results showed that water erosion likely increases where climate change reduces crop yields.

However, identifying regions where water erosion will change in the future has been difficult. This is primarily because changes in precipitation are most important, but these projections are highly uncertain. Nevertheless, the likelihood of a decline in vegetation cover due to climate change may indicate where water erosion could increase. Since the greatest negative impacts of climate change on crops worldwide are in low latitudes, these regions are also most at risk from increased water erosion due to decreasing vegetation cover.

Finally, the model results showed the strong influence of field management on the effects of climate change on water erosion. This indicates a huge potential for mitigating the negative effects of climate change on water erosion and thus crop yields.

8.2 Summary of research contributions

The results of this thesis contribute to a better understanding of regional differences in water erosion on global croplands and the effects of water erosion and climate change on global cereal production; thus, providing new insights into the impact of environmental changes on long-term agronomic trends. Specifically, this thesis addresses i) the need for improved large-scale indicators of water erosion rates and water erosion impacts on crops to inform major environmental and agricultural policies (Alewell et al., 2019; Montanarella, 2007; Nkonya et al., 2011), and ii) the neglect of land degradation processes in climate change impact assessments using GGCMs (Balkovič et al., 2018; Webb et al., 2017).

8.2.1 The first evaluation of global water erosion estimates from a GGCM

Proper model calibration is needed to match experimental data obtained from the field, foremost for a complex process such as soil loss due to water erosion (Favis-Mortlock, 1998). However, the evaluation of global water erosion estimates was lacking in previous global studies.

Whilst previous global erosion estimates lack evaluation, previous global crop yield estimates were evaluated against regional yield statistics to represent the crop yield patterns across regions and countries (Mueller et al., 2017). Following the same paradigm, this study evaluated the robustness of a GGCM to represent regional differences and spatial patterns in water erosion estimates based on the most important determinants for water erosion. Although the field data used in this study is insufficient to calibrate EPIC-IIASA to simulate actual erosion events observed in the field, it provides an initial indication of the robustness of estimated water erosion rates under certain topographical and climatic conditions.

As such, the evaluation against field data confirmed that EPIC-IIASA can be used to analyse global differences in water erosion rates and impacts between the most important crop production regions. Therefore, the results of Chapter 5 confirm the usefulness of EPIC-IIASA to address the previously neglected impact of water erosion in large-scale crop modelling studies and the previously neglected interactions between crops and water erosion in large-scale water erosion modelling studies.

8.2.2 New large-scale water erosion indicators

Large-scale indicators are necessary to address water erosion in policy programs (Alewell et al., 2019). To create the needed indicators simple models are required due to the limited availability of global input and experimental data. Therefore, most authors have recently opted for USLE-based models to estimate water erosion rates at large scales (Chapter 2.3.1). Since these models generally use static inputs, biomass-soil interactions in response to weather and field management have been neglected.

Unlike previous studies, the large-scale water erosion estimates in this study are based on USLEtype models combined with the process-based crop model EPIC. The more complex modelling approach allowed to simulate the effects of field management strategies on water erosion. As such, modelled global water erosion rates account for the first time for surface cover of individual crops, timing of planting and harvest, irrigation, fertilisation, cover crops, intercropping, tillage, and residue management.

The model results thus provide new insights into the variability of water erosion rates based on these management factors at large spatial scales and improve the availability of large-scale indicators for important environmental and agricultural policies.

8.2.3 New evidence on regional differences in the impact of water erosion in agriculture

Large-scale water erosion studies have focused primarily on the extent of erosion rather than the impact on crops. As a result, the large-scale impacts of water erosion have not been adequately quantified in physical and economic terms. In addition, the limited number of large-scale studies on the effects of water erosion on crops relied on field experiments based on different methodologies leading to different estimates, and they do not cover all agricultural areas worldwide (Chapter 2.4). Previous approaches are less suitable to extrapolate erosion-productivity relationships beyond field scale to compare regional differences and to derive robust conclusions on water erosion impacts on different crops on large and global scales.

The results in Chapter 6 provide the first globally consistent estimates of water erosion impacts on crops using state-of-the-art datasets on climate, soil, topography and management practices. These results provide the most accurate global assessment of the links between water erosion and crop productivity that is currently feasible, given data availability.

The globally consistent modelling method made it possible to illustrate the sharp contrast between the low on-site costs of erosion in the most productive agricultural regions and the significant impact of water erosion on crop yields in regions with heavy rainfall, hilly croplands, and low fertiliser use. By identifying agricultural regions most vulnerable to water erosion, the model results can highlight hotspots where anti-erosion measures are most needed to sustain crop productivity in the long-term.

In addition, the model results allow to question the globally uniform assumption by previous authors that annual water erosion rates up to $11 \ tha^{-1}$ are sustainable for crop production (Sartori et al., 2019; Pravalie et al., 2021). Whilst these rates may be sustainable to maintain agricultural productivity in high-input systems, the results in Chapter 6 demonstrates that lower water erosion rates can have a significant impact on crops in regions where low fertiliser use is a common limiting factor for crop yields.

Overall, the globally consistent modeling approach developed in this study provides a platform for assessing water erosion problems in multiple ways to identify risks and hotspots and support policy actions. For example, shifts in future risks and hotspots can be assessed by integrating models on land use change (e.g. new crops) and climate change. The latter has been explored in Chapter 7. In addition, the field management scenarios used for generating all model outputs demonstrate the decisive impact of farming techniques and soil conservation measures on water erosion rates and water erosion impacts on crop yields. Previous studies point out that there may be a yield penalty from employing soil conservation practices in the short-term and a yield gain in the long run (Giller et al., 2009). The model approach developed here can be used to create large-scale indicators on the trade-offs between lower water erosion rates due to soil conservation strategies and absolute crop yields to inform policies. In addition, the model outputs estimate the amount of organic nitrogen and carbon in sediment runoff that can potentially enter surrounding environments. These estimates can be used to assess off-site costs for different field management scenarios. Previous studies suggested that off-site impacts are the main concern of erosion in the highest developed agricultural regions and thus they are targeted by many policy programs (Graves

et al., 2015; Montanarella et al., 2016).

8.2.4 Improvements in crop yield projections by considering water erosion

Future changes in crop yields are discussed mainly as a result of climate change, while many important processes that contribute to ongoing soil degradation are neglected (Chapter 3.6). This can lead to inconsistencies in the validity of model results in a world with different management intensities, as the influence of weather variability on yield variability decreases with decreasing inputs to farming systems, while other factors such as soil quality become more important (Ray et al., 2015). Because water erosion is one of the most common soil degradation process affecting crop yields (Chapter 2.2) and is closely related to climate (Chapter 2.5), it should be especially considered in crop yield projections.

Model results in Chapter 7 showed that water erosion affects projected crop yields at vulnerable sites characterised primarily by heavy rainfall, steep slopes, and low fertiliser use. Without accounting for water erosion, crop yield projections at these sites ignored an important factor that strongly influences crop productivity, which can lead to overestimation of future crop yields. In addition, model results showed that the susceptibility of sites to water erosion may change in the future due to the effects of climate change on vegetation cover and precipitation patterns. By considering water erosion and its link to climate change, large regions in the lower latitudes have been identified as being at risk of crop yield losses due to the simultaneous effects of soil erosion and climate change.

Moreover, consideration of water erosion provides greater insight into the effects of different field management scenarios on crop yields, which is often the subject of climate change impact studies (Chapter 3.5). The analysis of model results has shown that water erosion affects the links between field management and crops in several ways. These include the impact of water erosion on fertiliser and nutrient runoff, which affects climate change adaptation strategies that aim to increase fertiliser use (Rosenzweig et al., 2014; Balkovič et al., 2014) or adopt cropping practices that replenish soil nutrients (Folberth et al., 2014). In addition, changes in soil quality due to water erosion affect crop sensitivity to climate change (Laban et al., 2018; Herrick et al., 2013).

By including water erosion in the projection of global maize and wheat yields, the model results have taken into account another important factor for future changes in crop yields in many regions, in addition to climate change. This contributes to a more complete picture of the stressors affecting future cereal production trends. In addition, this improve the usefulness of climate change adaptation studies by accounting for an important process within the interactions between field management and crops.

8.2.5 New insights into the potential impacts of climate change on global water erosion

The effects of climate change on water erosion are the result of complex interactions of many factors including soil, topography, agricultural practices, biomass, and climate. As these factors vary widely around the world, the effects of climate change on water erosion can vary widely across regions. Because previous studies of the effects of climate change on water erosion focused on individual fields in various

regions, they reached very different conclusions (Chapter 2.5). Due to the different modelling approaches used in previous studies, their results cannot be used to compare the effects of climate change on water erosion between regions.

The analysis of model results in Chapter 7 showed the interactions in different global regions based on the impact of varying precipitation changes and vegetation cover changes. While uncertainties in precipitation projections render them unsuitable for examining regional differences, the higher confidence in regional differences in future vegetation cover changes could provide insight into potential global differences in climate change impacts on water erosion.

The globally consistent modelling approach developed in this thesis enabled to analyse the interactions between crops and water erosion in response to climate change for the first time at the global scale. As such, the results of Chapter 7 contribute to a more comprehensive understanding of the potential impact of climate change on water erosion in different global regions.

8.2.6 New data on global water erosion rates and impacts on crops

Finally, this study provides novel data including global water erosion estimates for different farming techniques and crops, and estimates of water erosion impacts on global maize and wheat yields under historic climate and two contrasting climate change scenarios. All estimates are available on grid, national, regional and global scale.

8.3 Research needs

8.3.1 Most important needs to improve large-scale water erosion estimates

To improve the evaluation of simulated water erosion rates, the amount of appropriate field data needs to be increased to represent a wider range of global environments and crop management practices. A major challenge in this study was the collection of useful field data, which is often only published in aggregated values and needed to be collected from numerous publications, grey literature and conference proceedings. Moreover, field data is often not freely available (Benaud et al., 2020).

In addition, more data on field management would improve global water erosion estimates. Whilst information on rainfed and irrigated cropland, crop calendars and crop-specific fertiliser applications rates are based on spatially explicit datasets, the geographic variations in tillage intensity and soil conservation measures are based on rough assumptions in this study. Although they provide insights on the impact of changing surface cover and roughness on water erosion rates, they hardly capture the actual regional differences in farming techniques.

Also, a higher spatial resolution is needed for land use data to better approximate actual water erosion rates and impacts on crops. Improvements in spatial data informing about the distribution of cropland in mountainous regions is particularly important. Previous field studies identified highest water erosion rates on marginal cropland and on steep slopes, which also have been targeted by policy measures to reduce erosion (Long et al., 2006b; Deng et al., 2012). However, the evaluation against field

data shows that water erosion on fields with steep slopes and strong precipitation is overestimated. The low confidence in simulated water erosion rates in mountainous regions is partly the result of the heterogeneity of upland farming and fields and the missing data on field management, anti-erosion measures and the low spatial resolution on cropland distribution and topography datasets. In addition, previous studies suggested that model improvements are needed to improve erosion estimates for locations with steep slopes and strong precipitation (Naipal et al., 2018; Labrière et al., 2015).

Due to the current uncertainties, global studies cannot replace field-scale assessments based on precise information on management practices and site characteristics. Due to higher spatial detail, field-scale assessments can be based on more complex water erosion models, which may include special elements such as channels and ponds to identify potential sources and sinks of sediments and associated nutrients within a field (Jetten et al., 2003). By including depositional areas within the spatial unit studied, positive effects of topsoil accumulation on crop productivity can be considered (Bakker et al., 2007). In addition, studies based on data with a higher temporal resolution can consider the impact of individual rainfall events on sediment runoff instead of focusing on average erosion rates as is common in global studies. In other words, smaller-scale studies can more precisely inform about actual water erosion impacts on a field to support effective anti-erosion measures on-site.

However, studies on erosion-productivity relationships cannot normally be scaled-up as the robustness of locally observed relationships need to be re-evaluated for different environmental and socioeconomic conditions in each location. Given the current lack of consistent field studies representing all global environments, a bottom-up approach to deliver large-scale indicators on erosion rates and impacts to inform agricultural and environmental policy programs is not currently feasible (Alewell et al., 2019).

Future global studies on water erosion impacts may benefit from current efforts to compile spatial data on representative management practices such as tillage systems (Porwollik et al., 2019), and remote sensing products for spatial attribution of field management practices (Hively et al., 2018; Zheng et al., 2014). In addition, the increasing availability of high-resolution data through improvements in remote sensing techniques will benefit future global water erosion assessments (Buchhorn et al., 2020). Moreover, recent efforts to collate erosion measurements and metadata from existing studies in a single platform may improve the global coverage of appropriate field data in the future (Benaud et al., 2020). In addition to the need for more spatial data on representative management practices and higher-resolution datasets on land use patterns and topography, a more consistent approach to field-based data collection to evaluate model outputs would enable such studies to be used in future large-scale water erosion assessments.

8.3.2 Scenarios on future land use and field management changes are needed to estimate future water erosion rates

The main uncertainties in simulating the effects of climate change on water erosion were discussed in Chapter 7. Among them, uncertainties in precipitation projections and simulated changes in crop cover due to climate change must be considered to improve estimates of future changes in water erosion rates and their regional variations.

However, numerous studies concluded that future water erosion rates may be primarily influenced by land use change instead of climate change (Li and Fang, 2016). This poses a major challenge for improving research on future water erosion rates because land use changes are determined not only by climate, but also by many other factors such as population growth, socioeconomic changes, and policies. These factors are often difficult to project and can affect land use in ways that both increase and decrease water erosion.

For example, projected population growth and the associated increase in food demand in many developing countries may exacerbate water erosion on cropland as agriculture expands and intensifies. On the other hand, some projections of future agricultural intensification suggest that the impact of water erosion on crops may be reduced in the future as fertiliser consumption and fertiliser-use efficiency are projected to rise (FAO, 2006). Whilst this could lead to increased off-site impacts of water erosion, the impact of water erosion on crops may be less relevant. Moreover, high rates of urbanisation and rural-urban migration are predicted in many developing countries, which could reduce degradation stemming from the cultivation of marginal lands and slopes (FAO, 2006). Examples in Europe from the 1950s onwards and in China more recently showed that urbanisation in conjunction with agricultural policy programs led to the abandonment of steep slopes and marginal land, which reverted to forest and caused a reduction in water erosion (Deng et al., 2012; García-Ruiz, 2010; Renwick et al., 2013; Tasser et al., 2007). However, this development may not occur everywhere. In Indonesia, where the rate of population growth is high, population drift to cities has not significantly reduced the density of rural settlement (FAO, 2006). Evidence from Southeast Asia show that increasing pressure on food production recently led to the expansion of cropland to steep slopes (Turkelboom et al., 2008; Valentin et al., 2008).

In addition, it is important to consider future changes in field management to project changes in water erosion. For example, the cultivation of crops that are more resilient to climate change may increase in certain areas in the future, reducing the indirect negative effects of climate change on water erosion. Millet is less susceptible to warming than maize and wheat and thus may be more common in arid regions in the future (Liu et al., 2008). Moreover, current efforts to shift to more sustainable land management may decrease future water erosion. Several policies encouraging sustainable management practices that reduce soil erosion, promote vegetation growth, enhance soil organic carbon and halt land degradation are already in place on all continents (excluding Antarctica) (Smith et al., 2016). The *World Overview of Conservation Approaches and Technologies* (WOCAT) provides an overview of sustainable measures and has been recently endorsed by the UN (Liniger et al., 2019). In addition, the UN promotes halting of land degradation and land restoration through several initiatives such as their SDGs (Keesstra et al., 2018).

In summary, to improve projections of water erosion changes, various field management and land use scenarios must be considered in addition to climate change scenarios. Future research on modelling water erosion changes can use existing global land suitability change maps to estimate land use changes due to climate change (Zabel et al., 2014). Land use change scenarios based on other factors such as population growth, policies and socio-economic changes are likely to be made easier on a smaller scale because of their complexity. Scenarios about future changes in field management may also be easier to create on a smaller scale. In exploring future scenarios for field management, it is possible to draw on existing guidelines that have been endorsed by the UN and are likely to be used more frequently in the future.

8.3.3 The effects of other land degradation processes must be considered

Next to erosion, specific threats to soil function are compaction, acidification, contamination, sealing, salinisation, waterlogging, nutrient imbalance (i.e., both nutrient deficiency and nutrient excess), and losses of soil organic carbon and of biodiversity (Montanarella et al., 2016). Similar to water erosion, the impact of other land degradation process may be low on a global scale but can be high in specific areas (Pravalie et al., 2021). Due to uneven distribution of dominant land degradation processes globally, it is important that future research addresses the impact of all land degradation processes on crops to better assess long-term agronomic trends in different global regions.

Land degradation processes other than water erosion may become more important in certain regions. For example, agricultural intensification is projected to lead to an expansion in irrigated land in developing countries, which likely does not increase water erosion as irrigated land is generally on flat or terraced land. However, other land degradation processes such as salinisation may increase as a consequent, especially on arid land (Norse et al., 2001). In addition, climate change may increase the severity of other land degradation processes through changes in soil organic matter and soil water balances due to warming and droughts (Olsson et al., 2019).

Several authors state that more policies are required to fully consider the risks associated with ongoing land degradation, or opportunities arising from restoration of degraded land (Webb et al., 2017; Pravalie et al., 2021). Soil degradation does not occur because of any lack of knowledge on how to protect soils, but a lack in policy governance (Alewell et al., 2019; Montanarella et al., 2016; Panagos et al., 2016). Additional data on global patterns of land degradation and its impacts on agriculture and the environment will help identify the most vulnerable regions. By identifying the regions most vulnerable to land degradation worldwide, as has been done with water erosion in this thesis, more knowledge of land degradation hotspots can be created to know where control measures are most urgent.

8.4 Final conclusion

Knowledge on future pathways of crop production and their environmental impacts is important to predict potential threats to human well-being and the sustainability of environmental resources. However, many relationships between biomass, soil and climate in agricultural systems are still not understood. By developing a modelling approach to explore global water erosion patterns and impacts on maize and wheat yields in response to historic climate and climate change scenarios, this thesis contributes to an improved understanding of the links between crop production and environmental change and their role in long-term agronomic trends. This thesis improves global water erosion assessments by analysing both water erosion patterns and water erosion impacts. In addition, this thesis contributes to the further development of global crop yield projections by addressing the need to integrate land degradation processes such as water erosion into large-scale and global impact assessments on climate change in the agricultural sector. As such, this research improves large-scale indicators to inform environmental and agricultural policy programs on land degradation and agricultural productivity.

EPIC-IIASA was previously considered to be a reliable tool for depicting the different productivities between global agricultural regions. This study shows it can also adequately estimate global water erosion patterns for specific crops and interactions between soil erosion and productivity. Although improvements are needed to increase the robustness of estimated water erosion rates in areas most sensitive to water erosion, the model results can illustrate potential impacts of erosion on global crop production and identify soil erosion hotspots on global cropland.

The new assessment of water erosion impacts on global maize and wheat yields illustrates the strong contrast of water erosion impacts between different global regions, which is caused by different environmental conditions and agricultural management. Maize and wheat cultivation areas most vulnerable to water erosion are concentrated in Latin America, sub-Saharan Africa and South and Southeast Asia. In addition, regions in low latitudes where maize and wheat is grown, are most likely to suffer from increased water erosion impacts in the future due to the negative impacts of climate change on both crops. The combined effects of water erosion and climate change require farmers to mitigate both stressors, as land degradation caused by water erosion can reduce the resilience of agricultural systems to climate change and, conversely, climate change can increase water erosion and thus land degradation. In contrast to potentially significant global changes in crop production induced by climate change, the severe impacts of water erosion on maize and wheat yields are likely to be concentrated in specific regions. Despite considerable uncertainties, the projections indicated that the impact of water erosion on maize and wheat production will remain small on a global scale, since the cultivation areas most affected by water erosion are currently outside the large maize and wheat production regions. The main difficulties in predicting the effects of water erosion are the uncertainty in projected changes in precipitation and future land use changes.

Simulated water erosion rates and impacts are strongly driven by land use and field management assumptions, which can only partly be narrowed down at a global scale. To improve the robustness of future global studies on water erosion patterns and impacts, more spatial data on representative management practices and higher-resolution datasets on land use patterns and topography are needed. In addition, a well-defined field-based data collection to evaluate model outputs will be critical to improve future large-scale assessments of water erosion.

This study demonstrated that an integrated biophysical modelling framework - considering plant growth, soil processes and input requirements - can provide a link between land degradation and global agricultural assessments. The modelling approach developed in this study thus provides a platform for expanding research on the impacts of soil degradation and climate change on global crop production. Whilst maize and wheat are the most important staple crops affected by water erosion, focusing on just two crops is not sufficient to explore the threats of environmental change to future crop production. Thus, future research should focus on analysing the susceptibility and resilience of more crops to environmental change according to the impact assessment developed in this thesis. In addition, the modelling framework of this study can be used to incorporate additional climate change scenarios to capture the full spectrum of potential climate impacts on future crop yields and associated changes in water erosion. As global crop yield projections should ideally take into account all stress factors that affect crop yields, future research should also consider other land degradation processes. This study has shown that large-scale land degradation assessments can capture the regionally varying impacts on crop yields and identify priority areas for supportive management and policy interventions.

Appendix A

Chapter 4 - Methodology



Figure A.1: Global dataset of slope classes (%) that are used to derive the slope steepness per grid cell (IIASA/FAO, 2012).



Figure A.2: Nitrogen (N), Phosphorus (P) and Potassium (K) application rates (kg $ha^{-1} yr^{-1}$) for maize (a) and wheat (b) on global cropland (Mueller et al., 2012).



Figure A.3: Grid cells with irrigated and rainfed wheat and maize cultivation around the year 2000 (Portmann et al., 2010).



Figure A.4: Main climate zones based on the updated Koeppen-Geiger climate classification (Peel et al., 2007)



Figure A.5: Relative area covered by the dominant slope and cropland per grid cell.



Figure A.6: Locations of water erosion field data from cropland where coordinates were recorded (n=554).



Figure A.7: Distribution of erosion values (t ha⁻¹) measured in agricultural fields using ¹³⁷CS method (n = 315, Mean = 24 t ha⁻¹; Median = 18 t ha⁻¹), runoff and sediment collection (n = 188, Mean = 21 t ha⁻¹; Median = 4 t ha⁻¹) and volumetric surveys (n = 103, Mean = 2 t ha⁻¹; Median = 0.1 t ha⁻¹).



Figure A.8: (a) Distribution of slope steepness (%) for measured erosion values (n = 606; Mean = 16%; Median = 11%). (b) Distribution of annual precipitation (mm) for measured erosion values (n = 606; Mean = 879 mm; Median = 774 mm). (c) Distribution of recorded measurement periods for soil loss experiments excluding radioisotopic methods (n = 95; Mean = 15 a; Median = 1 a).



Figure A.9: Number of measured water erosion records (n=606) per country (n=36).

EPIC Parameter	Description	Distribution	Range
CHL	Channel length	Triangular	0.2; 0.8; 1.3
CHN	Manning N channel	Triangular	0.02; 0.1; 0.3
CHS	Channel slope	Triangular	0; 0.05; 0.3
EMX	Mixing efficiency	Triangular	0.2; 0.5; 0.8
HSG	Soil hydrologic group	Discrete	1; 2; 3; 4
LUN	Land use number	Discrete	1; 2; 3; 4; 5; 6; 7; 8; 9; 10; 11;
			12; 13
OC	Organic carbon concentration	Triangular	0.001; 0.02; 0.1
OPV5	Plant population at planting	Triangular	50; 150; 200
ORHI	Crop residues left after harvest	Discrete	0; 0.25; 0.5; 0.75; 1
P23	Coefficient in RUSLE C factor	Triangular	0.3; 0.5; 1.5
	equation		
P26	Coefficient in RUSLE C factor	Triangular	0.01; 0.05; 0.2
	equation		
P42	Curve number (CN) coeffi-	Triangular	0.5; 1; 1.5
	cient		
P65	CN coefficient	Triangular	0.05; 0.25; 0.5
P73	CN coefficient	Triangular	1;1.5;2
P75	CN coefficient	Triangular	0; 0.05; 0.3
PRCP	Daily rainfall	Triangular	0.5; 1; 1.5
RIN	Ridge interval	Discrete	0.2; 0.4; 0.6; 0.8
ROK	Coarse fragment content	Triangular	0; 0.05; 0.5
RR	Random surface roughness	Triangular	10; 20; 50
S18	Soil temperature effect	Triangular	0.97; 1; 1.02
S23	Fraction plant ground cover	Triangular	0.5; 1; 1.5
S301	CN2 as % of the volume be-	Triangular	10; 20; 70
	tween FC and WP		
S302	CN3 as % of the volume be-	Triangular	10; 20; 70
	tween saturation and FC		
SAND	Sand content	Triangular	0.1; 0.5; 0.8
SILT	Silt content	Triangular	0.1; 0.3; 0.8
SLP	Slope steepness	Triangular	0; 0.05; 0.3
SLPL	Slope length	Triangular	100; 120; 200
SN	Surface N channel	Triangular	0.02; 0.1; 0.3
TMX	Air temperature max	Triangular	-4; 0; 4
WSX	Watershed area	Triangular	10; 50; 200

Table A.1: Input parameters directly connected to the water erosion equations in EPIC. Range of discrete values or values used to define the triangular distribution to generate random values for each parameter for the sensitivity analysis.

Appendix B

Chapter 5 - Uncertainties, sensitivities and robustness of simulated water erosion with EPIC-IIASA



Figure B.1: World regions classified according to the United Nations geo-schema (UN, 1999) with minor modifications: Melanesia has been added to Southeast Asia and the Caribbean has been added to Central America.



Figure B.2: Median deviation (MD) in t ha^{-1} between measured and simulated water erosion using the baseline scenario with different water erosion equations. Measured and simulated medians were calculated for different slope and precipitation classes.


Figure B.3: Distribution of average water erosion values from 1980 - 2010 simulated with the baseline scenario and weighted for each grid cell. The dashed vertical line illustrates the median of the distribution, which represents global median water erosion of 6 t ha⁻¹ a⁻¹. Average water erosion at each grid and the global average water erosion of 19 t ha⁻¹ a⁻¹ has been calculated as a weighted average based on the distribution of irrigated and rainfed maize and wheat acreage (Portmann et al., 2010).



Figure B.4: Comparison of slope inputs and simulated water erosion outputs between the cropland distribution scenario using the most common slopes and the cropland distribution scenario using the flattest terrain available in Italy. (a, b) distribution of the cropland share (Portmann et al., 2010) per slope class. (c, d) distribution of grid cells per slope class. (e) Simulated water erosion for Italy using both cropland distribution scenarios. Midlines visualise median values, boxes include values from the 25th to the 75th percentiles and whiskers bracket values between the 10th and the 90th percentiles.

Appendix C

Chapter 6 - The impact of water erosion on global maize and wheat yields



Figure C.1: Maize (a) and wheat (b) yields (t $ha^{-1} yr^{-1}$) for each grid cell simulated with the baseline scenario with and without the impact of water erosion, averaged for the years 2001-2010 and weighted for irrigated and rainfed areas.



Figure C.2: Maize yield losses due to water erosion (% yr⁻¹) for all countries with significant crop yield losses due to water erosion. Countries contributing less than 0.01% to global maize production are excluded. The countries are ranked by median crop yield losses. Boxes include values from the 25^{th} to the 75^{th} percentiles and whiskers bracket values between the 10^{th} and the 90^{th} percentiles. The points illustrate minimum and maximum median crop yield losses generated from all field management scenarios. All values are converted to logarithmic scale. Countries without significant differences between average yields simulated with and without water erosion are excluded (T-Test, p>0.05). Grey barplots illustrate the share of grid cells affected by erosion impacts in each country, and errorbars indicate the variability of affected grid cells due to all management scenarios.



Figure C.3: Wheat yield losses due to water erosion (% yr⁻¹) for all countries with significant crop yield losses due to water erosion. Countries contributing less than 0.01% to global maize production are excluded. The countries are ranked by median crop yield losses. Boxes include values from the 25^{th} to the 75^{th} percentiles and whiskers bracket values between the 10^{th} and the 90^{th} percentiles. The points illustrate minimum and maximum median crop yield losses generated from all field management scenarios. All values are converted to logarithmic scale. Countries without significant differences between average yields simulated with and without water erosion are excluded (T-Test, p>0.05). Grey barplots illustrate the share of grid cells affected by erosion impacts in each country, and errorbars indicate the variability of affected grid cells due to all management scenarios.



Figure C.4: Yield loss due to water erosion plotted against Nitrogen and Phosphorus fertiliser use per grid cell. The fertiliser application rates are taken from (Mueller et al., 2012), which have been used to prepare the model inputs. The linear relationship between crop yield loss and fertiliser application rates is illustrated by the dashed regression line.



Figure C.5: Soil organic nitrogen loss with sediment runoff from the topsoil layer (kg ha⁻¹ yr⁻¹) simulated for each grid cell in maize fields (a) and wheat fields (b) averaged for management scenarios and the simulation period 1980-2010 and weighted for irrigated and rainfed areas.



Figure C.6: Soil organic carbon loss with sediment runoff from the topsoil layer (kg ha⁻¹ yr⁻¹) simulated for each grid cell in maize fields (a) and wheat fields (b) averaged for management scenarios and the simulation period 1980-2010 and weighted for irrigated and rainfed areas.

Appendix D

Chapter 7 - The combined effects of climate change and water erosion on global maize and wheat yields



Figure D.1: Relative changes in mean maize yields for the years 2050-2059 and 2090-2099 simulated with RCP 2.6 and RCP 8.5 compared to mean maize yields of the years 1980-2010 simulated with the AGMERRA climate data.



Figure D.2: Relative changes in mean wheat yields for the years 2050-2059 and 2090-2099 simulated with RCP 2.6 and RCP 8.5 compared to mean wheat yields of the years 1980-2010 simulated with the AGMERRA climate data.



Figure D.3: Regional trends in maize yields 1980 - 2099 expressed as the three-year rolling mean based on annual average maize yields simulated with the AGMERRA climate data (1980-2010) and the RCP 2.6 and RCP 8.5 (2005-2099). Annual average maize yields are weighted with the rainfed and irrigated area of the respective region.



Figure D.4: Regional trends in wheat yields 1980 - 2099 expressed as the three-year rolling mean based on annual average maize yields simulated with the AGMERRA climate data (1980-2010) and the RCP 2.6 and RCP 8.5 (2005-2099). Annual average wheat yields are weighted with the rainfed and irrigated area of the respective region.



Figure D.5: Regional trends in the relative decrease in maize and wheat yields due to water erosion, as well as share of grid cells where water erosion reduces crop yields for 2010-2099 simulated with RCP 2.6 and RCP 8.5.



Figure D.6: Maize and wheat yield reduction due to water erosion for 2090-2099 simulated with different field management scenarios per world region. The boxes represent median values and 25^{th} and 75^{th} percentiles. Each box per region represents a different field management scenario, including the baseline scenario and all combinations with the cover crop (cover, mono) and the tillage management scenarios (conventional (C), reduced (R), no-tillage (N)).



Figure D.7: Grid cells where maize and wheat yields are reduced due to water erosion per field management scenario for the years 2090-2099. The field management scenarios include the baseline scenario and all combinations with the cover crop (cover, mono) and the tillage management scenarios (conventional (C), reduced (R), no-tillage (N)).



Figure D.8: Regional trends in water erosion rates in maize and wheat fields between 2010-2099 simulated with the RCP 2.6 and RCP 8.5. Thick lines represent ten-year rolling means. Thin lines represent annual average. All values are log-transformed.



Figure D.9: Regional changes in average precipitation (%) plotted against regional changes in average water erosion (%) between the decade 2010-2019 and the decade 2090-2099 in maize fields. Both datasets are log-transformed.



Figure D.10: Regional changes in average precipitation (%) plotted against regional changes in average water erosion (%) between the decade 2010-2019 and the decade 2090-2099 in wheat fields. Both datasets are log-transformed.



Figure D.11: Regional gains and losses in annual average maize and wheat production (million tonnes) simulated with the RCP 2.6 and RCP 8.5 scenario for the decade 2090-2099 compared to average production for the years 1980-2010 simulated with the AGMERRA climate data. Values are log-transformed.



Figure D.12: Global maize and wheat production (million tonnes) and production losses (million tonnes) due to water erosion per field management scenario simulated with the RCP 2.6 and RCP 8.5 scenario, calculated for the decade 2090-2099. The field management scenarios include the baseline scenario and all combinations with the cover crop (cover, mono) and the tillage management scenarios (conventional (C), reduced (R), no-tillage (N)).

Bibliography

Abdelwahab, O. M., Ricci, G. F., De Girolamo, A. M., and Gentile, F. (2018). Modelling soil erosion in a Mediterranean watershed: Comparison between SWAT and AnnAGNPS models. *Environmental Research*, 166(June):363–376.

Ainsworth, E., Leakey, A., Ort, D., and Long, S. (2008). FACE-ing the facts: inconsistencies and interdependence among field, chamber and modeling studies of elevated CO2 impacts on crop yield and food supply. *New Phytologist*, 179:5–9.

Al-Bakri, J., Suleiman, A., Abdulla, F., and Ayad, J. (2010). Potential impact of climate change on rainfed agriculture of a semi-arid basin in Jordan. *Physics and Chemistry of the Earth*, 36(5-6):125–134.

Alewell, C., Borrelli, P., Meusburger, K., and Panagos, P. (2019). Using the USLE: Chances, challenges and limitations of soil erosion modelling. *International Soil and Water Conservation Research*, 7(3):203–225.

Alexandratos, N. and Bruinsma, J. (2012). World agriculture towards 2030/2050: the 2012 revision. *ESA Working paper*, 12(12):146.

Almas, M. and Jamal, T. (2009). Use of RUSLE for Soil Loss Prediction During Different Growth Periods. *Pakistan Journal of Biological Sciences*, 3(1):118–121.

Alvarez, R., Steinbach, H. S., and De Paepe, J. L. (2017). Cover crop effects on soils and subsequent crops in the pampas: A meta-analysis. *Soil and Tillage Research*, 170:53–65.

Asseng, S., Ewert, F., Rosenzweig, C., Jones, J. W., Hatfield, J. L., Ruane, A. C., Boote, K. J., Thorburn,
P. J., Rötter, R. P., Cammarano, D., Brisson, N., Basso, B., Martre, P., Aggarwal, P. K., Angulo, C.,
Bertuzzi, P., Biernath, C., Challinor, A. J., Doltra, J., Gayler, S., Gol, R., Williams, J. R., and Wolf,
J. (2013). Uncertainty in simulating wheat yields under climate change. *Nature Climate Change*, 3(9):627–632.

Auerswald, K., Fiener, P., and Dikau, R. (2009). Rates of sheet and rill erosion in Germany - A metaanalysis. *Geomorphology*, 111(3-4):182–193. Auerswald, K., Kainz, M., and Fiener, P. (2004). Soil erosion potential of organic versus conventional farming evaluated by USLE modelling of cropping statistics for agricultural districts in Bavaria. *Soil Use and Management*, 19(4):305–311.

Bagarello, V. and Ferro, V. (2010). Analysis of soil loss data from plots of differing length for the Sparacia experimental area, Sicily, Italy. *Biosystems Engineering*, 105(3):411–422.

Bakker, M. M., Govers, G., Jones, R. A., and Rounsevell, M. D. (2007). The effect of soil erosion on Europe's crop yields. *Ecosystems*, 10(7):1209–1219.

Bakker, M. M., Govers, G., and Rounsevell, M. D. A. (2004). The crop productivity – erosion relationship : an analysis based on experimental work. *Catena*, 57:55–76.

Balkovič, J., Skalský, R., Folberth, C., Khabarov, N., Schmid, E., Madaras, M., Obersteiner, M., and van der Velde, M. (2018). Impacts and Uncertainties of +2°C of Climate Change and Soil Degradation on European Crop Calorie Supply. *Earth's Future*, 6(3):373–395.

Balkovič, J., van der Velde, M., Schmid, E., Skalský, R., Khabarov, N., Obersteiner, M., Stürmer, B., and Xiong, W. (2013). Pan-European crop modelling with EPIC: Implementation, up-scaling and regional crop yield validation. *Agricultural Systems*, 120(September):61–75.

Balkovič, J., van der Velde, M., Skalský, R., Xiong, W., Folberth, C., Khabarov, N., Smirnov, A., Mueller, N. D., and Obersteiner, M. (2014). Global wheat production potentials and management flexibility under the representative concentration pathways. *Global and Planetary Change*, 122:107–121.

Beasley, D. B., Huggins, L. F., and Monke, E. J. (1980). ANSWERS: a model for watershed planning. *Transactions, American Society of Agricultural Engineers*, 23(4):938–944.

Below, T. B., Mutabazi, K. D., Kirschke, D., Franke, C., Sieber, S., Siebert, R., and Tscherning, K. (2012). Can farmers' adaptation to climate change be explained by socio-economic household-level variables? *Global Environmental Change*, 22(1):223–235.

Benaud, P., Anderson, K., Evans, M., Farrow, L., Glendell, M., James, M. R., Quine, T. A., Quinton, J. N., Rawlins, B., Jane Rickson, R., and Brazier, R. E. (2020). National-scale geodata describe widespread accelerated soil erosion. *Geoderma*, 371:114378.

Bezak, N., Ballabio, C., Mikoš, M., Petan, S., Borrelli, P., and Panagos, P. (2020). Reconstruction of past rainfall erosivity and trend detection based on the REDES database and reanalysis rainfall. *Journal of Hydrology*, 590(April).

Bindi, M., Palosuo, T., Trnka, M., and Semenov, M. (2015). Modelling climate change impacts on crop production for food security. *Climate Research*, 65(February 2014):3–5.

Blenkinsop, S., Fowler, H. J., Barbero, R., Chan, S. C., Guerreiro, S. B., Kendon, E., Lenderink, G., Lewis, E., Li, X.-F., Westra, S., Alexander, L., Allan, R. P., Berg, P., Dunn, R. J. H., Ekström, M.,

Evans, J. P., Holland, G., Jones, R., Kjellström, E., Klein-Tank, A., Lettenmaier, D., Mishra, V., Prein, A. F., Sheffield, J., and Tye, M. R. (2018). The INTENSE project: using observations and models to understand the past, present and future of sub-daily rainfall extremes. *Advances in Science and Research*, 15:117–126.

Boardman, J. (1990). Soil erosion on the South Downs: a review. In Boardman, J., Foster, I., and Dearing, J., editors, *Soil Erosion on Agricultural Land*, pages 87–105. John Wiley Sons Ltd, Chichester.

Boardman, J. (1998). An average soil erosion rate for Europe: Myth or reality? *Journal of Soil and Water Conservation*, 53(1):46–50.

Boardman, J. (2003). Soil erosion and flooding on the eastern South Downs, southern England, 1976-2001. *Transactions of the Institute of British Geographers*, 28(2):176–196.

Boardman, J. (2006). Soil erosion science: Reflections on the limitations of current approaches. *Catena*, 68(2-3):73–86.

Boardman, J. and Evans, R. (2020). The measurement, estimation and monitoring of soil erosion by runoff at the field scale: Challenges and possibilities with particular reference to Britain. *Progress in Physical Geography*, 44(1):31–49.

Boix-Fayos, C., Martínez-Mena, M., Arnau-Rosalén, E., Calvo-Cases, A., Castillo, V., and Albaladejo, J. (2006). Measuring soil erosion by field plots: Understanding the sources of variation. *Earth-Science Reviews*, 78(3-4):267–285.

Bondeau, A., Smith, P. C., Zaehle, S., Schaphoff, S., Lucht, W., Cramer, W., Gerten, D., Lotze-campen, H., Müller, C., Reichstein, M., and Smith, B. (2007). Modelling the role of agriculture for the 20th century global terrestrial carbon balance. *Global Change Biology*, 13(3):679–706.

Boogaard, H., Wolf, J., Supit, I., Niemeyer, S., and van Ittersum, M. (2013). A regional implementation of WOFOST for calculating yield gaps of autumn-sown wheat across the European Union. *Field Crops Research*, 143:130–142.

Boote, K. J., Jones, J. W., White, J. W., Asseng, S., and Lizaso, J. I. (2013). Putting mechanisms into crop production models. *Plant, Cell and Environment*, 36(9):1658–1672.

Borrelli, P., Alewell, C., Alvarez, P., Anache, J. A. A., Baartman, J., Ballabio, C., Bezak, N., Biddoccu,
M., Cerdà, A., Chalise, D., Chen, S., Chen, W., De Girolamo, A. M., Gessesse, G. D., Deumlich, D.,
Diodato, N., Efthimiou, N., Erpul, G., Fiener, P., Freppaz, M., Gentile, F., Gericke, A., Haregeweyn,
N., Hu, B., Jeanneau, A., Kaffas, K., Kiani-Harchegani, M., Villuendas, I. L., Li, C., Lombardo, L.,
López-Vicente, M., Lucas-Borja, M. E., Märker, M., Matthews, F., Miao, C., Mikoš, M., Modugno,
S., Möller, M., Naipal, V., Nearing, M., Owusu, S., Panday, D., Patault, E., Patriche, C. V., Poggio,
L., Portes, R., Quijano, L., Rahdari, M. R., Renima, M., Ricci, G. F., Rodrigo-Comino, J., Saia, S.,
Samani, A. N., Schillaci, C., Syrris, V., Kim, H. S., Spinola, D. N., Oliveira, P. T., Teng, H., Thapa, R.,

Vantas, K., Vieira, D., Yang, J. E., Yin, S., Zema, D. A., Zhao, G., and Panagos, P. (2021). Soil erosion modelling: A global review and statistical analysis. *Science of the Total Environment*, 780.

Borrelli, P., Robinson, D. A., Fleischer, L. R., Lugato, E., Ballabio, C., Alewell, C., Meusburger, K., Modugno, S., Schütt, B., Ferro, V., Bagarello, V., Oost, K. V., Montanarella, L., and Panagos, P. (2017). An assessment of the global impact of 21st century land use change on soil erosion. *Nature Communications*, 8(1):1–13.

Borrelli, P., Robinson, D. A., Panagos, P., Lugato, E., Yang, J. E., Alewell, C., Wuepper, D., Montanarella, L., and Ballabio, C. (2020). Land use and climate change impacts on global soil erosion by water (2015-2070). *Proceedings of the National Academy of Sciences*, page 202001403.

Bouman, B. A. M., Van Keulen, H., Van Laar, H. H., and Rabbinge, R. (1996). The 'School of de Wit' crop growth simulation models: A pedigree and historical overview. *Agricultural Systems*, 52(2-3):171–198.

Brazier, R. (2004). Quantifying soil erosion by water in the UK: A review of monitoring and modelling approaches. *Progress in Physical Geography*, 28(3):340–365.

Brazier, R. E., Beven, K. J., Freer, J., and Rowan, J. S. (2000). Equifinality and uncertainty in physically based soil erosion models: Application of the glue methodology to WEPP-the water erosion prediction project-for sites in the UK and USA. *Earth Surface Processes and Landforms*, 25(8):825–845.

Brazier, R. E., Parsons, A. J., Wainwright, J., Powell, D. M., and Schlesinger, W. H. (2007). Upscaling understanding of nitrogen dynamics associated with overland flow in a semi-arid environment. *Biogeochemistry*, 82(3):265–278.

Brisson, N., Mary, B., Ripoche, D., Jeuffroy, M. H., Ruget, F., Nicoullaud, B., Gate, P., Devienne-Barret, F., Antonioletti, R., Durr, C., Richard, G., Beaudoin, N., Recous, S., Tayot, X., Plenet, D., Cellier, P., Machet, J.-M., Meynard, J. M., and Delécolle, R. (1998). STICS: a generic model for the simulation of crops and their water and nitrogen balances. I. Theory and parameterization applied to wheat and corn. *Agronomie*, 18(5-6):311–346.

Bryan, E., Deressa, T. T., Gbetibouo, G. A., and Ringler, C. (2009). Adaptation to climate change in Ethiopia and South Africa: options and constraints. *Environmental Science and Policy*, 12(4):413–426.

Buchhorn, M., Lesiv, M., Tsendbazar, N. E., Herold, M., Bertels, L., and Smets, B. (2020). Copernicus global land cover layers-collection 2. *Remote Sensing*, 12(6):1–14.

Burt, T., Boardman, J., Foster, I., and Howden, N. (2016). More rain, less soil: Long-term changes in rainfall intensity with climate change. *Earth Surface Processes and Landforms*, 41(4):563–566.

Busari, M. A., Kukal, S. S., Kaur, A., Bhatt, R., and Dulazi, A. A. (2015). Conservation tillage impacts on soil, crop and the environment. *International Soil and Water Conservation Research*, 3(2):119–129.

Carr, T., Balkovič, J., Dodds, P., Folberth, C., Fulajtar, E., and Skalsky, R. (2020). Uncertainties, sensitivities and robustness of simulated water erosion in an EPIC-based global-gridded crop model. *Biogeosciences*, 17:5263–5283.

Carr, T. W., Mkuhlani, S., Segnon, A. C., Ali, Z., Zougmoré, R., Dangour, A. D., Green, R., and Scheelbeek, P. (2022). Climate change impacts and adaptation strategies for crops in West Africa: A systematic review. *Environmental Research Letters*, 17(5).

Carter, E. K., Melkonian, J., Riha, S. J., and Shaw, S. B. (2016). Separating heat stress from moisture stress : analyzing yield response to high temperature in irrigated maize Separating heat stress from moisture stress : analyzing yield response to high temperature in irrigated maize.

Casali, J., Loizu, J., Campo, M., De Santisteban, L., and Alvarez-Mozos, J. (2006). Accuracy of methods for field assessment of rill and ephemeral gully erosion. *Catena*, 67:128–138.

Cassman, K. G. and Grassini, P. (2013). Can there be a green revolution in Sub-Saharan Africa without large expansion of irrigated crop production? *Global Food Security*, 2(3):203–209.

Cerdan, O., Govers, G., Le Bissonnais, Y., Van Oost, K., Poesen, J., Saby, N., Gobin, A., Vacca, A., Quinton, J., Auerswald, K., Klik, A., Kwaad, F. J., Raclot, D., Ionita, I., Rejman, J., Rousseva, S., Muxart, T., Roxo, M. J., and Dostal, T. (2010). Rates and spatial variations of soil erosion in Europe: A study based on erosion plot data. *Geomorphology*, 122(1-2):167–177.

Cerdan, O., Poesen, J., Govers, G., Saby, N., Le Bissonnais, Y., Gobin, A., Vacca, A., Quinton, J., Auerswald, K., Klik, A., Kwaad, F., and Roxo, M. (2006). Sheet and rill erosion. In Boardman, J. and Poesen, J., editors, *Soil Erosion in Europe*, pages 501–515. J. Wiley, Chichester.

CGIAR-CSI (2006). NASA Shuttle Radar Topographic Mission (SRTM). The SRTM data is available as 3 arc second (approx. 90m resolution) DEMs. The dataset is available for download at: http://srtm.csi.cgiar.org/.

Challinor, A. J., Ewert, F., Arnold, S., Simelton, E., and Fraser, E. (2009). Crops and climate change: Progress, trends, and challenges in simulating impacts and informing adaptation. *Journal of Experimental Botany*, 60(10):2775–2789.

Challinor, A. J., Müller, C., Asseng, S., Deva, C., Nicklin, K. J., Wallach, D., Vanuytrecht, E., Whitfield, S., Ramirez-Villegas, J., and Koehler, A.-K. (2017). Improving the use of crop models for risk assessment and climate change adaptation. *Agricultural Systems*, (June).

Chaplot, V. (2007). Water and soil resources response to rising levels of atmospheric CO2 concentration and to changes in precipitation and air temperature. *Journal of Hydrology*, 337(1-2):159–171.

Chappell, A., Baldock, J., and Sanderman, J. (2016). The global significance of omitting soil erosion from soil organic carbon cycling schemes. *Nature Climate Change*, 6(2):187–191.

Chen, J., Xiao, H., Li, Z., Liu, C., Ning, K., and Tang, C. (2020). How effective are soil and water conservation measures (SWCMs) in reducing soil and water losses in the red soil hilly region of China? A meta-analysis of field plot data. *Science of the Total Environment*, 735:139517.

Cherlet, M., Hutchinson, C., Reynolds, J., Hill, J., Sommer, S., and Von Maltitz, G. (2018). *World Atlas of Desertification*. Publication Office of the European Union, Luxembourg.

Chung, S. W., Gassman, P. W., Kramer, L. A., Williams, J. R., and Gu, R. (1999). Validation of EPIC for two watersheds in southwest Iowa.

Claassen, R., Bowman, M., Mcfadden, J., Smith, D., and Wallander, S. (2018). Tillage Intensity and Conservation Cropping in the United States.

Claessens, L., Antle, J. M., Stoorvogel, J. J., Valdivia, R. O., Thornton, P. K., and Herrero, M. (2012). A method for evaluating climate change adaptation strategies for small-scale farmers using survey, experimental and modeled data. *Agricultural Systems*, 111:85–95.

Clark, E. (1985). The off-site costs of soil erosion. *Journal of Soil and Water Conservation*, 40(1):19–22.

Clark, M. A., Domingo, N. G. G., Colgan, K., Thakrar, S. K., Tilman, D., Lynch, J., Azevedo, I. L., and Hill, J. D. (2020). Global food system emissions could preclude achieving the 1.5° and 2°C climate change targets. *Science*, 370(6517):705–708.

Cohen, M. J., Shepherd, K. D., and Walsh, M. G. (2005). Empirical reformulation of the universal soil loss equation for erosion risk assessment in a tropical watershed. *Geoderma*, 124(3-4):235–252.

Collins, W., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, T., Jones, C., Liddicoat, S., Martin, G., O'Connor, F., Rae, J., Senior, C., Totterdell, I., Woodward, S., Reichler, T., and Kim, J. (2008). Evaluation of the HadGEM2 model. Met Office Hadley Centre Technical Note no. HCTN 74. Technical report.

Corsi, S., Friedrich, T., Kassam, A., Pisante, M., and Sà, J. D. M. (2012). Soil Organic Carbon Accumulation and Greenhouse Gas Emission Reductions from Conservation Agriculture: a literature review, volume 16.

Cowie, A. L., Orr, B. J., Castillo Sanchez, V. M., Chasek, P., Crossman, N. D., Erlewein, A., Louwagie,
G., Maron, M., Metternicht, G. I., Minelli, S., Tengberg, A. E., Walter, S., and Welton, S. (2018).
Land in balance: The scientific conceptual framework for Land Degradation Neutrality. *Environmental Science Policy*, 79:25–35.

Crosson, P. (1995). Soil Erosion Estimates and Costs. Science, 269(5233):461–464.

De la Rosa, D., Moreno, J. A., Mayol, F., and Bonsón, T. (2000). Assessment of soil erosion vulnerability in western Europe and potential impact on crop productivity due to loss of soil depth using the Impe1ERO model. *Agriculture, Ecosystems and Environment*, 81(3):179–190. De Ploey, J. and Gabriels, D. (1980). Measuring soil loss and experimental studies. In Kirkby, M. and Morgan, R., editors, *Soil Erosion*, pages 63–108. Willey, Chichester.

de Vente, J. and Poesen, J. (2005). Predicting soil erosion and sediment yield at the basin scale: Scale issues and semi-quantitative models. *Earth-Science Reviews*, 71(1-2):95–125.

de Vente, J., Poesen, J., Arabkhedri, M., and Verstraeten, G. (2007). The sediment delivery problem revisited. *Progress in Physical Geography*, 31(2):155–178.

De Vente, J., Poesen, J., Verstraeten, G., Govers, G., Vanmaercke, M., Van Rompaey, A., Arabkhedri, M., and Boix-Fayos, C. (2013). Predicting soil erosion and sediment yield at regional scales: Where do we stand? *Earth-Science Reviews*, 127:16–29.

De Wit, C. (1958). Transpiration and Crop Yields. Technical report, Wageningen.

De Wit, C. (1965). Photosynthesis of Leaf Canopies. *Centre for Agricultural Publications and Documentation, Wageningen*, 663:57.

den Biggelaar, C., Lal, R., Wiebe, K., and Breneman, V. (2001). Impact of soil erosion on crop yields in North America. *Advances in Agronomy*, 72:1–52.

Den Biggelaar, C., Lal, R., Wiebe, K., and Breneman, V. (2004a). The Global Impact of Soil Erosion on Productivity. *Advances in Agronomy*, 81(03):1–4.

Den Biggelaar, C., Lal, R., Wiebe, K., Eswaran, H., Breneman, V., and Reich, P. (2004b). The Global Impact Of Soil Erosion On Productivity*. II: Effects On Crop Yields And Production Over Time. *Advances in Agronomy*, 81(03):49–95.

Deng, L., ping Shangguan, Z., and Li, R. (2012). Effects of the grain-for-green program on soil erosion in China. *International Journal of Sediment Research*, 27(1):120–127.

Deryng, D., Conway, D., Ramankutty, N., Price, J., and Warren, R. (2014). Global crop yield response to extreme heat stress under multiple climate change futures. *Environmental Research Letters*, 9(3):34011.

Deryng, D., Sacks, W. J., Barford, C. C., and Ramankutty, N. (2011). Simulating the effects of climate and agricultural management practices on global crop yield. *Global Biogeochemical Cycles*, 25(2).

Dodds, P. E. (2010). *Development of a crop model to examine crop management and climate change in Senegal*. PhD thesis, University of Leeds.

Doetterl, S., Van Oost, K., and Six, J. (2012). Towards constraining the magnitude of global agricultural sediment and soil organic carbon fluxes. *Earth Surface Processes and Landforms*, 37(6):642–655.

Duan, X., Liu, B., Gu, Z., Rong, L., and Feng, D. (2016). Quantifying soil erosion effects on soil productivity in the dry-hot valley, southwestern China. *Environmental Earth Sciences*, 75(16):1–9.

Dzotsi, K. A., Jones, J. W., Adiku, S. G., Naab, J. B., Singh, U., Porter, C. H., and Gijsman, A. J. (2010). Modeling soil and plant phosphorus within DSSAT. *Ecological Modelling*, 221(23):2839–2849.

Enters, T. (1998). *Methods for the economic assessment of the on- and off-site impacts of soil erosion*. Number 2.

Erisman, J. W., Sutton, M. A., Galloway, J., Klimont, Z., and Winiwarter, W. (2008). How a century of ammonia synthesis changed the world. *Nature Geoscience*, 1(10):636–639.

Estes, L. D., Bradley, B. A., Beukes, H., Hole, D. G., Lau, M., Oppenheimer, M. G., Schulze, R., Tadross, M. A., and Turner, W. R. (2013). Comparing mechanistic and empirical model projections of crop suitability and productivity: Implications for ecological forecasting. *Global Ecology and Biogeography*, 22(8):1007–1018.

Evans, D. L., Quinton, J. N., Tye, A. M., Rodés, Á., Davies, J. A., Mudd, S. M., and Quine, T. A. (2019). Arable soil formation and erosion: A hillslope-based cosmogenic nuclide study in the United Kingdom. *Soil*, 5(2):253–263.

Evans, R. (1986). Finding out about water erosion. Teaching Geography, 12:17-20.

Evans, R. (2002). An alternative way to assess water erosion of cultivated land – field-based measurements: An analysis of some results. *Applied Geography*, 22:187–208.

Evans, R. (2013). Assessment and monitoring of accelerated water erosion of cultivated land - when will reality be acknowledged? *Soil Use and Management*, 29(1):105–118.

Evans, R. and Boardman, J. (2016a). A reply to panagos et al., 2016 (Environmental science policy 59 (2016) 53-57. *Environmental Science and Policy*, 60:63–68.

Evans, R. and Boardman, J. (2016b). The new assessment of soil loss by water erosion in Europe. Panagos P. et al., 2015 Environmental Science Policy 54, 438-447-A response. *Environmental Science and Policy*, 58:11–15.

Evans, R. and Brazier, R. (2005). Evaluation of modelled spatially distributed predictions of soil erosion by water versus field-based assessments. *Environ. Sci. Pol.*, 8:493–501.

Ewert, F., Rötter, R. P., Bindi, M., Webber, H., Trnka, M., Kersebaum, K. C., Olesen, J. E., van Ittersum, M. K., Janssen, S., Rivington, M., Semenov, M. A., Wallach, D., Porter, J. R., Stewart, D., Verhagen, J., Gaiser, T., Palosuo, T., Tao, F., Nendel, C., Roggero, P. P., Bartošová, L., and Asseng, S. (2015). Crop modelling for integrated assessment of risk to food production from climate change. *Environmental Modelling and Software*, 72:287–303.

Ewert, F., Rounsevell, M. D., Reginster, I., Metzger, M. J., and Leemans, R. (2005). Future scenarios of European agricultural land use: I. Estimating changes in crop productivity. *Agriculture, Ecosystems and Environment*, 107(2-3):101–116.

Ewert, F., van Ittersum, M. K., Heckelei, T., Therond, O., Bezlepkina, I., and Andersen, E. (2011). Scale changes and model linking methods for integrated assessment of agri-environmental systems. *Agriculture, Ecosystems and Environment*, 142(1-2):6–17.

FAO (2006). Agriculture and the environment: changing pressures, solutions and trade-offs. In *World Agriculture: Toward 2015/2030*, pages 331–356.

FAO (2016). AQUASTAT Main Database. available at: http://www.fao.org/nr/water/aquastat/data (last access: 08.03.2021).

FAO (2017). Voluntary Guidelines for Sustainable Soil Management. Technical report, Food and Agriculture Organization of the United Nations, Rome.

FAO (2018). The future of food and agriculture – Alternative pathways to 2050. Technical report, Rome.

FAO (2020). FAOSTAT. available at: http://www.fao.org/faostat/en/data (last access: 08.03.2021).

FAO and ITPS (2015). Status of the World's Soil Resources (SWSR) - Main Report. Technical report, Food and Agriculture Organization of the United Nations and Intergovernmental Technical Panel on Soils, Rome.

FAO/IIASA/ISRIC/ISSCAS/JRC (2012). Harmonized World Soil Database (version 1.2). Technical report, FAO, Rome, Italy and IIASA, Laxenburg, Austria.

Favis-Mortlock, D. (1998). Validation of Field-Scale Soil Erosion Models Using Common Datasets. Number Vol.1 55.

Favis-Mortlock, D. and Boardman, J. (1995). Nonlinear responses of soil erosion to climate change: a modelling study on the UK South Downs. *Catena*, 25(1-4):365–387.

Favis-Mortlock, D. and Mullan, D. (2011). *Soil erosion by water under future climate change*. Number August.

Favis-Mortlock, D. T., Evans, R., Boardman, J., and Harris, T. M. (1991). Climate change, winter wheat yield and soil erosion on the English south downs. *Agricultural Systems*, 37(4):415–433.

Feddema, J. J. and Freire, S. (2001). Soil degradation, global warming and climate impacts. *Climate Research*, 17(2 SPECIAL 8):209–216.

Fick, S. and Hijmans, R. (2017). Worldclim 2: New 1-km spatial resolution climate surfaces for global land areas. *International Journal of Climatology*, 37(12):4302–4315.

Fischer, E. M., Sedláček, J., Hawkins, E., and Knutti, R. (2014). Models agree on forced response pattern of precipitation and temperature extremes. *Geophysical Prospecting*, (April):6413–6419.

Fischer, F. K., Kistler, M., Brandhuber, R., Maier, H., Treisch, M., and Auerswald, K. (2018). Validation of official erosion modelling based on high-resolution radar rain data by aerial photo erosion classification. *Earth Surface Processes and Landforms*, 43(1):187–194.

Fleige, H. and Horn, R. (2000). Field experiments on the effect of soil compaction on soil properties, runoff, interflow and erosion. *Advances in Geoecology*, 32.

Folberth, C., Elliott, J., Müller, C., Balkovič, J., Chryssanthacopoulos, J., Izaurralde, R. C., Jones, C. D., Khabarov, N., Liu, W., Reddy, A., Schmid, E., Skalský, R., Yang, H., Arneth, A., Ciais, P., Deryng, D., Lawrence, P. J., Olin, S., Pugh, T. A. M., Ruane, A. C., and Wang, X. (2019). Parameterization-induced uncertainties and impacts of crop management harmonization in a global gridded crop model ensemble. *Plos One*, 14(9):e0221862.

Folberth, C., Skalský, R., Moltchanova, E., Balkovič, J., Azevedo, L. B., Obersteiner, M., and van der Velde, M. (2016). Uncertainty in soil data can outweigh climate impact signals in global crop yield simulations. *Nature Communications*, 7(May):11872.

Folberth, C., Yang, H., Gaiser, T., Abbaspour, K. C., and Schulin, R. (2013). Modeling maize yield responses to improvement in nutrient, water and cultivar inputs in sub-Saharan Africa. *Agricultural Systems*, 119:22–34.

Folberth, C., Yang, H., Gaiser, T., Liu, J., Wang, X., Williams, J., and Schulin, R. (2014). Effects of ecological and conventional agricultural intensification practices on maize yields in sub-Saharan Africa under potential climate change. *Environmental Research Letters*, 9(June):044004.

Frieler, K., Elliott, J., Folberth, C., Deryng, D., Müller, C., Olin, S., Pugh, T. A. M., and Schaphoff, S. (2017). Understanding the weather signal in national crop-yield variability. (May).

Fritz, S., See, L., Mccallum, I., You, L., Bun, A., Moltchanova, E., Duerauer, M., Albrecht, F., Schill,
C., Perger, C., Havlik, P., Mosnier, A., Thornton, P., Wood-Sichra, U., Herrero, M., Becker-Reshef, I.,
Justice, C., Hansen, M., Gong, P., Abdel Aziz, S., Cipriani, A., Cumani, R., Cecchi, G., Conchedda,
G., Ferreira, S., Gomez, A., Haffani, M., Kayitakire, F., Malanding, J., Mueller, R., Newby, T.,
Nonguierma, A., Olusegun, A., Ortner, S., Rajak, D. R., Rocha, J., Schepaschenko, D., Schepaschenko,
M., Terekhov, A., Tiangwa, A., Vancutsem, C., Vintrou, E., Wenbin, W., van der Velde, M., Dunwoody,
A., Kraxner, F., and Obersteiner, M. (2015). Mapping global cropland and field size. *Global Change Biology*, 21(5):1980–1992.

Fu, B. J., Zhao, W. W., Chen, L. D., Zhang, Q. J., Lü, Y. H., Gulinck, H., and Poesen, J. (2005). Assessment of soil erosion at large watershed scale using RUSLE and GIS: A case study in the Loess Plateau of China. *Land Degradation and Development*, 16(1):73–85.

Fulajtar, E., Mabit, L., Renschler, C., and Lee Zhi Yi, A. (2017). Use of 137Cs for soil erosion assessment. FAO/IAEA.

Gabbert, S., van Ittersum, M., Kroeze, C., Stalpers, S., Ewert, F., and Alkan Olsson, J. A. (2010). Uncertainty analysis in integrated assessment: The users' perspective. *Regional Environmental Change*, 10(2):131–143.

García-Ruiz, J. M. (2010). The effects of land uses on soil erosion in Spain: A review. *Catena*, 81(1):1–11.

García-Ruiz, J. M., Beguería, S., Nadal-Romero, E., González-Hidalgo, J. C., Lana-Renault, N., and Sanjuán, Y. (2015). A meta-analysis of soil erosion rates across the world. *Geomorphology*, 239:160–173.

Gasser, T., Guivarch, C., Tachiiri, K., Jones, C. D., and Ciais, P. (2015). Negative emissions physically needed to keep global warming below 2°C. *Nature Communications*, 6.

Gautam, S., Mbonimpa, E. G., Kumar, S., Bonta, J. V., and Lal, R. (2015). Agricultural Policy Environmental eXtender model simulation of climate change impacts on runoff from a small no-till watershed. *Journal of Soil and Water Conservation*, 70(2):101–109.

Gbetibouo, G. A., Hassan, R. M., and Ringler, C. (2010). Modelling farmers' adaptation strategies for climate change and variability: The case of the Limpopo Basin, South Africa. *Agrekon*, 49(2):217–234.

Gerland, P., Raftery, A. E., Ševčíková, H., Li, N., Gu, D., Spoorenberg, T., Alkema, L., Fosdick, B. K., Chunn, J., Lalic, N., Bay, G., Buettner, T., Heilig, G. K., and Wilmoth, J. (2014). World population stabilization unlikely this century. *Science*, 346(6206):234–237.

Gibbs, H. K. and Salmon, J. M. (2015). Mapping the world's degraded lands. *Applied Geography*, 57:12–21.

Giller, K. E., Andersson, J. A., Corbeels, M., Kirkegaard, J., Mortensen, D., Erenstein, O., and Vanlauwe, B. (2015). Beyond conservation agriculture. *Frontiers in Plant Science*, 6(OCTOBER).

Giller, K. E., Witter, E., Corbeels, M., and Tittonell, P. (2009). Conservation agriculture and smallholder farming in Africa: The heretics' view. *Field Crops Research*, 114(1):23–34.

Giorgi, F. and Lionello, P. (2008). Climate change projections for the Mediterranean region. *Global and Planetary Change*, 63(2-3):90–104.

Görlach, B., Landgrebe-trinkunaite, R., Interwies, E., Bouzit, M., Darmendrail, D., and Rinaudo, J.-d. (2004). Assessing the Economic Impacts of Soil Degradation - Final Report Volume III. IV(December):1–31.

Grassini, P., Yang, H., and Cassman, K. G. (2009). Limits to maize productivity in Western Corn-Belt: A simulation analysis for fully irrigated and rainfed conditions. *Agricultural and Forest Meteorology*, 149(8):1254–1265. Graves, A. R., Morris, J., Deeks, L. K., Rickson, R. J., Kibblewhite, M. G., Harris, J. A., Farewell, T. S., and Truckle, I. (2015). The total costs of soil degradation in England and Wales. *Ecological Economics*, 119:399–413.

Guerreiro, S. B., Fowler, H. J., Barbero, R., Westra, S., Lenderink, G., Blenkinsop, S., Lewis, E., and Li, X. F. (2018). Detection of continental-scale intensification of hourly rainfall extremes. *Nature Climate Change*, 8(9):803–807.

GYGA (2015). Global Yield Gap Atlas. available at: http://www.yieldgap.org/ (last access: 21.11.2020).

Haile, G. W. and Fetene, M. (2012). Assessment of soil erosion hazard in kilie catchment, East Shoa, Ethiopia. *Land Degradation and Development*, 23(3):293–306.

Hatfield, J. L., Boote, K. J., Kimball, B. A., Ziska, L. H., Izaurralde, R. C., Ort, D., Thomson, A. M., and Wolfe, D. (2011). Climate impacts on agriculture: Implications for crop production. *Agronomy Journal*, 103(2):351–370.

Hazell, P. and Wood, S. (2008). Drivers of change in global agriculture. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 363(1491):495–515.

Hein, L. (2007). Assessing the costs of land degradation: a case study for the puentes catchment, southeast Spain. *Land Degradation Development*, 18:631–642.

Hempel, S., Frieler, K., Warszawski, L., Schewe, J., and Piontek, F. (2013). A trend-preserving bias correction ndash; The ISI-MIP approach. *Earth System Dynamics*, 4(2):219–236.

Herrick, J. E., Sala, O. E., and Karl, J. W. (2013). Land degradation and climate change: A sin of omission? *Frontiers in Ecology and the Environment*, 11(6):283.

Herrick, J. E., Shaver, P., Pyke, D. A., Pellant, M., Toledo, D., and Lepak, N. (2019). A strategy for defining the reference for land health and degradation assessments. *Ecological Indicators*, 97(June 2018):225–230.

Hertel, T. and Lobell, D. B. (2014). Agricultural adaptation to climate change in rich and poor countries: Current modeling practice and potential for empirical contributions. *Energy Economics*, 46(C):562–575.

Herweg, K. (1988). The applicability of large-scale geomorphological mapping to erosion control and soil conservation in a research area in Tuscany. *Zeitschrift fur Geomorphologie, Supplementband*, 68:175–187.

Hively, W. D., Lamb, B. T., Daughtry, C. S., Shermeyer, J., McCarty, G. W., and Quemada, M. (2018). Mapping crop residue and tillage intensity using WorldView-3 satellite shortwave infrared residue indices. *Remote Sensing*, 10(10). Hjulstrom, F. (1935). Studies of the morphological activity of rivers as illustrated by the River Fyris. *Bulletin of the Geological Institute University of Uppsala*, 25:221–527.

Hoffmann, H., Zhao, G., Asseng, S., Bindi, M., Biernath, C., Constantin, J., Coucheney, E., Dechow,
R., Doro, L., Eckersten, H., Gaiser, T., Grosz, B., Heinlein, F., Kassie, B. T., Kersebaum, K. C.,
Klein, C., Kuhnert, M., Lewan, E., Moriondo, M., Nendel, C., Priesack, E., Raynal, H., Roggero, P. P.,
Rötter, R. P., Siebert, S., Specka, X., Tao, F., Teixeira, E., Trombi, G., Wallach, D., Weihermüller, L.,
Yeluripati, J., and Ewert, F. (2016). Impact of spatial soil and climate input data aggregation on regional
Yield Simulations. *PLoS ONE*, 11(4):1–23.

Hoogenboom, G., Porter, C., Shelia, V., Boote, K., Singh, U., White, J., Hunt, L., Ogoshi, R., Lizaso, J., Koo, J., Asseng, S., Singels, A., Moreno, L., and Jones, J. (2019). Decision Support System for Agrotechnology Transfer (DSSAT) Version 4.7.5.

Hsieh, Y. P., Grant, K., and Bugna, G. (2009). A field method for soil erosion measurements in agricultural and natural lands. *Journal of Soil and Water Conservation*, 64(6):374–382.

Hudson, N. (1993). *Field measurement of soil erosion and runoff*. Number 68 in FAO soils bulletin. Food and Agriculture Organization of the United Nations.

Humphries, R. N. and Brazier, R. E. (2018). Exploring the case for a national-scale soil conservation and soil condition framework for evaluating and reporting on environmental and land use policies. *Soil Use and Management*, 34(1):134–146.

Huntington, J. L., Hegewisch, K. C., Daudert, B., Morton, C. G., Abatzoglou, J. T., McEvoy, D. J., and Erickson, T. (2017). Climate engine: Cloud computing and visualization of climate and remote sensing data for advanced natural resource monitoring and process understanding. *Bulletin of the American Meteorological Society*, 98(11):2397–2409.

IIASA/FAO (2012). Global Agro-ecological Zones (GAEZ v3.0). Technical report, IIASA, Laxenburg, Austria and FAO, Rome, Italy.

Iizumi, T., Furuya, J., Shen, Z., Kim, W., Okada, M., Fujimori, S., Hasegawa, T., and Nishimori, M. (2017). Responses of crop yield growth to global temperature and socioeconomic changes. *Scientific Reports*, 7(1):1–10.

Iizumi, T., Shiogama, H., Imada, Y., Hanasaki, N., Takikawa, H., and Nishimori, M. (2018). Crop production losses associated with anthropogenic climate change for 1981–2010 compared with preindustrial levels. *International Journal of Climatology*, 38(14):5405–5417.

IPCC (2014). Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Core Writing Team, R.K. Pachauri and L.A. Meyer (eds.)]. Technical report, Geneva. Islam, R. and Reeder, R. (2014). No-till and conservation agriculture in the United States: An example from the David Brandt farm, Carroll, Ohio. *International Soil and Water Conservation Research*, 2(1):97–107.

Istanbulluoglu, E. and Bras, R. L. (2006). On the dynamics of soil moisture, vegetation, and erosion: Implications of climate variability and change. *Water Resources Research*, 42(6):1–17.

Ito, A. (2007). Simulated impacts of climate and land-cover change on soil erosion and implication for the carbon cycle, 1901 to 2100. *Geophysical Research Letters*, 34(9):1–5.

Izaurralde, R. C., Williams, J. R., McGill, W. B., Rosenberg, N. J., and Jakas, M. C. Q. (2006). Simulating soil C dynamics with EPIC: Model description and testing against long-term data. *Ecological Modelling*, 192(3-4):362–384.

Jacks, G. V. and Whyte, R. O. (1939). *The Rape of the Earth: A World Survey of Soil Erosion*. Faber and Faber, London.

Jägermeyr, J., Gerten, D., Schaphoff, S., Heinke, J., Lucht, W., and Rockström, J. (2016). Integrated crop water management might sustainably halve the global food gap. *Environmental Research Letters*, 11(2).

Jägermeyr, J., Müller, C., Ruane, A. C., Elliott, J., Balkovic, J., Castillo, O., Faye, B., Foster, I., Folberth, C., Franke, J. A., Fuchs, K., Guarin, J. R., Heinke, J., Hoogenboom, G., Iizumi, T., Jain, A. K., Kelly, D., Khabarov, N., Lange, S., Lin, T.-s., Liu, W., Mialyk, O., Minoli, S., Moyer, E. J., Okada, M., Phillips, M., Porter, C., Rabin, S. S., Scheer, C., Schneider, J. M., Schyns, J. F., Skalsky, R., Smerald, A., Stella, T., Stephens, H., Webber, H., Zabel, F., and Rosenzweig, C. (2021). Climate impacts on global agriculture emerge earlier in new generation of climate and crop models. *Nature Food*, 2(November).

Jenks, G. (1967). The Data Model Concept in Statistical Mapping. *International Yearbook of Cartography*, 7:186–190.

Jetten, V., Govers, G., and Hessel, R. (2003). Erosion models: Quality of spatial predictions. *Hydrological Processes*, 17(5):887–900.

Jones, J. W., Antle, J. M., Basso, B., Boote, K. J., Conant, R. T., Foster, I., Godfray, H. C. J., Herrero, M., Howitt, R. E., Janssen, S., Keating, B. A., Munoz-Carpena, R., Porter, C. H., Rosenzweig, C., and Wheeler, T. R. (2016). Toward a new generation of agricultural system data, models, and knowledge products: State of agricultural systems science. *Agricultural Systems*.

Kaiser, J. (2004). Wounding Earth's Fragile Skin. Science, 304(June):1616–1618.

Kaiser, V. (1978). Annual erosion survey of Whitman county, Washington. 1939/40-1975/76. Technical report, USDA-SCS, Spokane, WA 99201.
Kang, Y., Khan, S., and Ma, X. (2009). Climate change impacts on crop yield, crop water productivity and food security - A review. *Progress in Natural Science*, 19(12):1665–1674.

Karydas, C. G., Panagos, P., and Gitas, I. Z. (2014). A classification of water erosion models according to their geospatial characteristics. *International Journal of Digital Earth*, 7(3):229–250.

Karydas, C. G., Sekuloska, T., and Silleos, G. N. (2009). Quantification and site-specification of the support practice factor when mapping soil erosion risk associated with olive plantations in the Mediterranean island of Crete. *Environmental Monitoring and Assessment*, 149(1-4):19–28.

Keating, B. A., Carberry, P. S., Hammer, G. L., Probert, M. E., Robertson, M. J., Holzworth, D., Huth, N. I., Hargreaves, J. N. G., Meinke, H., Hochman, Z., McLean, G., Verburg, K., Snow, V., Dimes, J. P., Silburn, M., Wang, E., Brown, S., Bristow, K. L., Asseng, S., Chapman, S., McCown, R. L., Freebairn, D. M., and Smith, C. J. (2003). An overview of APSIM, a model designed for farming systems simulation. *European Journal of Agronomy*, 18(3-4):267–288.

Keesstra, S., Mol, G., de Leeuw, J., Okx, J., Molenaar, C., de Cleen, M., and Visser, S. (2018). Soilrelated sustainable development goals: Four concepts to make land degradation neutrality and restoration work. *Land*, 7(4).

Keesstra, S. D., Bouma, J., Wallinga, J., Tittonell, P., Smith, P., Cerdà, A., Montanarella, L., Quinton, J. N., Pachepsky, Y., Van Der Putten, W. H., Bardgett, R. D., Moolenaar, S., Mol, G., Jansen, B., and Fresco, L. O. (2016). The significance of soils and soil science towards realization of the United Nations sustainable development goals. *Soil*, 2(2):111–128.

Kersebaum, K. C. (2007). Modelling nitrogen dynamics in soil–crop systems with HERMES. *Nutrient Cycling in Agroecosystems*, 77(1):39–52.

Kinnell, P. I. (2010). Event soil loss, runoff and the Universal Soil Loss Equation family of models: A review. *Journal of Hydrology*, 385(1-4):384–397.

Klik, A. and Eitzinger, J. (2010). Impact of climate change on soil erosion and the efficiency of soil conservation practices in Austria. *Journal of Agricultural Science*, 148(5):529–541.

Knox, J., Hess, T., Daccache, A., and Wheeler, T. (2012). Climate change impacts on crop productivity in Africa and South Asia. *Environmental Research Letters*, 7(3):034032.

Kort, J., Collins, M., and Ditsch, D. (1998). A review of soil erosion potential associated with biomass crops. *Biomass and Bioenergy*, 14(4):351–359.

Krishna Kumar, K., Kumar, R., Ashrit, R., Deshpande, N., and Hansen, J. (2004). Climate impacts on Indian agriculture. *International Journal of Climatology*, 24(11):1375–1393.

Laban, P., Metternicht, G., and Davies, J. (2018). Soil biodiversity and soil organic carbon: keeping drylands alive.

Labrière, N., Locatelli, B., Laumonier, Y., Freycon, V., and Bernoux, M. (2015). Soil erosion in the humid tropics: A systematic quantitative review. *Agriculture, Ecosystems and Environment*, 203:127–139.

Lado, M., Beh-Hur, M., and Shainberg, I. (2004). Soil Wetting and Texture Effects on Aggregate Stability, Seal Formation, and Erosion. *Soil Science Society of America Journal*, 68(6):1992–1999.

Lal, R. (1995). Erosion-crop productivity relationships for soils of Africa. Soil Science Society of America Journal, 59(3):661–667.

Lal, R. (2002). Soil carbon dynamics in cropland and rangeland. *Environmental Pollution*, 116(3):353–362.

Lane, L. J., Nearing, M. A., Laflen, J. M., Foster, G. R., and Nichols, M. H. (1992). Description of the US Department of Agriculture water erosion prediction project (WEPP) model. In Parsons, A. and Abrahams, A., editors, *Overland Flow: Hydraulics and Erosion Mechanics.*, pages 377–391. UCL Press, London.

Langbein, W. B. and Schumm, S. A. (1958). Yield of sediment in relation to mean annual precipitation. *Eos, Transactions American Geophysical Union*, 39(6):1076–1084.

Laniak, G. F., Olchin, G., Goodall, J., Voinov, A., Hill, M., Glynn, P., Whelan, G., Geller, G., Quinn, N., Blind, M., Peckham, S., Reaney, S., Gaber, N., Kennedy, R., and Hughes, A. (2013). Integrated environmental modeling: A vision and roadmap for the future. *Environmental Modelling and Software*, 39:3–23.

Larney, F. J., Janzen, H. H., Olson, B. M., and Olson, A. F. (2009). Erosion-productivity-soil amendment relationships for wheat over 16 years. *Soil and Tillage Research*, 103(1):73–83.

Lee, J. J., Phillips, D. L., and Dodson, R. F. (1996). Sensitivity of the US corn belt to climate change and elevated CO2: II. Soil erosion and organic carbon. *Agricultural Systems*, 52(4):503–521.

Leek, R. and Olsen, P. (2000). Modelling climatic erosivity as a factor for soil erosion in Denmark : changes and temporal trends. *Soil Use and Management*, 16:61–65.

Leng, G. and Hall, J. (2019). Crop yield sensitivity of global major agricultural countries to droughts and the projected changes in the future. *Science of the Total Environment*, 654:811–821.

Lenton, T. M., Rockström, J., Gaffney, O., Rahmstorf, S., Richardson, K., Steffen, W., and Schellnhuber, H. J. (2019). An aeroplane flies over a glacier in the Wrangell St Elias National Park in Alaska. *Nature*, 575:592–595.

Leopold, L. B. (1951). Rainfall Frequency: An aspect of climatic variation. *Eos, Transactions American Geophysical Union*, 32(3):347–357.

Lesiv, M., Laso Bayas, J. C., See, L., Duerauer, M., Dahlia, D., Durando, N., Hazarika, R., Kumar Sahariah, P., Vakolyuk, M., Blyshchyk, V., Bilous, A., Perez-Hoyos, A., Gengler, S., Prestele, R., Bilous, S., Akhtar, I. u. H., Singha, K., Choudhury, S. B., Chetri, T., Malek, Ž., Bungnamei, K., Saikia, A., Sahariah, D., Narzary, W., Danylo, O., Sturn, T., Karner, M., McCallum, I., Schepaschenko, D., Moltchanova, E., Fraisl, D., Moorthy, I., and Fritz, S. (2019). Estimating the global distribution of field size using crowdsourcing. *Global Change Biology*, 25(1):174–186.

Levis, S., Badger, A., Drewniak, B., Nevison, C., and Ren, X. (2018). CLMcrop yields and water requirements: avoided impacts by choosing RCP 4.5 over 8.5. *Climatic Change*, 146(3-4):501–515.

Li, Z. and Fang, H. (2016). Impacts of climate change on water erosion: A review. *Earth-Science Reviews*, 163:94–117.

Licker, R., Johnston, M., Foley, J. A., Barford, C., Kucharik, C. J., Monfreda, C., and Ramankutty, N. (2010). Mind the gap: How do climate and agricultural management explain the 'yield gap' of croplands around the world? *Global Ecology and Biogeography*, 19(6):769–782.

Liniger, H., Harari, N., van Lynden, G., Fleiner, R., de Leeuw, J., Bai, Z., and Critchley, W. (2019). Achieving land degradation neutrality: The role of SLM knowledge in evidence-based decision-making. *Environmental Science and Policy*, 94(June 2018):123–134.

Littleboy, M., Cogle, A. L., Smith, G. D., Rao, C. K., and Yule, D. F. (1996). Soil management and production of Alfisols in the semi-arid tropics. IV. Simulation of decline in productivity caused by soil erosion. *Australian Journal of Soil Research*, 34(1):127–138.

Littleboy, M., Silburn, D., Freebairn, D., Woodruff, D., and Hammer, G. (1989). *PERFECT - a computer simulation model of Productivity Erosion Runoff Functions to Evaluate Conservation Techniques*. Department of Primary Industries.

Liu, B., Asseng, S., Müller, C., Ewert, F., Elliott, J., Lobell, D. B., Martre, P., Ruane, A. C., Wallach, D., Jones, J. W., Rosenzweig, C., Aggarwal, P. K., Alderman, P. D., Anothai, J., Basso, B., Biernath, C., Cammarano, D., Challinor, A., Deryng, D., De Sanctis, G., Doltra, J., Fereres, E., Folberth, C., Garcia-Vila, M., Gayler, S., Hoogenboom, G., Hunt, L. A., Izaurralde, R. C., Jabloun, M., Jones, C. D., Kersebaum, K. C., Kimball, B. A., Koehler, A. K., Kumar, S. N., Nendel, C., O'Leary, G. J., Olesen, J. E., Ottman, M. J., Palosuo, T., Prasad, P. V., Priesack, E., Pugh, T. A., Reynolds, M., Rezaei, E. E., Rötter, R. P., Schmid, E., Semenov, M. A., Shcherbak, I., Stehfest, E., Stöckle, C. O., Stratonovitch, P., Streck, T., Supit, I., Tao, F., Thorburn, P., Waha, K., Wall, G. W., Wang, E., White, J. W., Wolf, J., Zhao, Z., and Zhu, Y. (2016). Similar estimates of temperature impacts on global wheat yield by three independent methods. *Nature Climate Change*, 6(12):1130–1136.

Liu, B. Y., Nearing, M. A., Shi, P. J., and Jia, Z. W. (2000). Slope Length Effects on Soil Loss for Steep Slopes. *Soil Science Society of America Journal*, 64(5):1759–1763.

Liu, H., Kiesel, J., Hörmann, G., and Fohrer, N. (2011). Effects of DEM horizontal resolution and methods on calculating the slope length factor in gently rolling landscapes. *CATENA*, 87(3):368–375.

Liu, J., Fritz, S., van Wesenbeeck, C. F., Fuchs, M., You, L., Obersteiner, M., and Yang, H. (2008). A spatially explicit assessment of current and future hotspots of hunger in Sub-Saharan Africa in the context of global change. *Global and Planetary Change*, 64(3-4):222–235.

Liu, J., Williams, J. R., Zehnder, A. J. B., and Yang, H. (2007). GEPIC - modelling wheat yield and crop water productivity with high resolution on a global scale. *Agricultural Systems*, 94(2):478–493.

Liu, S. C., Fu, C., Shiu, C. J., Chen, J. P., and Wu, F. (2009). Temperature dependence of global precipitation extremes. *Geophysical Research Letters*, 36(17):2–5.

Lobell, D. B. (2014). *Climate change adaptation in crop production: Beware of illusions*, volume 3. Elsevier.

Lobell, D. B. and Asseng, S. (2017). yields under two emission scenarios emissions scenarios Comparing estimates of climate change impacts from process- based and statistical crop models. *Environmental Research Letters*, 12:1–12.

Lobell, D. B. and Burke, M. B. (2008). Why are agricultural impacts of climate change so uncertain? The importance of temperature relative to precipitation. *Environmental Research Letters*, 3(3):034007.

Lobell, D. B. and Burke, M. B. (2010). On the use of statistical models to predict crop yield responses to climate change. *Agricultural and Forest Meteorology*, 150(11):1443–1452.

Lobell, D. B., Burke, M. B., Tebaldi, C., Mastrandrea, M. D., Falcon, W. P., and Naylor, R. L. (2008). Prioritizing climate change adaptation needs for food security in 2030. *Science*, 319(5863):607–610.

Lobell, D. B. and Field, C. B. (2007). Global scale climate–crop yield relationships and the impacts of recent warming. *Environmental Research Letters*, 2(1):014002.

Lobell, D. B., Schlenker, W., and Costa-Roberts, J. (2011). Climate Trends and Global Crop Production Since 1980. *Science*, 333(6042):616–620.

Lobotka, V. (1955). Terraced fields in Slovakia (In Slovak: Terasove polia na Slovensku). *Agriculture* (*Polnohospodarstvo*), 2(6):539–549.

Loch, R. (1996). Using rill/interrill comparisons to infer likely responses of erosion to slope length: implications for land management. *Australian Journal of Soil Research*, 34:489–502.

Long, H. L., Heilig, G. K., Wang, J., Li, X. B., Luo, M., Wu, X. Q., and Zhang, M. (2006a). Land use and soil erosion in the upper reaches of the Yangtze River: Some socio-economic considerations on China's Grain-for-Green Programme. *Land Degradation and Development*, 17(6):589–603.

Long, S. P., Ainsworth, E. A., Leakey, A. D. B., Ort, D. R., and No, J. (2006b). Food for Thought : Lower-Than-Expected Crop Yield Stimulation with Rising CO2 Concentrations. *Science*, 312(June):1918–1922.

Loughran, R. J., Elliott, G. L., Campbell, B. L., and Shelly, D. J. (1988). Estimation of soil erosion from caesium-137 measurements in a small, cultivated catchment in Australia. *International Journal of Radiation Applications and Instrumentation. Part*, 39(11):1153–1157.

Luo, H., Wang, Q., Zhang, J., Wang, L., Li, Y., and Yang, G. (2020). Minimum fertilization at the appearance of the first flower benefits cotton nutrient utilization of nitrogen, phosphorus and potassium. *Scientific Reports*, 10(1):1–12.

Mabit, L., Chhem-Kieth, S., Dornhofer, P., Toloza, A., Benmansour, M., Bernard, C., Fulajtar, E., and Walling, D. (2014a). 137Cs: A widely used and validated medium-term soil tracer. In *Guidelines for using fallout radionuclides to assess erosion and effectiveness of soil conservation strategies. IAEA-TECDOC-1741.*, pages 27–78. IAEA, Vienna.

Mabit, L., Meusburger, K., Fulajtar, E., and Alewell, C. (2013). The usefulness of 137Cs as a tracer for soil erosion assessment: A critical reply to Parsons and Foster (2011). *Earth-Science Reviews*, 127:300–307.

Mabit, L., Zapata, F., Dercon, G., Benmansour, M., Bernard, C., and Walling, D. (2014b). Guidelines for using fallout radionuclides to assess erosion and effectiveness of soil conservation strategies. IAEA-TECDOC-1741. Technical report, IAEA, Vienna.

Maeda, E. E., Pellikka, P. K., Siljander, M., and Clark, B. J. (2010). Potential impacts of agricultural expansion and climate change on soil erosion in the Eastern Arc Mountains of Kenya. *Geomorphology*, 123(3):279–289.

Matthews, R. B., Rivington, M., Muhammed, S., Newton, A. C., and Hallett, P. D. (2013). Adapting crops and cropping systems to future climates to ensure food security: The role of crop modelling. *Global Food Security*, 2(1):24–28.

Mbow, C., Rosenzweig, C., Barioni, L., Benton, T., Herrero, M., Krishnapillai, M., Liwenga, E., Pradhan, P., Rivera-Ferre, M., Sapkota, T., Tubiello, F., and Xu, Y. (2019). Food Security. In Shukla, P., Skea, J., Calvo Buendia, E., Masson-Delmotte, V., Pörtner, H.-O., Roberts, D., Zhai, P., Slade, R., Connors, S., Van Diemen, R., Ferrat, M., Haughey, E., Luz, S., Neogi, S., Pathak, M., Petzold, J., Portugal Pereira, J., Vyas, P., Huntley, E., Kissick, K., Belkacemi, M., and Malley, J., editors, *Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems.*

McCool, D., Foster, G., Mutchler, C., and Meyer, L. (1989). Revised slope length factor for the Universal Soil Loss Equation. *Transactions of the ASAE*, 32:1571–1576.

Mearns, L. O., Mavromatis, T., Tsvetsinskaya, E., Hays, C., and Easterling, W. (1999). Comparative responses of EPIC and CERES crop models to high and low spatial resolution climate change scenarios. *Journal of Geophysical Research Atmospheres*, 104(D6):6623–6646.

Melorose, J., Perroy, R., and Careas, S. (2015). World population prospects. *United Nations*, 1(6042):587–92.

Merritt, W. S., Letcher, R. A., and Jakeman, A. J. (2003). A review of erosion and sediment transport models. *Environmental Modelling and Software*, 18(8-9):761–799.

Meyer, L. (1984). Evolution of the Universal Soil Loss Equation. *Journal of Soil and Water Conservation*, 39(2):99–104.

Mittler, R. (2006). Abiotic stress, the field environment and stress combination. *Trends Plant Sci.*, 11(1):16–9.

Mkwambisi, D., Gomani, M., and Kambani, C. (2010). Assessing the impact of climate change on agriculture and rural livelihoods in Malawi. Technical report, Environmental Affairs Department, Lilongwe. Malawi.

Mohamadi, M. A. and Kavian, A. (2015). Effects of rainfall patterns on runoff and soil erosion in field plots. *International Soil and Water Conservation Research*, 3(4):273–281.

Mondal, A., Khare, D., Kundu, S., Mukherjee, S., Mukhopadhyay, A., and Mondal, S. (2017). Uncertainty of soil erosion modelling using open source high resolution and aggregated DEMs. *Geoscience Frontiers*, 8(3):425–436.

Montanarella, L. (2007). Trends in Land Degradation in Europe. In Sivakumar, M. V. K. and Ndiang'ui, N., editors, *Climate and Land Degradation*, pages 83–104. Springer Berlin Heidelberg, Berlin, Heidelberg.

Montanarella, L., Pennock, D. J., McKenzie, N., Badraoui, M., Chude, V., Baptista, I., Mamo, T., Yemefack, M., Aulakh, M. S., Yagi, K., Hong, S. Y., Vijarnsorn, P., Zhang, G. L., Arrouays, D., Black, H., Krasilnikov, P., Sobocká, J., Alegre, J., Henriquez, C. R., Mendonça-Santos, M. d. L., Taboada, M., Espinosa-Victoria, D., AlShankiti, A., AlaviPanah, S. K., Mustafa Elsheikh, E. A. E., Hempel, J., Arbestain, M. C., Nachtergaele, F., and Vargas, R. (2016). World's soils are under threat. *Soil*, 2(1):79–82.

Montanarella, L., Scholes, R., and Brainich, A. (2018). The IPBES Assessment Report on Land Degradation and Restoration. Secretariat of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. Technical report, Bonn.

Monteith, J. (1965). Light distribution and photosynthesis in field crops. Annals of Botany, 29:17-37.

Monteith, J. (1977). Climate and the efficiency of crop production in Britain. *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences*, 281:277–294. Montgomery, D. R. (2007). Soil erosion and agricultural sustainability. *Proceedings of the National Academy of Sciences of the United States of America*, 104(33):13268–72.

Mora, C., Caldwell, I. R., Caldwell, J. M., Fisher, M. R., Genco, B. M., and Running, S. W. (2015). Suitable days for plant growth disappear under projected climate change: Potential human and biotic vulnerability. *PLoS Biology*, 13(6):1–15.

Morgan, R. (2005). Soil erosion and conservation. Blackwell Science Ltd, 3 edition.

Morgan, R. P. C., Quinton, J. N., Smith, R. E., Govers, G., Poesen, J. W. A., Auerswald, K., Chisci, G., Torri, D., and Styczen, M. E. (1998). The European soil erosion model (EUROSEM): a dynamic approach for predicting sediment transport from fields and small catchments. *Earth Surface Processes and Landforms*, 23(6):527–544.

Moss, R. H., Edmonds, J. A., Hibbard, K. A., Manning, M. R., Rose, S. K., Van Vuuren, D. P., Carter, T. R., Emori, S., Kainuma, M., Kram, T., Meehl, G. A., Mitchell, J. F., Nakicenovic, N., Riahi, K., Smith, S. J., Stouffer, R. J., Thomson, A. M., Weyant, J. P., and Wilbanks, T. J. (2010). The next generation of scenarios for climate change research and assessment. *Nature*, 463(7282):747–756.

Mueller, C., Elliott, J., Chryssanthacopoulos, J., Arneth, A., Balkovic, J., Ciais, P., Deryng, D., Folberth, C., Glotter, M., Hoek, S., Iizumi, T., Izaurralde, R. C., Jones, C., Khabarov, N., Lawrence, P., Liu, W., Olin, S., Pugh, T. A. M., Ray, D. K., Reddy, A., Rosenzweig, C., Ruane, A. C., Sakurai, G., Schmid, E., Skalsky, R., Song, C. X., Wang, X., De Wit, A., and Yang, H. (2017). Global gridded crop model evaluation: Benchmarking, skills, deficiencies and implications. *Geoscientific Model Development*, 10(4):1403–1422.

Mueller, N. D., Gerber, J. S., Johnston, M., Ray, D. K., Ramankutty, N., and Foley, J. A. (2012). Closing yield gaps through nutrient and water management. *Nature*, 494(7437):390–390.

Mukundan, R., Pradhanang, S. M., Schneiderman, E. M., Pierson, D. C., Anandhi, A., Zion, M. S., Matonse, A. H., Lounsbury, D. G., and Steenhuis, T. S. (2013). Suspended sediment source areas and future climate impact on soil erosion and sediment yield in a New York City water supply watershed, USA. *Geomorphology*, 183:110–119.

Mullan, D., Favis-Mortlock, D., and Fealy, R. (2012). Addressing key limitations associated with modelling soil erosion under the impacts of future climate change. *Agricultural and Forest Meteorology*, 156:18–30.

Müller, C., Cramer, W., Hare, W. L., and Lotze-Campen, H. (2011). Climate change risks for African agriculture. *Proceedings of the National Academy of Sciences of the United States of America*, 108(11):4313–4315.

Mutchler, C. K., Murphree, C. E., and McGregor, K. C. (1994). Laboratory and Field Plots for Erosion Research. In Lal, R., editor, *Soil Erosion Research Methods*, page 352. Routledge.

Myers, N. (1993). Gaia: An Atlas of Planet Management. Anchor/Doubleda, Garden City, NY.

Myers, S. S., Zanobetti, A., Kloog, I., Huybers, P., Leakey, A. D., Bloom, A. J., Carlisle, E., Dietterich, L. H., Fitzgerald, G., Hasegawa, T., Holbrook, N. M., Nelson, R. L., Ottman, M. J., Raboy, V., Sakai, H., Sartor, K. A., Schwartz, J., Seneweera, S., Tausz, M., and Usui, Y. (2014). Increasing CO2 threatens human nutrition. *Nature*, 510(7503):139–142.

Nachtergaele, F. O., Petri, M., Biancalani, R., van Lynden, G., van Velthuizen, H., and Bloise, M. (2011). Global Land Degradation Information System (GLADIS) - An Information database for Land Degradation Assessment at Global Level. *Management*, (September):110.

Naipal, V., Ciais, P., Wang, Y., Lauerwald, R., Guenet, B., and Van Oost, K. (2018). Global soil organic carbon removal by water erosion under climate change and land use change during AD-1850-2005. *Biogeosciences*, 15(14):4459–4480.

Naipal, V., Reick, C., Pongratz, J., and Van Oost, K. (2015). Improving the global applicability of the RUSLE model - Adjustment of the topographical and rainfall erosivity factors. *Geoscientific Model Development*, 8(9):2893–2913.

Nearing, M., Romkens, M., Norton, L., Stott, D., Rhoton, F., Laflen, J., Flanagan, D., Alonso, C., Binger, R., Dabney, S., Doering, O., Huang, C., McGregor, K., and Simon, A. (2000). Measurements and models of soil loss rates. *Science*, 290(5495):1300–1301.

Nearing, M. a., Pruski, F. F., and O'Neal, M. R. (2004). Expected climate change impacts on soil erosion rates: A review. *Journal of Soil and Water Conservation*, 59(1):43–50.

Neitsch, S., Arnold, J., Kiniry, J., Srinivasan, R., and Williams, J. (2011). Soil Water Assessment Tool Theoretical Documentation Version 2009. TWRI Report TR-192. Technical report, Texas Water Resources Institute.

Nelson, G. C., Valin, H., Sands, R. D., Havlík, P., Ahammad, H., Deryng, D., Elliott, J., Fujimori, S., Hasegawa, T., Heyhoe, E., Kyle, P., Von Lampe, M., Lotze-Campen, H., Mason D'Croz, D., Van Meijl, H., Van Der Mensbrugghe, D., Müller, C., Popp, A., Robertson, R., Robinson, S., Schmid, E., Schmitz, C., Tabeau, A., and Willenbockel, D. (2014). Climate change effects on agriculture: Economic responses to biophysical shocks. *Proceedings of the National Academy of Sciences of the United States of America*, 111(9):3274–3279.

Nelson, R., Dimes, J., Silburn, D., and Carberry, P. (1996). Erosion/ productivity modelling of maize farming in the Philippine uplands. Part I: a simple description of the agricultural production systems simulator. SEARCA-UQ Uplands Research Project. Working Paper No. 12.

Nendel, C., Wieland, R., Mirschel, W., Specka, X., Guddat, C., and Kersebaum, K. C. (2013). Simulating regional winter wheat yields using input data of different spatial resolution. *Field Crops Research*, 145:67–77. Nkonya, E., Gerber, N., Baumgartner, P., Braun, J. V., Pinto, A. D., Graw, V., Kato, E., Kloos, J., and Walter, T. (2011). *The Economics of Desertification, Land Degradation, and Drought Toward an Integrated Global Assessment*, volume 37 of *Development economics and policy*. Lang.

Norse, D., Li, J., Jin, L., and Zhang, Z. (2001). *Environmental Costs of Rice production in China: Lessons from Hunan and Hubei*. Aileen International Press.

Nossent, J., Elsen, P., and Bauwens, W. (2011). Sobol' sensitivity analysis of a complex environmental model. *Environmental Modelling and Software*, 26(12):1515–1525.

Nunes, J. P., Seixas, J., and Keizer, J. J. (2013). Modeling the response of within-storm runoff and erosion dynamics to climate change in two Mediterranean watersheds: A multi-model, multi-scale approach to scenario design and analysis. *Catena*, 102:27–39.

Nyssen, J., Frankl, A., Zenebe, A., Deckers, J., and Poesen, J. (2015). Land Management in the Northern Ethiopian Highlands: Local and Global Perspectives; Past, Present and Future. *Land Degradation and Development*, 26(7):759–764.

Nyssen, J., Tielens, S., Gebreyohannes, T., Araya, T., Teka, K., van de Wauw, J., Degeyndt, K., Descheemaeker, K., Amare, K., Haile, M., Zenebe, A., Munro, N., Walraevens, K., Gebrehiwot, K., Poesen, J., Frankl, A., Tsegay, A., and Deckers, J. (2019). *Understanding spatial patterns of soils for sustainable agriculture in northern Ethiopia's tropical mountains*, volume 14.

Oldeman, L. (1994). The global extent of soil degradation.

Oldeman, L., Hakkeling, R., and Sombroek, G. (1991). Status of Human-Induced Soil Degradation 2nd ed. Technical report, International Soil Reference and Information Center, Wageningen.

Oliveira, P. T. S., Wendland, E., and Nearing, M. A. (2013). Rainfall erosivity in Brazil: A review. *Catena*, 100:139–147.

Olsson, L., Barbosa, H., Bhadwal, S., Cowie, A., Delusca, K., Flores-Renteria, D., Hermans, K., Jobbagy, E., Kurz, W., Li, D., Sonwa, D., and Stringer, L. (2019). Land Degradation. In Shukla, R., Skea, J., Buendia, E. C., Masson-Delmotte, V., Pörtner, H.-O., Roberts, D. C., Zhai, P., Slade, R., Connors, S., van Diemen, R., Ferrat, M., Haughey, E., Luz, S., Neogi, S., Pathak, M., Petzold, J., Pereira, J. P., Vyas, P., Huntley, E., Kissick, K., M.Belkacemi, and Malley, J., editors, *Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems.*

O'Neal, M. R., Nearing, M. A., Vining, R. C., Southworth, J., and Pfeifer, R. A. (2005). Climate change impacts on soil erosion in Midwest United States with changes in crop management. *Catena*, 61(2-3 SPEC. ISS.):165–184.

Onstad, C. and Foster, G. (1975). Erosion modeling on a watershed. *Transactions of the ASAE*, 18:288–292.

Oreskes, N., Shrader-Frechette, K., and Belitz, K. (1994). Verification, Validation, and Confirmation of Numerical Models in the Earth Sciences. *Science*, 263(5147):641–646.

Osbahr, H., Twyman, C., Adger, W. N., and Thomas, D. S. (2010). Evaluating Successful Livelihood Adaptation to Climate Variability and change in Southern Africa. *Ecology and Society*, 15(2):22.

Osborne, T., Rose, G., and Wheeler, T. (2013). Variation in the global-scale impacts of climate change on crop productivity due to climate model uncertainty and adaptation. *Agricultural and Forest Meteorology*, 170:183–194.

Osborne, T. M., Lawrence, D. M., Challinor, A. J., Slingo, J. M., and Wheeler, T. R. (2007). Development and assessment of a coupled crop-climate model. *Global Change Biology*, 13(1):169–183.

Oyedele, D. J. and Aina, P. O. (1998). A study of soil factors in relation to erosion and yield of maize on a Nigerian soil. *Soil and Tillage Research*, 48(1-2):115–125.

Panagos, P., Borrelli, P., Matthews, F., Liakos, L., Bezak, N., Diodato, N., and Ballabio, C. (2022). Global rainfall erosivity projections for 2050 and 2070. *Journal of Hydrology*, 610(September 2021):127865.

Panagos, P., Borrelli, P., Meusburger, K., van der Zanden, E. H., Poesen, J., and Alewell, C. (2015). Modelling the effect of support practices (P-factor) on the reduction of soil erosion by water at European scale. *Environmental Science and Policy*, 51:23–34.

Panagos, P., Borrelli, P., Meusburger, K., Yu, B., Klik, A., Lim, K. J., Yang, J. E., Ni, J., Miao, C., Chattopadhyay, N., Sadeghi, S. H., Hazbavi, Z., Zabihi, M., Larionov, G. A., Krasnov, S. F., Gorobets, A. V., Levi, Y., Erpul, G., Birkel, C., Hoyos, N., Naipal, V., Oliveira, P. T. S., Bonilla, C. A., Meddi, M., Nel, W., Dashti, H. A., Boni, M., Diodato, N., Van Oost, K., Nearing, M., and Ballabio, C. (2017). Global rainfall erosivity assessment based on high-temporal resolution rainfall records. *Scientific Reports*, 7(June):1–12.

Panagos, P., Borrelli, P., Poesen, J., Meusburger, K., Ballabio, C., Lugato, E., Montanarella, L., and Alewell, C. (2016). Reply to "The new assessment of soil loss by water erosion in Europe. Panagos P. et al., 2015 Environ. Sci. Policy 54, 438-447-A response" by Evans and Boardman [Environ. Sci. Policy 58, 11-15]. *Environmental Science and Policy*, 59:53–57.

Panagos, P., Standardi, G., Borrelli, P., Lugato, E., Montanarella, L., and Bosello, F. (2018). Cost of agricultural productivity loss due to soil erosion in the European Union: From direct cost evaluation approaches to the use of macroeconomic models. *Land Degradation and Development*, 29(3):471–484.

Pannell, D. J., Llewellyn, R. S., and Corbeels, M. (2014). The farm-level economics of conservation agriculture for resource-poor farmers. *Agriculture, Ecosystems and Environment*, 187:52–64.

Paroissien, J.-B., Darboux, F., Couturier, A., Devillers, B., Mouillot, F., Raclot, D., and Le Bissonnais, Y. (2015). A method for modeling the effects of climate and land use changes on erosion and sustainability of soil in a Mediterranean watershed (Languedoc, France). *Journal of Environmental Management*, 150:57–68.

Parsons, A. J. (2019). How reliable are our methods for estimating soil erosion by water? *Science of the Total Environment*, 676:215–221.

Parsons, A. J., Brazier, R. E., Wainwright, J., and Powell, D. M. (2006). Scale relationships in hillslope runoff and erosion. *Earth Surface Processes and Landforms*, 31(11):1384–1393.

Parsons, A. J. and Foster, I. D. (2011). What can we learn about soil erosion from the use of 137Cs? *Earth-Science Reviews*, 108(1-2):101–113.

Parsons, A. J., Wainwright, J., Powell, D. M., Kaduk, J., and Brazier, R. E. (2004). A conceptual model for determining soil erosion by water. *Earth Surface Processes and Landforms*, 29(10):1293–1302.

Parton, W. J., Schimel, D. S., Cole, C. V., and Ojima, D. S. (1987). Analysis of Factors Controlling Soil Organic Matter Levels in Great Plains Grasslands. *Soil Science Society of America Journal*, 51(5):1173–1179.

Patt, A. G., van Vuuren, D. P., Berkhout, F., Aaheim, A., Hof, A. F., Isaac, M., and Mechler, R. (2010). Adaptation in integrated assessment modeling: Where do we stand? *Climatic Change*, 99(3):383–402.

Peel, M. C., Finlayson, B. L., and McMahon, T. A. (2007). Updated world map of the Koeppen-Geiger climate classification. *Hydrol. Earth Syst. Sci.*, 11:1633–1644.

Pelletier, J. D., Broxton, P. D., Hazenberg, P., Zeng, X., Troch, P. A., Niu, G.-Y., Williams, Z., Brunke, M. A., and Gochis, D. (2016). A gridded global data set of soil, intact regolith, and sedimentary deposit thicknesses for regional and global land surface modeling. *Journal of Advances in Modeling Earth Systems*, 8(1):41–65.

Pendergrass, A. G. (2018). What precipitation is extreme? Science, 360(6393):1072-1073.

Pendergrass, A. G., Knutti, R., Lehner, F., Deser, C., and Sanderson, B. M. (2017). Precipitation variability increases in a warmer climate. *Scientific Reports*, 7(1):1–9.

Peters, G. P. and Hausfather, Z. (2020). Emissions - the 'business as usual' story is misleading. *Nature*, 577:618–620.

Pimentel, D. (2006). Soil erosion: A food and environmental threat. *Environment, Development and Sustainability*, 8(1):119–137.

Pimentel, D., Harvey, C., Resosudarmo, P., Sinclair, K., Kurz, D., McNair, M., Crist, S., Shpritz, L., Fitton, L., Saffouri, R., and Blair, R. (1995). Environmental and economic costs of soil erosion and conservation benefits.

Poeplau, C. and Don, A. (2015). Carbon sequestration in agricultural soils via cultivation of cover crops
A meta-analysis. *Agriculture, Ecosystems and Environment*, 200:33–41.

Poesen, J., Nachtergaele, J., Verstraeten, G., and Valentin, C. (2003). Gully erosion and environmental change: Importance and research needs. *Catena*, 50(2-4):91–133.

Polyakov, V. and Lal, R. (2004). Modeling soil organic matter dynamics as affected by soil water erosion. *Environment International*, 30(4):547–556.

Ponzi, D. (1993). Soil erosion and productivity: A brief review. Desertification Bull. 22, 22:36-44.

Porter, J. and Megan, G. (1999). Temperatures and the growth and development of wheat: a review. *European Journal of Agronomy*, 10:23–36.

Portmann, F. T., Siebert, S., and Döll, P. (2010). MIRCA2000—Global monthly irrigated and rainfed crop areas around the year 2000: A new high-resolution data set for agricultural and hydrological modeling. *Global Biogeochemical Cycles*, 24(1).

Porwollik, V., Müller, C., Elliott, J., Chryssanthacopoulos, J., Iizumi, T., Ray, D. K., Ruane, A. C., Arneth, A., Balkovič, J., Ciais, P., Deryng, D., Folberth, C., Izaurralde, R. C., Jones, C. D., Khabarov, N., Lawrence, P. J., Liu, W., Pugh, T. A. M., Reddy, A., Sakurai, G., Schmid, E., Wang, X., de Wit, A., and Wu, X. (2017). Spatial and temporal uncertainty of crop yield aggregations. *European Journal of Agronomy*, 88:10–21.

Porwollik, V., Rolinski, S., Heinke, J., and Müller, C. (2019). Generating a rule-based global gridded tillage dataset. *Earth System Science Data*, 11(2):823–843.

Poulton, P., Johnston, J., Macdonald, A., White, R., and Powlson, D. (2018). Major limitations to achieving "4 per 1000" increases in soil organic carbon stock in temperate regions: Evidence from long-term experiments at Rothamsted Research, United Kingdom. *Global Change Biology*, 24(6):2563–2584.

Prasuhn, V., Liniger, H., Gisler, S., Herweg, K., Candinas, A., and Clément, J. P. (2013). A high-resolution soil erosion risk map of Switzerland as strategic policy support system. *Land Use Policy*, 32:281–291.

Pravalie, R., Patriche, C., Borrelli, P., Panagos, P., Roșca, B., Dumitrascu, M., Nita, I.-A., Savulescu, I., Birsan, M.-V., and Bandoc, G. (2021). Arable lands under the pressure of multiple land degradation processes . A global perspective. *Environmental Research*, 194(September 2020).

Prosser, I. P., Rutherfurd, I. D., Olley, J. M., Young, W. J., Wallbrink, P. J., and Moran, C. J. (2001). Large-scale patterns of erosion and sediment transport in river networks, with examples from Australia. *Marine and Freshwater Research*, 52(1):81–99.

Pruski, F. F. and Nearing, M. A. (2002a). Climate-induced changes in erosion during the 21st century for eight U.S. locations. *Water Resources Research*, 38(12):34–1–34–11.

Pruski, F. F. and Nearing, M. A. (2002b). Runoff and soil-loss responses to changes in precipitation: A computer simulation study. *Journal of Soil and Water Conservation*, 57(1):7–16.

Quinn, C. H., Ziervogel, G., Taylor, A., Takama, T., and Thomalla, F. (2011). Coping with multiple stresses in rural South Africa. *Ecology and Society*, 16(3):10.

Rabbinge, R. (1993). The ecological background of food production. In *Ciba Foundation Symposium* 177-Crop Protection and Sustainable Agriculture, number 177, pages 2–29.

Rallison, R. (1980). Origin and Evolution of the SCS Runoff Equation. In *Proceeding of the Symposium* on Watershed Management '80 American Society of Civil Engineering Boise ID.

Ramirez-Cabral, N. Y., Kumar, L., and Shabani, F. (2017). Global alterations in areas of suitability for maize production from climate change and using a mechanistic species distribution model (CLIMEX). *Scientific Reports*, 7(1):1–13.

Ravishankara, A. R., Daniel, J. S., and Portmann, R. W. (2009). Nitrous oxide (N2O): The dominant ozone-depleting substance emitted in the 21st century. *Science*, 326(5949):123–125.

Ray, D. K., Gerber, J. S., MacDonald, G. K., and West, P. C. (2015). Climate variation explains a third of global crop yield variability. *Nature Communications*, 6:5989.

Reed, M. S. and Stringer, L. C. (2016). Land Degradation, Desertification and Climate Change. Anticipating, assessing and adapting to future change. Routledge.

Ren, X., Weitzel, M., O'Neill, B. C., Lawrence, P., Meiyappan, P., Levis, S., Balistreri, E. J., and Dalton, M. (2018). Avoided economic impacts of climate change on agriculture: integrating a land surface model (CLM) with a global economic model (iPETS). *Climatic Change*, 146(3-4):517–531.

Renard, K., Foster, G., Weesies, G., McCool, D., and Yoder, D. (1997). Predicting soil erosion by water: a guide to conservation planning with the Revised Universal Soil Loss Equation (RUSLE).

Renwick, A., Jansson, T., Verburg, P. H., Revoredo-Giha, C., Britz, W., Gocht, A., and McCracken, D. (2013). Policy reform and agricultural land abandonment in the EU. *Land Use Policy*, 30(1):446–457.

Reyer, C. P., Adams, S., Albrecht, T., Baarsch, F., Boit, A., Canales Trujillo, N., Cartsburg, M., Coumou, D., Eden, A., Fernandes, E., Langerwisch, F., Marcus, R., Mengel, M., Mira-Salama, D., Perette, M., Pereznieto, P., Rammig, A., Reinhardt, J., Robinson, A., Rocha, M., Sakschewski, B., Schaeffer, M., Schleussner, C. F., Serdeczny, O., and Thonicke, K. (2017). Climate change impacts in Latin America and the Caribbean and their implications for development. *Regional Environmental Change*, 17(6):1601–1621.

Riahi, K., Grübler, A., and Nakicenovic, N. (2007). Scenarios of long-term socio-economic and environmental development under climate stabilization. *Technological Forecasting and Social Change*, 74(7):887–935.

Risse, L. M., Nearing, M. A., Laflen, J. M., and Nicks, A. D. (1993). Error Assessment in the Universal Soil Loss Equation. *Soil Science Society of America Journal*, 57(3):825–833.

Ritchie, J. T. and Alagarswamy, G. (2002). Overview of Crop Models for Assessment of Crop Production. In Doering, O. C., Randolph, J. C., Southworth, J., and Pfeifer, R. A., editors, *Effects of Climate Change and Variability on Agricultural Production Systems*, pages 43–68. Springer US, Boston, MA.

Ritchie, J. T. and Nesmith, D. S. (1991). Temperature and Crop Development. In *Modeling Plant and Soil Systems*, chapter 2, pages 5–29. John Wiley Sons, Ltd.

Rogelj, J., Den Elzen, M., Höhne, N., Fransen, T., Fekete, H., Winkler, H., Schaeffer, R., Sha, F., Riahi, K., and Meinshausen, M. (2016). Paris Agreement climate proposals need a boost to keep warming well below 2 °c. *Nature*, 534(7609):631–639.

Romeo, R., Vita, A., Manuelli, S., Zanini, E., Freppaz, M., and Stanchi, S. (2015). Understanding Mountain Soils: A contribution from mountain areas to the International Year of Soils 2015. Technical report, FAO, Rome.

Roose, E. (1996). *Land husbandry - Components and strategy.* 70 FAO soils bulletin. Food and Agriculture Organization of the United Nations, Rome.

Rosegrant, M. W., Ringler, C., and Zhu, T. (2009). Water for agriculture: Maintaining food security under growing scarcity. *Annual Review of Environment and Resources*, 34:205–222.

Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A. C., Müller, C., Arneth, A., Boote, K. J., Folberth, C., Glotter, M., Khabarov, N., Neumann, K., Piontek, F., Pugh, T. A. M., Schmid, E., Stehfest, E., Yang, H., and Jones, J. W. (2014). Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. *Proceedings of the National Academy of Sciences*, 111(9):3268–3273.

Rosenzweig, C., Jones, J. W., Hatfield, J. L., Ruane, A. C., Boote, K. J., Thorburn, P., Antle, J. M., Nelson, G. C., Porter, C., Janssen, S., Asseng, S., Basso, B., Ewert, F., Wallach, D., Baigorria, G., and Winter, J. M. (2013). The Agricultural Model Intercomparison and Improvement Project (AgMIP): Protocols and pilot studies. *Agricultural and Forest Meteorology*, 170:166–182.

Rosenzweig, C., Tubiello, F. N., Goldberg, R., Mills, E., and Bloomfield, J. (2002). Increased crop damage in the US from excess precipitation under climate change. *Global Environmental Change*, 12(3):197–202.

Rötter, R. P., Carter, T. R., Olesen, J. E., and Porter, J. R. (2011). Crop-climate models need an overhaul. *Nature Climate Change*, 1(4):175–177.

Routschek, A., Schmidt, J., and Kreienkamp, F. (2014). Impact of climate change on soil erosion - A high-resolution projection on catchment scale until 2100 in Saxony/Germany. *Catena*, 121:99–109.

Ruane, A. C., Goldberg, R., and Chryssanthacopoulos, J. (2015). Climate forcing datasets for agricultural modeling: Merged products for gap-filling and historical climate series estimation. *Agricultural and Forest Meteorology*, 200:233–248. Rüttimann, M., Schaub, D., Prasuhn, V., and Rüegg, W. (1995). Measurement of runoff and soil erosion on regularly cultivated fields in Switzerland - some critical considerations. *Catena*, 25(1-4):127–139.

Sacks, W. J., Deryng, D., Foley, J. A., and Ramankutty, N. (2010). Crop planting dates: An analysis of global patterns. *Global Ecology and Biogeography*, 19(5):607–620.

Sadeghi, S. H. and Mizuyama, T. (2007). Applicability of the Modified Universal Soil Loss Equation for prediction of sediment yield in Khanmirza watershed, Iran. *Hydrological Sciences Journal*, 52(5):1068–1075.

Sánchez, B., Rasmussen, A., and Porter, J. R. (2014). Temperatures and the growth and development of maize and rice: A review. *Global Change Biology*, 20(2):408–417.

Sánchez, P. A. (2010). Tripling crop yields in tropical Africa. Nature Geoscience, 3(5):299-300.

Sanderman, J., Hengl, T., and Fiske, G. J. (2017). Erratum: Soil carbon debt of 12,000 years of human land use (Proc Natl Acad Sci USA (2018)114 (9575–9580) DOI: 10.1073/pnas.1706103114). *Proceedings of the National Academy of Sciences of the United States of America*, 115(7):E1700.

Saraphirom, P., Wirojanagud, W., and Srisuk, K. (2013). Impact of climate change on waterlogging and salinity distributions in Huai Khamrian subwatershed, NE Thailand. *Environmental Earth Sciences*, 70(2):887–900.

Sartori, M., Philippidis, G., Ferrari, E., Borrelli, P., Lugato, E., Montanarella, L., and Panagos, P. (2019). A linkage between the biophysical and the economic: Assessing the global market impacts of soil erosion. *Land Use Policy*, 86(December 2018):299–312.

Schauberger, B., Archontoulis, S., Arneth, A., Balkovic, J., Ciais, P., Deryng, D., Elliott, J., Folberth,
C., Khabarov, N., Müller, C., Pugh, T. A., Rolinski, S., Schaphoff, S., Schmid, E., Wang, X., Schlenker,
W., and Frieler, K. (2017). Consistent negative response of US crops to high temperatures in observations and crop models. *Nature Communications*, 8.

Scherer, L. and Pfister, S. (2015). Modelling spatially explicit impacts from phosphorus emissions in agriculture. *International Journal of Life Cycle Assessment*, 20(6):785–795.

Schertz, D. and Nearing, M. A. (2006). Erosion Tolerance / Soil Loss Tolerance. In Lal, R., editor, *Encyclopedia of Soil Science*, pages 640–624.

Schleussner, C. F., Rogelj, J., Schaeffer, M., Lissner, T., Licker, R., Fischer, E. M., Knutti, R., Levermann, A., Frieler, K., and Hare, W. (2016). Science and policy characteristics of the Paris Agreement temperature goal. *Nature Climate Change*, 6(9):827–835.

Shaffer, M. J., Gupta, S. C., Linden, D. R., Molina, J. A. E., Clapp, C. E., and Larson, W. E. (1983).
Simulation of Nitrogen, Tillage, and Residue Management Effects on Soil Fertility. In LAUENROTH,
W. K., SKOGERBOE, G. V., and FLUG, M., editors, *Analysis of Ecological Systems*, volume 5 of *Developments in Environmental Modelling*, pages 525–544. Elsevier.

Sharpley, A. and Williams, J. R. (1990). EPIC — Erosion / Productivity Impact Calculator: 1. Model Documentation. *U.S. Department of Agriculture Technical Bulletin*, 1768:235.

Sharratt, B. S., Tatarko, J., Abatzoglou, J. T., Fox, F. A., and Huggins, D. (2015). Implications of climate change on wind erosion of agricultural lands in the Columbia plateau. *Weather and Climate Extremes*, 10:20–31.

Shen, Z., Chen, L., Liao, Q., Liu, R., and Hong, Q. (2012). Impact of spatial rainfall variability on hydrology and nonpoint source pollution modeling. *Journal of Hydrology*, 472-473:205–215.

Siebert, S., Kummu, M., Porkka, M., Döll, P., Ramankutty, N., and Scanlon, B. R. (2015). A global data set of the extent of irrigated land from 1900 to 2005. *Hydrology and Earth System Sciences*, 19(3):1521–1545.

Sivakumar, M. V., Das, H. P., and Brunini, O. (2005). Impacts of present and future climate variability and change on agriculture and forestry in the arid and semi-arid tropics. *Increasing Climate Variability and Change: Reducing the Vulnerability of Agriculture and Forestry*, pages 31–72.

Skalský, R., Tarasovičová, Z., Balkovič, J., Schmid, E., Fuchs, M., Moltchanova, E., Kindermann, G., and Scholtz, P. (2008). GEO-BENE global database for bio-physical modeling. GEOBENE project. Technical report.

Slingo, J. M., Challinor, A. J., Hoskins, B. J., and Wheeler, T. R. (2005). Introduction: Food crops in a changing climate. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 360(1463):1983–1989.

Smith, P., House, J. I., Bustamante, M., Sobocká, J., Harper, R., Pan, G., West, P. C., Clark, J. M.,
Adhya, T., Rumpel, C., Paustian, K., Kuikman, P., Cotrufo, M. F., Elliott, J. A., Mcdowell, R., Griffiths,
R. I., Asakawa, S., Bondeau, A., Jain, A. K., Meersmans, J., and Pugh, T. A. (2016). Global change pressures on soils from land use and management. *Global Change Biology*, 22(3):1008–1028.

Smith, R. E. and Goodrich, D. C. (2006). Rainfall Excess Overland Flow. In *Encyclopedia of Hydrological Sciences*, chapter 111. American Cancer Society.

Sobol, I. (1990). On sensitivity estimation for nonlinear mathematical models. *Matem. Mod.*, 2:112–118.

Sorooshian, S. (1991). Parameter Estimation, Model Identification, and Model Validation: Conceptual-Type Models. In Bowles, D. S. and O'Connell, P. E., editors, *Recent Advances in the Modeling of Hydrologic Systems*, pages 443–467. Springer Netherlands, Dordrecht.

Southworth, J., Pfeifer, R., and Habeck, M. (2002a). Crop modeling results under climate change for the upper Midwestern United States. In Doering, O., Randolph, J., Southworth, J., and Pfeifer, R., editors, *Effects of Climate Change and Variability on Agricultural Production Systems*. Kluwer Academic Publishers, Boston.

Southworth, J., Pfeifer, R. A., Habeck, M., Randolph, J. C., Doering, O. C., Johnston, J. J., and Rao, D. G. (2002b). Changes in soybean yields in the Midwestern United States as a result of future changes in climate, climate variability, and CO2 fertilization. *Climatic Change*, 53(4):447–475.

Southworth, J., Randolph, J. C., Habeck, M., Doering, O. C., Pfeifer, R. A., Rao, D. G., and Johnston, J. J. (2000). Consequences of future climate change and changing climate variability on maize yields in the midwestern United States. *Agriculture, Ecosystems and Environment*, 82(1-3):139–158.

Squire, G. R., Hawes, C., Valentine, T. A., and Young, M. W. (2015). Degradation rate of soil function varies with trajectory of agricultural intensification. *Agriculture, Ecosystems and Environment*, 202:160–167.

Stehfest, E., Heistermann, M., Priess, J. A., Ojima, D. S., and Alcamo, J. (2007). Simulation of global crop production with the ecosystem model DayCent. *Ecological Modelling*, 209(2):203–219.

Stocker, T., Dahe, Q., and Plattner, G.-K. (2013). Climate change 2013: the physical science basis. In: Working Group I Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Summary for Policymakers (IPCC, 2013). Technical report.

Stringer, L., Twyman, C., and Thomas, D. (2007). Learning to reduce degradation on Swaziland's arable land: Enhancing understandings of Striga asiatica. *Land Degradation and Development*, 18:163–177.

Stroosnijder, L. (2005). Measurement of erosion: Is it possible? Catena, 64(2-3):162-173.

Su, L., Wang, Q., and Bai, Y. (2013). An analysis of yearly trends in growing degree days and the relationship between growing degree day values and reference evapotranspiration in Turpan area, China. *Theoretical and Applied Climatology*, 113(3):711–724.

Tabari, H., Hosseinzadehtalaei, P., Aghakouchak, A., and Willems, P. (2019). Latitudinal heterogeneity and hotspots of uncertainty in projected extreme precipitation. *Environmental Research Letters*, 14(12).

Tanner, C. B. and Sinclair, T. R. (1983). Efficient Water Use in Crop Production: Research or Re-Search? In *Limitations to Efficient Water Use in Crop Production*, chapter 1A, pages 1–27. John Wiley Sons, Ltd.

Tasser, E., Walde, J., Tappeiner, U., Teutsch, A., and Noggler, W. (2007). Land-use changes and natural reforestation in the Eastern Central Alps. *Agriculture, Ecosystems and Environment*, 118(1-4):115–129.

Tebaldi, C. and Lobell, D. (2018). Estimated impacts of emission reductions on wheat and maize crops. *Climatic Change*, 146(3-4):533–545.

Telles, T. S., Guimarães, M. d. F., and Dechen, S. C. F. (2011). The costs of soil erosion. *Rev. Bras. Ciênc. Solo*, 35(195).

Terranova, O., Antronico, L., Coscarelli, R., and Iaquinta, P. (2009). Soil erosion risk scenarios in the Mediterranean environment using RUSLE and GIS: An application model for Calabria (southern Italy). *Geomorphology*, 112(3-4):228–245.

Tiffen, M., Mortimore, M., and Gichuki, F. (1994). *More people, less erosion: environmental recovery in Kenya.* J. Wiley.

Tilman, D., Fargione, J., Wolff, B., D'Antonio, C., Dobson, A., Howarth, R., Schindler, D., Schlesinger,
W. H., Simberloff, D., and Swackhamer, D. (2001). Forecasting agriculturally driven global environmental change. *Science*, 292(5515):281–284.

Tollefson, J. (2020). How hot will earth get by 2100? Nature, 250:10.

Trenberth, K. E. (2011). Changes in precipitation with climate change. *Climate Research*, 47(1-2):123–138.

Trimble, S. W. (1999). Decreased Rates of Alluvial Sediment Storage in the Coon Creek Basin, Wisconsin, 1975-93. *Science*, 285(5431):1244–1246.

Trimble, S. W. and Crosson, P. (2000). U.S. Soil Erosion Rates–Myth and Reality. *Science*, 289(5477):248–250.

Tubiello, F. N. and Ewert, F. (2002). Simulating the effects of elevated CO2 on crops: Approaches and applications for climate change. *European Journal of Agronomy*, 18(1-2):57–74.

Turkelboom, F., Poesen, J., and Trébuil, G. (2008). The multiple land degradation effects caused by land-use intensification in tropical steeplands: A catchment study from northern Thailand. *Catena*, 75(1):102–116.

UN (1999). Standard Country or Area Codes for Statistical Use (Revision 4). Series M, No. 49/Rev.4. Technical report, United Nations, New York.

UN (2020). United Nations, Department of Economic and Social Affairs, Population Division (2019). World Population Prospects 2019, Online Edition. Rev. 1.

UNCCD (2017). The Global Land Outlook, first edition. Technical report, Bonn, Germany.

USDA (1972). National Engineering Handbook, Section 4, Hydrology. Technical report, USDA Soil Conservation Service, Washington DC.

USDA-ARC (2013). Science documentation. Revised Universal Soil Loss Equation, Version 2 (RUSLE 2). Technical report, USDA-ARC, Washington, D.C.

USGS (1997). USGS 30 ARC-second Global Elevation Data, GTOPO30.

Våje, P., Singh, B., and Lal, R. (2005). Soil Erosion and Nutrient Losses from a Volcanic Ash Soil in Kilimanjaro Region, Tanzania. *J. Sustain. Agr.*, 26(4):23–42.

Valentin, C., Agus, F., Alamban, R., Boosaner, A., Bricquet, J. P., Chaplot, V., de Guzman, T., de Rouw, A., Janeau, J. L., Orange, D., Phachomphonh, K., Do Duy Phai, Podwojewski, P., Ribolzi, O., Silvera, N., Subagyono, K., Thiébaux, J. P., Tran Duc Toan, and Vadari, T. (2008). Runoff and sediment losses from 27 upland catchments in Southeast Asia: Impact of rapid land use changes and conservation practices. *Agriculture, Ecosystems and Environment*, 128(4):225–238.

Valin, H., Sands, R. D., van der Mensbrugghe, D., Nelson, G. C., Ahammad, H., Blanc, E., Bodirsky, B., Fujimori, S., Hasegawa, T., Havlik, P., Heyhoe, E., Kyle, P., Mason-D'Croz, D., Paltsev, S., Rolinski, S., Tabeau, A., van Meijl, H., von Lampe, M., and Willenbockel, D. (2014). The future of food demand: understanding differences in global economic models. *Agricultural Economics*, 45:n/a–n/a.

van den Born, G., de Haan, B., Pearche, D., and Howarth, A. (2000). Technical Report on Soil Degradation in Europe: an integrated economic and environmental assessment. Technical Report May, National institute of public health and the environment.

Van Dijk, A. I., Bruijnzeel, L. A., and Rosewell, C. J. (2002). Rainfall intensity-kinetic energy relationships: A critical literature appraisal. *Journal of Hydrology*, 261(1-4):1–23.

Van Ittersum, M. K., Cassman, K. G., Grassini, P., Wolf, J., Tittonell, P., and Hochman, Z. (2013). Yield gap analysis with local to global relevance-A review. *Field Crops Research*, 143:4–17.

Van Oost, K., Quine, T. A., Govers, G., Gryze, S. D., Six, J., Harden, J. W., Mccarty, G. W., Heckrath, G., Kosmas, C., Giraldez, J. V., and Silva, J. R. M. (2007). The Impact of Agricultural Soil Erosion on the Global Carbon Cycle. *Science*, 318(5850):626–629.

van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., Hurtt, G. C., Kram, T., Krey, V., Lamarque, J. F., Masui, T., Meinshausen, M., Nakicenovic, N., Smith, S. J., and Rose, S. K. (2011). The representative concentration pathways: An overview. *Climatic Change*, 109(1):5–31.

Vandekerckhove, L., Poesen, J., Wijdenes, D. O., Nachtergaele, J., Kosmas, C., Roxo, M. J., and De Figueiredo, T. (2000). Thresholds for gully initiation and sedimentation in Mediterranean Europe. *Earth Surface Processes and Landforms*, 25(11):1201–1220.

Vanuytrecht, E., Raes, D., and Willems, P. (2016). Regional and global climate projections increase mid-century yield variability and crop productivity in Belgium. *Regional Environmental Change*, 16(3):659–672.

Verdin, J., Funk, C., Senay, G., and Choularton, R. (2005). Climate science and famine early warning. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 360(1463):2155–2168.

Verheijen, F. G., Jones, R. J., Rickson, R. J., and Smith, C. J. (2009). Tolerable versus actual soil erosion rates in Europe. *Earth-Science Reviews*, 94(1-4):23–38.

Vitousek, P. M., Mooney, H. A., Lubchenco, J., and Melillo, J. M. (1997). Human domination of Earth's ecosystems. *Science*, 227(5325):494–499.

Vogt, J. V., Safriel, U., Von Maltitz, G., Sokona, Y., Zougmore, R., Bastin, G., and Hill, J. (2011). Monitoring and assessment of land degradation and desertification: Towards new conceptual and integrated approaches. *Land Degradation and Development*, 22(2):150–165.

von Grebmer, K., Ringler, C., Rosegrant, M. W., Olofinbiyi, T., Wiesmann, D., Fritschel, H., Badiane, O., Torero, M., Yohannes, Y., Thompson, J., von Oppeln, C., and Rahall, J. (2012). 2012 Global hunger index: The challenege of hunger: Ensuring sustainable food security under land, water, and energy stresses. Technical report, Deutsche Welthungerhilfe (German AgroAction); International Food Policy Research Institute (IFPRI); Concern Worldwide, Bonn; Dublin; Washington, D.C.

Waha, K., Krummenauer, L., Adams, S., Aich, V., Baarsch, F., Coumou, D., Fader, M., Hoff, H., Jobbins, G., Marcus, R., Mengel, M., Otto, I. M., Perrette, M., Rocha, M., Robinson, A., and Schleussner, C. F. (2017). Climate change impacts in the Middle East and Northern Africa (MENA) region and their implications for vulnerable population groups. *Regional Environmental Change*, 17(6):1623–1638.

Walling, D., He, Q., and Zhang, Y. (2014). Conversion Models And Related Software. In *Guidelines* for Using Fallout Radionuclides to Assess Erosion and Effectiveness of Soil Conservation Strategies. IAEA, Vienna.

Walling, D. and Webb, B. W. (1983). Patterns of sediment yield. *Background to Palaeohydrology*, pages 69–100.

Walling, D. E. (1999). Linking land use, erosion and sediment yields in river basins. *Hydrobiologia*, 410(0):223–240.

Walling, D. E. and Quine, T. A. (1992). The use of caesium-137 measurements in soil erosion surveys. In *Erosion and sediment monitoring programmes in river basins*. *Proc. international symposium, Oslo,* 1992, pages 143–152.

Wang, X., Huang, G., and Liu, J. (2014). Projected increases in intensity and frequency of rainfall extremes through a regional climate modeling approach. *JGR: Atmospheres*, 119(23):13,271–286.

Wang, X., Kemanian, A. R., Williams, J. R., Ahuja, L. R., and Ma, L. (2011). Special Features of the EPIC and APEX Modeling Package and Procedures for Parameterization, Calibration, Validation, and Applications. 16802.

Warszawski, L., Frieler, K., Huber, V., Piontek, F., Serdeczny, O., and Schewe, J. (2014). The intersectoral impact model intercomparison project (ISI-MIP): Project framework. *Proceedings of the National Academy of Sciences of the United States of America*, 111(9):3228–3232.

Watson, A. and Evans, R. (1991). A comparison of estimates of soil erosion made in the field and from photographs. *Soil Tillage Research*, 19:17–27.

Webb, N. P., Marshall, N. A., Stringer, L. C., Reed, M. S., Chappell, A., and Herrick, J. E. (2017). Land degradation and climate change: building climate resilience in agriculture. *Frontiers in Ecology and the Environment*, 15(8):450–459.

Webber, H., Gaiser, T., and Ewert, F. (2014). What role can crop models play in supporting climate change adaptation decisions to enhance food security in Sub-Saharan Africa? *Agricultural Systems*, 127:161–177.

Webber, H. A., Madramootoo, C. A., Bourgault, M., Horst, M. G., Stulina, G., and Smith, D. L. (2010). Adapting the CROPGRO model for saline soils: The case for a common bean crop. *Irrigation Science*, 28(4):317–329.

Wheater, H. S., Jakeman, A. J., and Beven, K. J. (1993). Progress and directions in rainfall-runoff modelling. In Jakeman, A., Beck, M., and McAleer, M., editors, *Modelling Change in Environmental Systems*, pages 101–132.

Wheeler, T. and Von Braun, J. (2013). Climate change impacts on global food security. *Science*, 341(6145):508–513.

White, J. W., Hoogenboom, G., Kimball, B. A., and Wall, G. W. (2011). Methodologies for simulating impacts of climate change on crop production. *Field Crops Research*, 124(3):357–368.

Wildemeersch, J. C., Garba, M., Sabiou, M., Sleutel, S., and Cornelis, W. (2015). The Effect of Water and Soil Conservation (WSC) on the Soil Chemical, Biological, and Physical Quality of a Plinthosol in Niger. *Land Degradation and Development*, 26(7):773–783.

Willcocks, T. J. and Twomlow, S. J. (1993). A review of tillage methods and soil and water conservation in southern Africa. *Soil and Tillage Research*, 27(1-4):73–94.

Williams, J. R. (1975). Sediment yield prediction with universal equation on using runoff energy factor. In *Present and prospective technology for predicting sediment yields and sources, ARS S-40*, pages 244–252. USDA-ARS, Washington.D.C.

Williams, J. R. (1990). The Erosion-Productivity Impact Calculator (EPIC) Model: A Case History. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 329(1255):421–428.

Williams, J. R. (1995). The EPIC model. In Singh, V. P., editor, *Computer Models of Watershed Hydrology*, pages 909–1000. Water Resources Publications.

Williams, J. R., Izaurralde, R. C., and Steglich, E. M. (2012). Agricultural Policy/Environmental eXtender Model, Theoretical documentation version 0806. Technical report.

Wirsenius, S. (2000). *Human Use of Land and Organic materials - Modeling the Turnover of Biomass in the Global Food System*. PhD thesis, Chalmers University of Technology and Göteborg University.

Wischmeier, W. and Smith, D. (1978). Predicting rainfall erosion losses. *Agriculture handbook no.* 537, (537):285–291.

Wischmeier, W. H. and Smith, D. D. (1965). Predicting rainfall-erosion losses from cropland east of the Rocky Mountains. Technical report, US Department of Agriculture, Washington DC.

World Bank (2020a). World Bank Commodities Price Data (The Pink Sheet). available at: https://www.worldbank.org/en/research/commodity-markets (last access: 08.03.2021).

World Bank (2020b). World Development Indicators. available at: https://databank.worldbank.org/source/world-development-indicators (last access: 08.03.2021).

Wu, W., Shibasaki, R., Yang, P., Tan, G., ichiro Matsumura, K., and Sugimoto, K. (2007). Global-scale modelling of future changes in sown areas of major crops. *Ecological Modelling*, 208(2-4):378–390.

Wynants, M., Kelly, C., Mtei, K., Munishi, L., Patrick, A., Rabinovich, A., Nasseri, M., Gilvear, D., Roberts, N., Boeckx, P., Wilson, G., Blake, W. H., and Ndakidemi, P. (2019). Drivers of increased soil erosion in East Africa's agro-pastoral systems: changing interactions between the social, economic and natural domains. *Regional Environmental Change*, 19(7):1909–1921.

Yang, D., Kanae, S., Oki, T., Koike, T., and Musiake, K. (2003). Global potential soil erosion with reference to land use and climate changes. *Hydrological Processes*, 17(14):2913–2928.

Yin, X. and Leng, G. (2021). Modelling global impacts of climate variability and trend on maize yield during 1980–2010. *International Journal of Climatology*, 41(S1):E1583–E1596.

Zabel, F., Putzenlechner, B., and Mauser, W. (2014). Global agricultural land resources - A high resolution suitability evaluation and its perspectives until 2100 under climate change conditions. *PLoS ONE*, 9(9):1–12.

Zachar, D. (1982). Soil Erosion. Elsevier, Amsterdam.

Zapata, F. (2002). Handbook for the Assessment of Soil Erosion and Sedimentation Using Environmental Radionuclides. Springer.

Zhang, B., Yang, Y. S., and Zepp, H. (2004). Effect of vegetation restoration on soil and water erosion and nutrient losses of a severely eroded clayey Plinthudult in southeastern China. *Catena*.

Zhang, X. C. and Nearing, M. A. (2005). Impact of climate change on soil erosion, runoff, and wheat productivity in central Oklahoma. *Catena*, 61(2-3 SPEC. ISS.):185–195.

Zhao, C., Liu, B., Piao, S., Wang, X., Lobell, D. B., Huang, Y., Huang, M., Yao, Y., Bassu, S., Ciais, P., Durand, J.-L., Elliott, J., Ewert, F., Janssens, I. A., Li, T., Lin, E., Liu, Q., Martre, P., Müller, C., Peng, S., Peñuelas, J., Ruane, A. C., Wallach, D., Wang, T., Wu, D., Liu, Z., Zhu, Y., Zhu, Z., and Asseng, S. (2017). Temperature increase reduces global yields of major crops in four independent estimates. *Proceedings of the National Academy of Sciences*, page 201701762.

Zheng, B., Campbell, J. B., Serbin, G., and Galbraith, J. M. (2014). Remote sensing of crop residue and tillage practices: Present capabilities and future prospects. *Soil and Tillage Research*, 138:26–34.

Zheng, H., Liu, W., Zheng, J., Luo, Y., Li, R., Wang, H., and Qi, H. (2018). Effect of long-term tillage on soil aggregates and aggregate-associated carbon in black soil of northeast China. *PLoS ONE*, 13(6):1–18.